ANALYSIS OF STRUCTURED POLYNOMIAL EIGENVALUE PROBLEMS

A thesis submitted to the University of Manchester for the degree of Doctor of Philosophy in the Faculty of Engineering and Physical Sciences

2011

Maha Al-Ammari School of Mathematics

Contents

A	bstra	nct		8				
D	eclar	ation		9				
С	opyri	\mathbf{ight}		10				
P	ublic	ations		11				
D	edica	tion		12				
A	ckno	wledge	ements	13				
1	Introduction							
	1.1	Notat	ions	16				
	1.2	Prelin	ninaries	18				
		1.2.1	Some structured matrix polynomials	19				
		1.2.2	Elementary divisors	20				
		1.2.3	Linearization	21				
2	Classification of Quasidefinite Matrix Polynomials 23							
	2.1	Introduction						
	2.2	ninaries	24					
		2.2.1	Eigenvalue type and sign characteristic $\ldots \ldots \ldots \ldots \ldots$	24				
		2.2.2	Motivation	28				
		2.2.3	Homogeneous rotation	29				
		2.2.4	Hermitian linearizations	32				
	2.3	2.3 Definite matrix polynomials						
		2.3.1	Definite pencils	34				
		2.3.2	Hyperbolic matrix polynomials	37				
		2.3.3	Overdamped quadratic matrix polynomials	43				
		2.3.4	Definiteness of matrix polynomials	45				
	2.4	Quasidefinite matrix polynomials						

		2.4.1	Definitizable pencils	47				
		2.4.2	Quasihyperbolic matrix polynomials	50				
		2.4.3	Gyroscopically stabilized systems	52				
	2.5	Diagor	nalizable quasidefinite matrix polynomials	53				
	2.6	Summ	ary and further discussion	57				
3	Sta	ndard '	Triples of Structured Matrix Polynomials	58				
	3.1	Introd	uction	58				
	3.2	Prelim	ninaries	60				
		3.2.1	Structured linearizations	60				
		3.2.2	Standard triples	64				
	3.3	$\mathcal{S} ext{-stru}$	ctured standard triples	66				
	3.4	$\mathcal{S} ext{-stru}$	ctured Jordan triples	73				
	3.5	Explic	it expressions for J and S_J	75				
		3.5.1	Hermitian structure	76				
		3.5.2	Real symmetric structure	76				
		3.5.3	Complex symmetric structure	77				
		3.5.4	*-even structure	77				
		3.5.5	Real T -even structure $\ldots \ldots \ldots$	77				
		3.5.6	Complex T -even structure \ldots	78				
		3.5.7	*-odd structure	79				
		3.5.8	Real T -odd structure	79				
		3.5.9	Complex T -odd structure $\ldots \ldots \ldots$	80				
		3.5.10	*-(anti)palindromic structure	80				
		3.5.11	Real T -(anti)palindromic structure $\ldots \ldots \ldots \ldots \ldots \ldots$	81				
		3.5.12	Complex T -(anti)palindromic structure	82				
	3.6	Applic	eations	83				
	3.7	Conclu	ıding remarks	86				
4	Stru	uctured	d Inverse Polynomial Eigenvalue Problems	87				
	4.1	Introd	uction	87				
	4.2	Preliminaries						
	4.3	Building \mathcal{S} -structured Jordan triples $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$						
		4.3.1	Hermitian and real symmetric quadratics	90				
		4.3.2	Complex symmetric quadratics	94				
		4.3.3	*-even quadratics	95				
		4.3.4	Real T -even quadratics	97				
		4.3.5	Complex T -even quadratics $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	98				
		4.3.6	*-odd quadratics	98				

		4.3.7	Real T -odd quadratics $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	99		
		4.3.8	Complex T -odd quadratics	100		
		4.3.9	*-(anti)-palindromic quadratics	100		
		4.3.10	Real T -(anti)-palindromic quadratics $\ldots \ldots \ldots \ldots \ldots$	102		
		4.3.11	Complex T -(anti)-palindromic quadratics	102		
	4.4	Other	structured inverse eigenvalue problems	103		
		4.4.1	Generating quasidefinite cubics $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	103		
		4.4.2	Generating tridiagonal hyperbolic quadratics	105		
	4.5	Conclu	ding remarks	109		
5	Con	clusior	ns and Future Work	110		
Bi	Bibliography 1					

(Word count: 25622)

List of Tables

1.1	Matrix polynomials	$P(\lambda) =$	$=\sum_{j=0}^m \lambda^j \lambda^j$	$A_j, \det(A_j)$	$(A_m) \neq 0$	with structure \mathcal{S} .	19
-----	--------------------	----------------	-------------------------------------	------------------	----------------	--------------------------------	----

3.1 Eigentriple pairings (if any) for $P(\lambda) \in \mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$ with structure $\mathcal{S} \in \mathbb{S}$. When $\mathbb{F} = \mathbb{R}$, the eigenvalues of *T*-even and *T*-odd $P(\lambda)$ occur in quadruples $(\lambda, \overline{\lambda}, -\lambda, -\overline{\lambda})$ and those of *T*-palindromic and *T*-antipalindromic $P(\lambda)$ occur in quadruples $(\lambda, \overline{\lambda}, 1/\lambda, 1/\overline{\lambda})$ 59

List of Figures

1.1	A mass-spring damped system with one degree of freedom	16
2.1	Quasidefinite $n \times n$ matrix polynomials $P(\lambda) = \sum_{j=0}^{m} \lambda^{j} A_{j}$ of degree	
	m and their subclasses. A subclass ${\bf A}$ pointing to a subclass ${\bf B}$ with	
	a solid line (dotted line) and property "C" means that the subclass ${\bf A}$	
	with the property "C" is exactly (is contained in) the subclass ${\bf B}.$	25
2.2	Correspondence between λ and (α, β) .	29
2.3	Distribution of the eigenvalue types for three types of definite pencils	
	$L(\lambda) = \lambda A - B$. On the shaded intervals L is indefinite.	35
2.4	Pictorial representation of (2.14).	36
2.5	Distribution of eigenvalue types of $n \times n$ hyperbolic polynomials of	
	even degree m .	38
2.6	The eigenvalues interlacing of a 3×3 quadratic matrix polynomial Q	
	and its leading 2×2 submatrix \widehat{Q}	40
2.7	Definiteness diagram	41
2.8	Eigenvalue distribution of gyroscopically stabilized quadratics $Q(\lambda) =$	
	$\lambda^2 A_2 + \lambda A_1 + A_0$, where p is the number of positive eigenvalues of A_1 .	53

The University of Manchester

Maha Al-Ammari Doctor of Philosophy Analysis of Structured Polynomial Eigenvalue Problems August 4, 2011

This thesis considers Hermitian/symmetric, alternating and palindromic matrix polynomials which all arise frequently in a variety of applications, such as vibration analysis of dynamical systems and optimal control problems. A classification of Hermitian matrix polynomials whose eigenvalues belong to the extended real line, with each eigenvalue being of definite type, is provided first. We call such polynomials quasidefinite. Definite pencils, definitizable pencils, overdamped quadratics, gyroscopically stabilized quadratics, (quasi)hyperbolic and definite matrix polynomials are all quasidefinite. We show, using homogeneous rotations, special Hermitian linearizations and a new characterization of hyperbolic matrix polynomials, that the main common thread between these many subclasses is the distribution of their eigenvalue types. We also identify, amongst all quasihyperbolic matrix polynomials, those that can be diagonalized by a congruence transformation applied to a Hermitian linearization of the matrix polynomial while maintaining the structure of the linearization.

Secondly, we generalize the notion of self-adjoint standard triples associated with Hermitian matrix polynomials in Gohberg, Lancaster and Rodman's theory of matrix polynomials to present spectral decompositions of structured matrix polynomials in terms of standard pairs (X, T), which are either real or complex, plus a parameter matrix S that acquires particular properties depending on the structure under investigation. These decompositions are mainly an extension of the Jordan canonical form for a matrix over the real or complex field so we investigate the important special case of structured Jordan triples.

Finally, we use the concept of structured Jordan triples to solve a structured inverse polynomial eigenvalue problem. As a consequence, we can enlarge the collection of nonlinear eigenvalue problems [NLEVP, 2010] by generating quadratic and cubic quasidefinite matrix polynomials in different subclasses from some given spectral data by solving an appropriate inverse eigenvalue problem. For the quadratic case, we employ available algorithms to provide tridiagonal definite matrix polynomials.

Declaration

No portion of the work referred to in this thesis has been submitted in support of an application for another degree or qualification of this or any other university or other institute of learning.

Copyright

- i. The author of this thesis (including any appendices and/or schedules to this thesis) owns certain copyright or related rights in it (the "Copyright") and s/he has given The University of Manchester certain rights to use such Copyright, including for administrative purposes.
- ii. Copies of this thesis, either in full or in extracts and whether in hard or electronic copy, may be made only in accordance with the Copyright, Designs and Patents Act 1988 (as amended) and regulations issued under it or, where appropriate, in accordance with licensing agreements which the University has from time to time. This page must form part of any such copies made.
- iii. The ownership of certain Copyright, patents, designs, trade marks and other intellectual property (the "Intellectual Property") and any reproductions of copyright works in the thesis, for example graphs and tables ("Reproductions"), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property and/or Reproductions.
- iv. Further information on the conditions under which disclosure, publication and commercialisation of this thesis, the Copyright and any Intellectual Property and/or Reproductions described in it may take place is available in the University IP Policy (see http://www.campus.manchester.ac.uk/medialibrary/ policies/intellectual-property.pdf), in any relevant Thesis restriction declarations deposited in the University Library, The University Librarys regulations (see http://www.manchester.ac.uk/library/aboutus/regulations) and in The Universitys policy on presentation of Theses.

Publications

- The material in Chapter 2 is based on the paper "Hermitian matrix polynomials with real eigenvalues of definite type- part I: Classification" [3] (with Françoise Tisseur), *Linear Algebra Appl.* (2010), doi:10.1016/j.laa.2010.08.035.
- The material in Chapter 3 is based on the technical report "Standard triples for structured matrix polynomials" [4] (with Françoise Tisseur), MIMS EPrint 2011.37, Manchester Institute for Mathematical Sciences, The University of Manchester, UK, May 2011. 20 pp., submitted to *Linear Algebra Appl.*

Dedication

To all the exceptional people involved in my academic life.

Acknowledgements

First of all, my great thanks go to *Allah* who made most of my dreams come true.

I am utterly grateful to my supervisor, Dr Françoise Tisseur, who introduced me to the interesting area of matrix polynomials and offered patience, kind supervision and continual support. Over the past four years, she has been a role model from whom I have learnt so much more than mathematical research skills. This work would not have been a thesis without Fran.

It was an honor to be a part of the numerical linear algebra group in Manchester, which is led by Professor Nicholas. J. Higham, an academic legend to whom I cannot express strongly enough my gratitude for his teaching, both through direct comments and indirect influence.

I appreciate the valuable comments given by Dr Christian Mehl from TU Berlin on the whole thesis. Dr Fernando Terán from Universidad Carlos III de Madrid and Dr Ion Zaballa from Universidad del Pais Vasco provided useful suggestions on an earlier draft of the technical report on which Chapter 3 of this thesis is based. I am also thankful to both Professor Roy Mathias for his course on Matrix Analysis and Dr D. Steven Mackey from Western Michigan University for the discussions and the potential collaboration we are planning to have about structured matrix polynomials.

My profound gratitude is due to the Department of Mathematics at The University of Manchester for providing an excellent research environment and to the Ministry of Higher Education in Saudi Arabia for its generous funding of my studies.

I owe my special friend and office mate, Lijing Lin, more than words can say. She was always there to supply technical support, and share the good and bad moments, with her sweet nature. Many thanks also go to Alaa, Chris, Elham, Eyman, Haitham, Qusay, Ram, Rudiger and Tarifa for the nice memories they provided in Manchester.

Finally, I am thankful to each member of my extended family, especially my father, grandmother, aunt Souad, uncle Zaki, Reem and Ammar. This thesis is completed with the support of my beloved mother, who has been a devoted parent, a close friend and an astounding enthusiast throughout all my life.

Chapter 1

Introduction

Matrix polynomials [31], λ -matrices [50] and polynomial matrices [73] are different names for polynomials in λ whose coefficients are matrices, or equivalently, matrices whose entries are polynomials in λ with scalar coefficients. The polynomial eigenvalue problem (PEP), which is to find eigentriples $(\lambda, x, y) \in \mathbb{C} \cup \{\infty\} \times \mathbb{C}^n \setminus \{0\} \times \mathbb{C}^n \setminus \{0\}$ that satisfy $P(\lambda)x = 0$ and $y^*P(\lambda) = 0$, where

$$P(\lambda) = \lambda^m A_m + \lambda^{m-1} A_{m-1} + \dots + \lambda A_1 + A_0, \qquad (1.1)$$

 $A_j \in \mathbb{C}^{n \times n}$ for j = 0: m, extends the well-known standard eigenvalue problem (SEP) and the generalized eigenvalue problem (GEP), for which $P(\lambda)$ in (1.1) has the forms $-\lambda I + A$ and $\lambda A - B$ respectively, where $A, B \in \mathbb{C}^{n \times n}$. In general, the finite eigenvalues of P are the solutions of the characteristic equation

$$\det\left(P(\lambda)\right) = 0. \tag{1.2}$$

Infinity is an eigenvalue of P when $\det(A_m) = 0$. The vectors x and y are right and left eigenvectors corresponding to the eigenvalue λ , respectively. If $A_m \neq 0$ then m is the degree of P, denoted by $\deg(P)$. $P(\lambda)$ is *regular* whenever its determinant does not vanish identically. Throughout this thesis, $P(\lambda)$ is assumed to be regular with $n \times n$ matrix coefficients and degree m.

The PEP is a subclass of the more general nonlinear eigenvalue problem (NLEVP) that takes the form of finding eigentriples $(\lambda, x, y) \in \mathbb{C} \cup \{\infty\} \times \mathbb{C}^n \setminus \{0\} \times \mathbb{C}^n \setminus \{0\}$ satisfying $T(\lambda)x = 0$ and $y^*T(\lambda) = 0$, where $T : \mathbb{C} \to \mathbb{C}^{n \times n}$ is a nonlinear operator. PEPs arise in many applications in science and engineering. Examples include acoustic structural coupled systems, fluid mechanics, multiple input multiple output systems in control theory, signal processing, and constrained least square problems, for all of which details and references are listed in [90]. Frazer et al.'s book on Elementary Matrices [26] includes a collection of PEPs' applications to aerodynamical systems, for example equations of motion of an aeroplane and small motions of systems subject to aerodynamical forces. The most commonly occurring PEPs are quadratic eigenvalue problems (QEPs)

$$(\lambda^2 A_2 + \lambda A_1 + A_0)x = 0, (1.3)$$

where in mechanical systems A_2 is a mass matrix, A_1 a damping matrix, A_0 a stiffness matrix, λ a resonant frequency, and x a mode shape. Consider a simple standard mass-spring damper system with a free body of mass m, a spring of stiffness coefficient k, viscous damper of damping coefficient d and assume that no external force acts on the system (see Figure 1.1). The solutions of the differential equation

$$m\frac{d^2v}{dt^2} + d\frac{dv}{dt} + kv = 0 \tag{1.4}$$

determine the dynamics of the system, where v is the displacement of the mass relative to a fixed point. Clearly, the solutions of (1.4) have the form

$$v(t) = xe^{\lambda t},$$

where x is a right eigenvector associated with λ as an eigenvalue of the quadratic scalar polynomial $\lambda^2 m + \lambda d + k$. Such a system is overdamped [11] if $d^2 > 4mk$. Note that, as m, d and k are positive, the overdamped system has real negative distinct eigenvalues. Consequently, all the solutions v(t) approach zero as $t \to \infty$. Similarly, systems with multiple degrees of freedom give rise to (1.3) where A_2, A_1 are Hermitian positive definite and A_0 is Hermitian positive semidefinite, which extends the idea of overdamping. In fact, overdamped quadratics form a subclass of special Hermitian matrix polynomials that have interesting stability properties under Hermitian perturbations. Such Hermitian matrix polynomials and some closely related structured



Figure 1.1: A mass-spring damped system with one degree of freedom.

matrix polynomials are the topics of concern in this work. Note that $P(\lambda)$ of degrees larger than two also arise in applications [44], [76], [84].

This chapter covers the background material, introduces the notation that is used throughout the thesis and outlines its contents.

1.1 Notations

- F is the field of real or complex numbers.
- $\mathbb{F}^{n \times n}$ is the set of $n \times n$ matrices with coefficients in \mathbb{F} .
- A^T is the transpose of the matrix A.
- $A^* = (\overline{A})^T$ is the conjugate transpose of the matrix A.
- \star denotes the transpose T for real matrices and either the transpose T or the conjugate transpose \star for matrices with complex entries.
- ker A is the kernel of the matrix A.
- A > 0 denotes $A \in \mathbb{C}^{n \times n}$ that is positive definite, i.e.

$$A = A^*, \quad x^*Ax > 0, \quad \forall x \in \mathbb{C}^n \setminus \{0\}.$$

- A < 0 denotes $A \in \mathbb{C}^{n \times n}$ that is negative definite, i.e. -A > 0.
- $A \in \mathbb{C}^{n \times n}$ is definite if A > 0 or -A > 0.

- A₁, A₂ ∈ C^{n×n} have opposite parity if one is positive definite and the other is negative definite. Similarly, a sequence A₀, A₁, A₂,... of definite matrices has alternating parity if A_j and A_{j+1} have opposite parity for all j.
- $A \ge 0$ denotes $A \in \mathbb{C}^{n \times n}$ that is positive semi-definite, i.e.

$$A = A^*, \quad x^* A x \ge 0 \quad \forall x \in \mathbb{C}^n.$$

- For A ∈ F^{n×n} with no eigenvalues on the closed negative real axis, A^{1/2} denotes the principal square root of A which is the unique square root of A all of whose eigenvalues lie in the segment {z : −π/2 < arg(z) < π/2} [36, Thm. 1.29].
- A^{-1} denotes the inverse of $A \in \mathbb{F}^{n \times n}$.
- I_n is the identity $n \times n$ matrix.
- e_i is the vector that is equal to the *i*-th column of I_n .
- If $A_m \neq 0$ then the positive integer m is the degree of $P(\lambda) = \sum_{j=0}^m \lambda^j A_j$, denoted by deg (P), where $A_j \in \mathbb{F}^{n \times n}$ for j = 0: m.
- The homogenous form of the matrix polynomial $P(\lambda) = \sum_{j=0}^{m} \lambda^{j} A_{j}$ is

$$P(\alpha,\beta) = \sum_{j=0}^{m} \alpha^{j} \beta^{m-j} A_{j},$$

where λ is identified with any pair $(\alpha, \beta) \neq (0, 0)$ for which $\lambda = \alpha/\beta$.

• rev $P(\lambda) := \sum_{j=0}^{m} \lambda^{m-j} A_j$ is the reversal polynomial of $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j$.

•
$$P^{\star}(\lambda) := \sum_{j=0}^{m} \lambda^j A_j^{\star}.$$

- $P^{\star}(\alpha,\beta) := \sum_{j=0}^{m} \alpha^{j} \beta^{m-j} A_{j}^{\star}.$
- P'_λ is the first derivative of P with respect to λ. The subscript λ is dropped when there is no ambiguity.
- $\mathcal{P}(\mathbb{F}^n)$ denotes the set of all matrix polynomials with coefficient matrices in $\mathbb{F}^{n \times n}$. When the polynomials are structured with structure \mathcal{S} , the corresponding space is denoted by $\mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$.

- $\sigma(P)$ is the set of the eigenvalues of P.
- $\rho(P) = \max\{|\lambda| : \lambda \in \sigma(P)\}$ is the spectral radius of P.
- sig(S) denotes the signature of S, that is, the difference between the number of positive eigenvalues and the number of negative eigenvalues of S.
- If $A = (A_{ij})$ is a block $k \times \ell$ matrix with $m \times n$ blocks A_{ij} , then the block transpose of A is the block $\ell \times k$ matrix $A^{\mathcal{B}}$ with $m \times n$ blocks defined by $(A^{\mathcal{B}})_{ij} = A_{ji}$.
- The Kronecker product of $A \in \mathbb{F}^{m \times n}$ and $B \in \mathbb{F}^{q \times p}$ is [43, Section 4.2],

$$A \otimes B = \begin{bmatrix} a_{11}B & \cdots & a_{1n}B \\ \vdots & & \vdots \\ a_{m1}B & \cdots & a_{mn}B \end{bmatrix} \in \mathbb{F}^{mp \times nq}, \quad \text{where} \quad A = \begin{bmatrix} a_{11} & \cdots & a_{1n} \\ \vdots & & \vdots \\ a_{m1} & \cdots & a_{mn} \end{bmatrix}.$$

• The direct sum of $A \in \mathbb{F}^{n \times n}$ and $B \in \mathbb{F}^{m \times m}$ is [42, Section 0.9.2],

$$A \oplus B = \begin{bmatrix} A & 0 \\ 0 & B \end{bmatrix} \in \mathbb{F}^{(n+m) \times (n+m)}.$$

- $\bigoplus_{j=1}^m A_j = A_1 \oplus \cdots \oplus A_m.$
- $\mathbb{L}_1(P) = \{ L(\lambda) : L(\lambda)(\Lambda \otimes I_n) = v \otimes P(\lambda), v \in \mathbb{C}^m \}, \text{ where}$ $L(\lambda) = \lambda A - B, \quad A, B \in \mathbb{C}^{n \times n} \text{ and } \Lambda = [\lambda^{m-1}, \lambda^{m-2}, \dots, 1]^T \in \mathbb{C}^m.$

•
$$\mathbb{H}(P) = \left\{ L(\lambda) = \lambda A - B \in \mathbb{L}_1(P) : A = A^*, B = B^* \right\}.$$

• $\mathbb{L}_2(P) = \{ L(\lambda) : (\Lambda^T \otimes I_n) L(\lambda) = w^T \otimes P(\lambda), w \in \mathbb{C}^m \}, \text{ where}$ $L(\lambda) = \lambda A - B, \quad A, B \in \mathbb{C}^{n \times n} \text{ and } \Lambda = [\lambda^{m-1}, \lambda^{m-2}, \dots, 1]^T \in \mathbb{C}^m.$

•
$$\mathbb{DL}(P) = \mathbb{L}_1(P) \cap \mathbb{L}_2(P).$$

1.2 Preliminaries

This section collects the basic definitions and concepts that we need in the sequel.

1.2.1 Some structured matrix polynomials

Table 1.1 defines the structures of interest in this work. We use \star -alternating to refer to either \star -even or \star -odd polynomials, where \star denotes the transpose T for real matrices and either the transpose T or the conjugate transpose \star for matrices with complex entries. For real-life applications associated with these structured matrix polynomials, see [65, Section 6.4] and [13, Section 3].

Structure ${\mathcal S}$ Definition Coefficients property Hermitian $P(\lambda) = P^*(\lambda)$ $A_i = A_i^*$ $P(\lambda) = P^T(\lambda)$ $A_j = A_j^T$ symmetric $P(\lambda) = P^{\star}(-\lambda) \qquad \qquad A_j = (-1)^j A_j^{\star}$ *-even $P(\lambda) = -P^{\star}(-\lambda) \qquad \qquad A_j = (-1)^{j+1} A_j^{\star}$ *-odd $P(\lambda) = \lambda^m P^{\star}(\frac{1}{\lambda})$ $A_j = A_{m-j}^{\star}$ *-palindromic *-antipalindromic $| P(\lambda) = -\lambda^m P^{\star}(\frac{1}{\lambda}) | \qquad A_j = -A_{m-j}^{\star}$

Table 1.1: Matrix polynomials $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j$, $\det(A_m) \neq 0$ with structure \mathcal{S} .

In Chapter 2 we present the diverse subclasses of Hermitian λ -matrices with real eigenvalues, specifically those for which every eigenvalue λ_0 satisfies $x^*P'(\lambda_0)x \neq 0$ for all nonzero $x \in \ker P(\lambda_0)$, which we call *quasidefinite* matrix polynomials. We collect and unify existing results scattered in the literature about such polynomials, propose a new characterization of hyperbolicity and show how to avoid the requirement of having a nonsingular or a definite leading coefficient matrix of P. Moreover, we identify an important new class of diagonalizable matrix polynomials. For the sake of completeness, we provide a survey of the available algorithms for detecting some specific structures of Hermitian matrix polynomials in Figure 2.1.

The notion of a Jordan triple (X, J, Y), introduced by Gohberg, Lancaster and Rodman, summarizes the spectral properties of $P(\lambda)$ (see [58, Section 14.2]). In particular if (X, J, Y) is a Jordan triple for $P(\lambda)$, then

$$\sum_{j=0}^{m} A_j X J^j = 0, \qquad \sum_{j=0}^{m} J^j Y A_j = 0.$$

In Chapter 3 we investigate Jordan triples of the structured λ -matrices listed in Table 1.1, which have extra properties depending on the underlying structure. For instance, GLR in [29] show that if (X, J, Y) is a Jordan triple for a Hermitian matrix polynomial then $Y = SX^*$ for some nonsingular $mn \times mn$ matrix S such that $S = S^*$, $JS = (JS)^*$. They call Jordan triples with such properties *self-adjoint*. We show that this type of result also holds for all the structures S listed in Table 1.1. We introduce the notion of S-structured standard triples and show that, apart from T-(anti)palindromic matrix polynomials of even degree with both -1 and 1 as eigenvalues, if $P(\lambda)$ admits an S-structured standard triple then $P(\lambda)$ has structure S. Reciprocally, if $P(\lambda)$ has structure S then its standard triples are S-structured.

Chapter 4 highlights the use of the S-structured standard triples theory in solving a structured inverse polynomial eigenvalue problem (SIPEP) of the following form: Construct a quadratic matrix polynomial $Q(\lambda)$ with structure S from a given admissible list of elementary divisors. Additionally, Chapter 4 concerns generating quadratic and cubic matrix polynomials in different subclasses displayed in Figure 2.1 from some given spectral data by solving an appropriate inverse polynomial eigenvalue problem (IPEP). Furthermore, it points out that generating hyperbolic quadratic matrix polynomials that are real monic (i.e., with I_n as the leading coefficient matrix) and tridiagonal is always possible when the eigenvalues are given. As a result, we contribute to increase the collection of NLEVPs [13].

1.2.2 Elementary divisors

In order to define elementary divisors, we recall some fundamental notions from the theory of matrix polynomials [31, Chap. S1].

Theorem 1.2.1 [31, Thm. S1.1] Every $n \times n$ matrix polynomial $P(\lambda)$ admits the representation

$$P(\lambda) = E(\lambda) D(\lambda) F(\lambda), \qquad (1.5)$$

where

$$D(\lambda) = \operatorname{diag}(d_1(\lambda), \dots, d_r(\lambda), 0, \dots, 0),$$
(1.6)

is a diagonal λ -matrix with monic scalar polynomials $d_i(\lambda)$ such that $d_i(\lambda)$ is divisible by $d_{i-1}(\lambda)$, $r = \max_{\lambda \in \mathbb{C}} \{ \operatorname{rank} P(\lambda) \}$ and the $n \times n$ matrix polynomials $E(\lambda)$, $F(\lambda)$ have constant nonzero determinants. Representation (1.5) as well as the diagonal matrix $D(\lambda)$ from (1.6) is called the *Smith form* of $P(\lambda)$. The diagonal matrix $D(\lambda)$ is unique while the matrix polynomials $E(\lambda)$ and $F(\lambda)$ are not defined uniquely. The diagonal elements $d_1(\lambda), \ldots, d_r(\lambda)$ in the Smith form are called the *invariant polynomials* of $P(\lambda)$. Each invariant polynomial can be written as

$$d_i(\lambda) = (\lambda - \lambda_{i1})^{\alpha_{i1}} \cdots (\lambda - \lambda_{ik_i})^{\alpha_{ik_i}}, \quad i = 1: r,$$

where the complex numbers $\lambda_{i1}, \ldots, \lambda_{ik_i}$ are distinct and $\alpha_{i1}, \ldots, \alpha_{ik_i}$ are positive integers. The factors $(\lambda - \lambda_{ij})^{\alpha_{ij}}$, $j = 1: k_i$, i = 1: r, are the *elementary divisors* of $P(\lambda)$.

1.2.3 Linearization

A standard way of treating the PEP to find *eigenpairs* $(\lambda, x) \in \mathbb{C} \cup \{\infty\} \times \mathbb{C}^n \setminus \{0\}$ that satisfy $P(\lambda)x = \left(\sum_{j=0}^m \lambda^j A_j\right)x = 0$, both theoretically and numerically, is to convert $P(\lambda)$ into an equivalent linear matrix pencil $L(\lambda) = \lambda A - B \in \mathbb{C}^{mn \times mn}$ using the process known as *linearization*. To be more specific, L is a linearization of P if it satisfies

$$E(\lambda)L(\lambda)F(\lambda) = \begin{bmatrix} P(\lambda) & 0\\ 0 & I_{(m-1)n} \end{bmatrix}$$
(1.7)

for some matrix polynomials $E(\lambda)$ and $F(\lambda)$ with constant nonzero determinants. It is clear that the eigenvalues of L and P coincide. As an example, if A_m is nonsingular, the pencil $\lambda I - C$ with

$$C = \begin{bmatrix} -A_m^{-1} & & \\ & I_n & \\ & & \ddots & \\ & & & I_n \end{bmatrix} \begin{bmatrix} A_{m-1} & A_{m-2} & \dots & A_0 \\ I_n & 0 & \dots & 0 \\ & & \ddots & \ddots & \vdots \\ 0 & & & I_n & 0 \end{bmatrix}$$
(1.8)

is the first companion form linearization of $A_m^{-1}P(\lambda)$. In recent work [70] vector spaces of pencils have been studied, namely,

$$\mathbb{L}_1(P) = \left\{ L(\lambda) : L(\lambda)(\Lambda \otimes I_n) = v \otimes P(\lambda), \ v \in \mathbb{C}^m \right\},\tag{1.9}$$

and $\mathbb{D}\mathbb{L}(P) := \{ L(\lambda) = \lambda A - B \in \mathbb{L}_1(P) : A^{\mathcal{B}} = A, B^{\mathcal{B}} = B \}$, where $\Lambda = [\lambda^{m-1}, \lambda^{m-2}, \ldots, 1]^T \in \mathbb{C}^m$ and $A^{\mathcal{B}}$ is the block transpose of A [37, Def. 2.1]. $\mathbb{L}_1(P)$ generalizes the first companion form (indeed for \mathcal{C} in (1.8), $\lambda I - \mathcal{C} \in \mathbb{L}_1(A_m^{-1}P)$ with $v = e_1$) and supplies a rich source of interesting linearizations.

The following result provides a condition that $L \in \mathbb{DL}(P)$ be a linearization of P.

Theorem 1.2.2 (eigenvalue exclusion theorem) [70, Thm. 6.7]

Suppose that $P(\lambda)$ is a regular matrix polynomial of degree m and $L(\lambda) \in \mathbb{DL}(P)$ with vector $v \in \mathbb{C}^m$. Then $L(\lambda)$ is a linearization for $P(\lambda)$ if and only if no root of the v-polynomial

$$\mathbf{p}(x;v) := v_1 x^{m-1} + v_2 x^{m-2} + \dots + v_{m-1} x + v_m \tag{1.10}$$

is an eigenvalue of $P(\lambda)$, where, by convention, p(x; v) has a root at ∞ whenever $v_1 = 0$.

It is worth mentioning that in (1.7), the identity block can be of a size less than (m-1)n, which introduces the *trimmed* linearizations that allow partial deflation of ∞ as an eigenvalue (i.e., obtaining as short as possible Jordan chains associated with ∞) [14]. For lower bounds of the size of the identity block appearing in the definition of linearization, see [23].

The first companion form linearization $\lambda I - C$ of $A_m^{-1}P(\lambda)$ with C in (1.8) is easily constructed from the coefficients of $P(\lambda)$ but it does not reflect its structure. For example, C is not Hermitian when $P(\lambda)$ is Hermitian. This might lead to problems such as the loss of spectral symmetry when eigenvalues are computed in finite precision arithmetic. Linearizations that respect the structure are the subject of Section 2.2.4 and Section 3.2.1. A huge number of linearizations is available in the literature; see for example [5], [7], [8], [24], [37], [69], and [70].

Concerning solving GEPs and related computational issues such as the best algorithms regarding speed and accuracy for different structures of interest, we refer to [10], [32], [41], [47], [67], [74], [75] and [87]. Schreiber's thesis [85] provides a review of the available methods for solving NLEVPs, among which PEPs have experienced a surge of research activity in the last few years, motivated by its applications and the development of new theoretical ideas and algorithms.

Chapter 2

Classification of Quasidefinite Matrix Polynomials

2.1 Introduction

The spectral properties of Hermitian matrix polynomials with real eigenvalues have been extensively studied, through classes such as the definite or definitizable pencils, definite, hyperbolic, or quasihyperbolic matrix polynomials, and overdamped or gyroscopically stabilized quadratics [25], [38], [51], [59], [72], [90]. Indeed, we have found inconsistency in the definitions of some of these classes, in particular whether the leading coefficient have to be nonsingular or positive definite. This chapter gives a unified treatment of these and related classes that uses the *eigenvalue type* (or *sign characteristic*) as a common thread. Equivalent conditions are given for each class in a consistent format. We show that these classes form a hierarchy, all of which are contained in the new class of quasidefinite matrix polynomials. As well as collecting and unifying existing results, we make several new contributions. We propose a new characterization of hyperbolicity in terms of the distribution of the eigenvalue types on the real line. We show that definite matrix polynomials can be characterized in terms of the eigenvalue types—something that does not seem well known for definite pencils. We prove interlacing inequalities for the eigenvalues of the submatrices of hyperbolic matrix polynomials of any degree and point out that each hyperbolic quadratic matrix polynomial is strictly isospectral to a real monic tridiagonal quadratic matrix polynomial. By analyzing their effect on eigenvalue type, we show that homogeneous rotations allow results for matrix polynomials with nonsingular or definite leading coefficient to be translated into results with no such requirement on the leading coefficient, which is important for treating definite and quasidefinite polynomials. We use definite linearizations of definite matrix polynomials [38] as another tool in obtaining the relations between the many subclasses of quasidefinite matrix polynomials which allow producing the diagram in Figure 2.1 that provides a diagrammatic summary of most of the results in this chapter. We also give a sufficient and necessary condition for a quasihyperbolic matrix polynomial to be strictly isospectral to a real diagonal quasihyperbolic matrix polynomial of the same degree, and show that this condition is always satisfied in the quadratic case and for any hyperbolic matrix polynomial, thereby identifying an important new class of diagonalizable matrix polynomials.

The chapter is organized as follows. We recall in Section 2.2 the notion of eigenvalue type and show how homogeneous rotations of matrix polynomials as well as linearizations in $\mathbb{H}(P)$ may affect the eigenvalue type. Section 2.3 investigates definite matrix polynomials and their subclasses, while Section 2.4 deals with quasidefinite polynomials and their subclasses. Finally, quasidefinite matrix polynomials that can be diagonalized by structure preserving congruences are identified in Section 2.5 followed by a summary and further discussion in Section 2.6.

2.2 Preliminaries

An $n \times n$ Hermitian matrix polynomial P of degree m as in (1.1) has mn eigenvalues, which are all finite when A_m is nonsingular. Infinite eigenvalues occur when A_m is singular and zero eigenvalues are present when A_0 is singular. Because P is Hermitian, $\sigma(P)$ is symmetric with respect to the real axis.

Since we concentrate here on Hermitian matrix polynomials whose eigenvalues are all real and of definite type, we begin this section with a brief review of the eigenvalue types and sign characteristic (detailed discussions can be found in [30], [31]).

2.2.1 Eigenvalue type and sign characteristic

We start by defining the concept of eigenvalue type. Here $P'(\lambda)$ denotes the first derivative of P with respect to λ .



so that
$$\underbrace{\lambda_{i_m} < \cdots < \lambda_{i_2} < \lambda_{i_1}}_{\text{alternating types}}, i=1:n.$$

Figure 2.1: Quasidefinite $n \times n$ matrix polynomials $P(\lambda) = \sum_{j=0}^{m} \lambda^{j} A_{j}$ of degree m and their subclasses. A subclass **A** pointing to a subclass **B** with a solid line (dotted line) and property "C" means that the subclass **A** with the property "C" is exactly (is contained in) the subclass **B**.

Definition 2.2.1 (positive type/ negative type) Let $P(\lambda)$ be a Hermitian matrix polynomial. A finite real eigenvalue λ_0 of P is of positive type (negative type) if $x^*P'(\lambda_0)x > 0$ ($x^*P'(\lambda_0)x < 0$) for all nonzero $x \in \ker P(\lambda_0)$, respectively.

Thus for an eigenvalue λ_0 of positive type (negative type), the graph of the scalar polynomial $x^*P(\lambda)x$ for any nonzero $x \in \ker P(\lambda_0)$ crosses the x-axis at λ_0 with a positive slope (negative slope). Note that simple eigenvalues are either of positive type or of negative type since for nonzero $x \in \ker P(\lambda_0), x^*P'(\lambda_0)x \neq 0$ [6, Thm. 3.2]. This does not necessarily hold for semisimple eigenvalues: for example the pencil

$$L(\lambda) = \lambda \begin{bmatrix} 1 & 0 \\ 0 & -1 \end{bmatrix} + \begin{bmatrix} -a & 0 \\ 0 & a \end{bmatrix}$$

has a semisimple¹ eigenvalue $\lambda_0 = a$ with corresponding eigenvectors $e_1 = \begin{bmatrix} 1 \\ 0 \end{bmatrix}$ and $e_2 = \begin{bmatrix} 0 \\ 1 \end{bmatrix}$ and it is easily seen that $e_1^*L'(a)e_1 = 1$ and $e_2^*L'(a)e_2 = -1$. The eigenvalue $\lambda_0 = a$ is of mixed type. Note that $x^*L'(\lambda_0)x = 0$ for $x = e_1 + e_2$.

Definition 2.2.2 (definite type/mixed type) A finite real eigenvalue of a Hermitian matrix polynomial is of definite type if it is either of positive type or of negative type. It is of mixed type otherwise.

If λ_0 is a real eigenvalue of P of mixed type then there exist $x, y \in \ker P(\lambda_0)$ such that $x^*P'(\lambda_0)x > 0$ and $y^*P'(\lambda_0)y < 0$. But $x + \alpha y \in \ker P(\lambda_0)$, $\alpha \in \mathbb{C}$ and clearly $(x + \alpha y)^*P'(\lambda_0)(x + \alpha y) = 0$ for some nonzero α (see the previous example).

Lemma 2.2.3 A finite real eigenvalue λ_0 of a Hermitian matrix polynomial is of definite type if and only if $x^*P'(\lambda_0)x \neq 0$ for all nonzero $x \in \ker P(\lambda_0)$.

As shown in [12, Lem. 2.1], eigenvalues of definite type are necessarily semisimple. Indeed, if λ_0 is not semisimple then there is an eigenvector x and a generalized eigenvector y such that $P(\lambda_0)y + P'(\lambda_0)x = 0$. Multiplying on the left by x^* yields

$$x^* \left(P(\lambda_0) y + P'(\lambda_0) x \right) = x^* P'(\lambda_0) x = 0.$$

Hence λ_0 is of mixed type.

¹An eigenvalue of a matrix polynomial $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j$ is semisimple if it appears only in 1×1 Jordan blocks in a Jordan form for P [31].

We will need a notion of eigenvalue type at infinity. To this end use the reversal of P denoted by revP, which is obtained by reversing the order of the coefficient matrices of P:

$$\operatorname{rev} P(\lambda) = \lambda^m P(1/\lambda) = \lambda^m A_0 + \lambda^{m-1} A_1 + \dots + \lambda A_{m-1} + A_m.$$
(2.1)

Note that λ_0 is an eigenvalue of $P(\lambda)$ if and only if $1/\lambda_0$ is an eigenvalue of $\operatorname{rev} P(\lambda)$ with 0 and ∞ regarded as reciprocals. Easy manipulations show that when $\lambda_0 \neq 0$ the equation $(m/\lambda_0)x^*P(\lambda_0)x = 0$ can be rewritten as

$$x^* P'(\lambda_0) x = -\lambda_0^{m-2} x^* (\text{rev}P)'(1/\lambda_0) x.$$
(2.2)

This suggests the following definition.

Definition 2.2.4 (type at ∞) The type of $\lambda_0 = \infty$ as an eigenvalue of a Hermitian matrix polynomial P is given by the type of $1/\lambda_0 = 0$ as an eigenvalue of -revP. In other words, $\lambda_0 = \infty$ is of positive type if $x^*A_{m-1}x < 0$ for every nonzero $x \in \text{ker rev}P(0)$ and of negative type if $x^*A_{m-1}x > 0$ for every nonzero $x \in \text{ker rev}P(0)$.

The notion of eigenvalue type is connected with the more general notion of sign characteristic of a Hermitian matrix polynomial with nonsingular leading term [30], [31]. When all the eigenvalues λ_j of P are real, finite and of definite type, the sign characteristic, for a given ordering $\lambda_1, \ldots, \lambda_{mn}$, is a set of signs $\{\varepsilon_j\}_{j=1}^{mn}$ with $\varepsilon_j = \operatorname{sign}(x_j^* P'(\lambda_j) x_j)$, where x_j is an eigenvector corresponding to λ_j .

We will show in Sections 2.3 and 2.4 that the sign characteristic of definite pencils, overdamped and gyroscopically stabilized quadratics, and hyperbolic and definite polynomials has a particular distribution over the extended real line. Indeed the eigenvalues of these matrix polynomials belong to disjoint intervals, each interval containing eigenvalues of a single type. We say that an interval \mathcal{I} of \mathbb{R} is of positive (negative) type for a matrix polynomial P if every $\lambda \in \sigma(P) \cap \mathcal{I}$ is of positive (negative) type. The interval \mathcal{I} is of definite type if every $\lambda \in \sigma(P) \cap \mathcal{I}$ is of definite type. We also use the wording " ε type" to denote positive type for $\varepsilon > 0$ and negative type for $\varepsilon < 0$.

2.2.2 Motivation

Eigenvalue problems $Ax = \lambda x$, with Hermitian A, have many desirable properties which lead to a variety of special algorithms. Here we consider what can be regarded as the closest analogues of this class of problems for the generalized eigenvalue problem $L(\lambda)x = 0$, with $L(\lambda) = \lambda A - B$, $A = A^*$, $B = B^*$, and for the polynomial eigenvalue problem $P(\lambda)x = 0$, with Hermitian $P(\lambda)$ of degree m. Namely, the classes of definite, definitizable, hyperbolic, quasihyperbolic, overdamped and gyroscopically stabilized eigenproblems [25], [38], [51], [59], [72], [90]. A property common to all these problems is that the eigenvalues are all real and of definite type.

The interest in matrix polynomials with real eigenvalues of definite type comes from systems of differential equations with constant coefficients of the form

$$\sum_{j=0}^{m} i^{j} A_{j} \frac{d^{j} u}{dt^{j}} = 0, \qquad t \in \mathbb{R},$$
(2.3)

where $i = \sqrt{-1}$, $A_j = A_j^* \in \mathbb{C}^{n \times n}$, j = 0; m, and A_m nonsingular. It is known [30, Thm. 13.1.1] that the general solution of (2.3) is given by

$$u(t) = \begin{bmatrix} 0 & \cdots & 0 & I_n \end{bmatrix} e^{-it\mathcal{C}} u_0,$$
 (2.4)

where \mathcal{C} is as in (1.8) and $u_0 \in \mathbb{C}^{nm}$ is arbitrary. The solutions (2.4) are bounded on the half line $[0, \infty)$ if and only if \mathcal{C} , or equivalently $P(\lambda)$, has all its eigenvalues real and semisimple, and these solutions remain bounded under small perturbations of the matrix coefficients A_j of $P(\lambda)$ if and only if the eigenvalues of P are real and of definite type [30, Thm. 13.2.1].

The results presented in this chapter are useful in the solution of the inverse problem of constructing quasidefinite matrix polynomials and their subclasses from given spectral data, as will be shown in Chapter 4.



Figure 2.2: Correspondence between λ and (α, β) .

2.2.3 Homogeneous rotation

We will use the homogenous forms of the matrix polynomial $P(\lambda)$ in (1.1) and the pencil $L(\lambda) = \lambda A - B$, which are given by

$$P(\alpha,\beta) = \sum_{j=0}^{m} \alpha^{j} \beta^{m-j} A_{j}, \qquad L(\alpha,\beta) = \alpha A - \beta B$$

This form is particularly useful when A_m or A is singular or indefinite. An eigenvalue λ is identified with any pair $(\alpha, \beta) \neq (0, 0)$ for which $\lambda = \alpha/\beta$. Note that $P(0, \beta) = \beta^m A_0$ so that $\lambda = 0$ represented by $(0, \beta)$ is an eigenvalue of P if and only if A_0 is singular. Similarly, $\lambda = \infty$ represented by $(\alpha, 0)$ is an eigenvalue of P if and only if A_m is singular. Without loss of generality we can take $\alpha^2 + \beta^2 = 1$. We then have a direct correspondence between eigenvalues on the extended real line $\mathbb{R} \cup \{\infty\}$ and the unit circle (see Figure 2.2). Note the two copies of $\mathbb{R} \cup \{\infty\}$, represented by the upper semicircle and the lower semicircle.

The matrix polynomial $\widetilde{P}(\widetilde{\alpha}, \widetilde{\beta})$ is obtained from $P(\alpha, \beta)$ by homogenous rotation if $\left[\alpha\right] \left[c \quad s\right] \left[\alpha\right] \quad \left[\widetilde{\alpha}\right]$

$$G\begin{bmatrix}\alpha\\\beta\end{bmatrix} = \begin{bmatrix}c & s\\-s & c\end{bmatrix}\begin{bmatrix}\alpha\\\beta\end{bmatrix} =: \begin{bmatrix}\widetilde{\alpha}\\\widetilde{\beta}\end{bmatrix}, \quad c, s \in \mathbb{R}, \quad c^2 + s^2 = 1 \quad (2.5)$$

and

$$P(\alpha,\beta) = \sum_{j=0}^{m} (c\widetilde{\alpha} - s\widetilde{\beta})^{j} (s\widetilde{\alpha} + c\widetilde{\beta})^{m-j} A_{j} =: \sum_{j=0}^{m} \widetilde{\alpha}^{j} \widetilde{\beta}^{m-j} \widetilde{A}_{j} =: \widetilde{P}(\widetilde{\alpha},\widetilde{\beta}).$$
(2.6)

Homogeneous rotations can be seen as an analogue of translations of λ in the nonhomogeneous case: one common feature is that they both preserve the eigenvectors. Indeed P and \tilde{P} have the same eigenvectors but the corresponding eigenvalues are rotated. On using $P(\alpha, \beta) = \tilde{P}(\tilde{\alpha}, \tilde{\beta})$, the binomial expansion theorem leads to an expression for each \tilde{A}_j . In particular we find that

$$\widetilde{A}_{m} = P(c,s),
\widetilde{A}_{m-1} = \sum_{j=0}^{m} \left(-jc^{j-1}s^{m+1-j} + (m-j)c^{j+1}s^{m-j-1} \right) A_{j},$$

$$\widetilde{A}_{0} = P(-s,c).$$
(2.7)

We will use homogeneous rotations to transform a polynomial P with singular or indefinite leading coefficient A_m to a polynomial \tilde{P} with nonsingular or positive definite leading coefficient $\tilde{A}_m = P(c, s)$, which we can do provided that a pair (c, s)on the unit circle is known such that $\det(P(c, s)) \neq 0$ or P(c, s) > 0, respectively (see Example 2.2.5).

Example 2.2.5 The pencil

$$L(\lambda) = \lambda \operatorname{diag}(1, 1, -1) - \operatorname{diag}(2, 3, -5) =: \lambda A - B$$

has indefinite leading coefficient matrix A. Note that for $\mu = 4$, $L(\mu) = \text{diag}(2, 1, 1) > 0$. We homogeneously rotate L into \tilde{L} so that μ corresponds to ∞ . This is achieved by taking $c = \mu/\sqrt{\mu^2 + 1}$ and $s = 1/\sqrt{\mu^2 + 1}$ in (2.5). Then G rotates $L(\lambda)$ into $\tilde{L}(\tilde{\lambda}) =: \tilde{\lambda}\tilde{A} - \tilde{B}$, where $\tilde{A} = L(\mu)/\sqrt{\mu^2 + 1} > 0$. Note that L has eigenvalues 2 and 3 of positive type and eigenvalue 5 of negative type. These eigenvalues are rotated to 4.5, 13 and -21, respectively, all of positive type since \tilde{A} is positive definite.

Example 2.2.5 shows that homogeneous rotation does not preserve the eigenvalue types, but as the next lemma shows it always preserves definite type. Now, we study the effects of homogeneous rotation and linearization on the eigenvalue types. To avoid ambiguity, P'_{λ} denotes the first derivative of P with respect to the variable λ .

Lemma 2.2.6 Let \widetilde{P} of degree m be obtained from P by homogeneous rotation (2.5). Let the real numbers $\lambda_0 = \frac{\alpha_0}{\beta_0}$ and $\widetilde{\lambda}_0 = \frac{\widetilde{\alpha}_0}{\widetilde{\beta}_0}$ with $\begin{bmatrix} \widetilde{\alpha}_0 \\ \widetilde{\beta}_0 \end{bmatrix} = G\begin{bmatrix} \alpha_0 \\ \beta_0 \end{bmatrix}$ be eigenvalues of P and \tilde{P} , respectively, with corresponding eigenvector x.

(i) If λ_0 and $\widetilde{\lambda}_0$ are both real and finite then $c - \lambda_0 s \neq 0$ and

$$x^* P'_{\lambda}(\lambda_0) x = (c - \lambda_0 s)^{m-2} x^* \widetilde{P}'_{\widetilde{\lambda}}(\widetilde{\lambda}_0) x$$

(ii) If λ_0 is real and finite and $\widetilde{\lambda}_0 = \infty$ then $s \neq 0$ and

$$x^* P'_{\lambda}(\lambda_0) x = s^{2-m} x^* \big(-(\operatorname{rev} \widetilde{P})'_{\widetilde{\lambda}}(0) \big) x.$$

(iii) If $\lambda_0 = \infty$ and $\widetilde{\lambda}_0$ is real and finite then $s \neq 0$ and

$$x^* \big(-(\operatorname{rev} P)'_{\lambda}(0) \big) x = (-s)^{m-2} x^* \widetilde{P}'_{\widetilde{\lambda}}(\widetilde{\lambda}_0) x.$$

Proof. (i) Note that $\begin{bmatrix} \tilde{\alpha} \\ \tilde{\beta} \end{bmatrix} = G \begin{bmatrix} \alpha \\ \beta \end{bmatrix}$ where $(\alpha, \beta) \neq (0, 0)$ and $(\tilde{\alpha}, \tilde{\beta}) \neq (0, 0)$ represent λ and $\tilde{\lambda}$ respectively, that is $\lambda = \alpha/\beta$ and $\tilde{\lambda} = \tilde{\alpha}/\tilde{\beta}$. When λ and $\tilde{\lambda}$ are finite, $\beta \neq 0$ and $\tilde{\beta} = \beta c - \alpha s \neq 0$ hence $c - \lambda s \neq 0$. It follows from (2.6) that $\beta^m P(\lambda) = \tilde{\beta}^m \tilde{P}(\tilde{\lambda}) = (\beta c - \alpha s)^m \tilde{P}(\tilde{\lambda})$ so that

$$P(\lambda) = (c - \lambda s)^m \widetilde{P}(g(\lambda)), \quad g(\lambda) = \frac{\lambda c + s}{c - \lambda s} = \widetilde{\lambda}.$$
(2.8)

Differentiating with respect to λ yields

$$P_{\lambda}'(\lambda) = -ms(c - \lambda s)^{m-1}\widetilde{P}(g(\lambda)) + (c - \lambda s)^{m-2}\widetilde{P}_{\widetilde{\lambda}}'(g(\lambda)).$$
(2.9)

Multiplying (2.9) on the left by x^* and on the right by x, evaluating at λ_0 and using $\widetilde{\lambda}_0 = g(\lambda_0)$ and $\widetilde{P}(\widetilde{\lambda}_0)x = 0$ yield the desired result.

(ii) When λ_0 is finite, $\beta_0 \neq 0$ and $\tilde{\lambda}_0$ infinite implies that $\tilde{\beta}_0 = \beta_0 c - \alpha_0 s = 0$, that is, $c = s\lambda_0$ with $s \neq 0$ since $c^2 + s^2 = 1$. Using (2.7) we obtain

$$s^{-m}\widetilde{A}_{m-1} = \sum_{j=0}^{m} \left(-j\lambda_0^{j-1} + (m-j)\lambda_0^{j+1} \right) A_j.$$
(2.10)

Multiplying (2.10) on the left by x^* and on the right by x we find that

• if $\lambda_0 \neq 0$ then

$$s^{-m}x^{*}\widetilde{A}_{m-1}x = -x^{*}P_{\lambda}'(\lambda_{0})x + \lambda_{0}^{m}x^{*}\left(A_{m-1} + \dots + \frac{m-1}{\lambda_{0}^{m-2}}A_{1} + \frac{m}{\lambda_{0}^{m-1}}A_{0}\right)x$$

$$= -x^{*}P_{\lambda}'(\lambda_{0})x + \lambda_{0}^{m}x^{*}(\operatorname{rev} P)_{\lambda}'(1/\lambda_{0})x$$

$$= -x^{*}P_{\lambda}'(\lambda_{0})x - \lambda_{0}^{2}x^{*}P_{\lambda}'(\lambda_{0})x,$$

where we used (2.2) for the last equality. The relation in (ii) follows on noting that $1 + \lambda_0^2 = s^{-2}$ and $x^* ((\operatorname{rev} \widetilde{P})'_{\widetilde{\lambda}}(0)) x = x^* \widetilde{A}_{m-1} x$.

• if $\lambda_0 = 0$ then $c = 0, s = \pm 1$ and

$$s^{-m}x^*\widetilde{A}_{m-1}x = -x^*A_1x$$

which is the relation in (ii) since $s = \pm 1$.

(iii) Suppose that G rotates $\lambda_0 = \infty$ to a finite eigenvalue $\tilde{\lambda}_0$ then G^{-1} rotates $\tilde{\lambda}_0$ to $\lambda_0 = \infty$ and we can apply (ii) to obtain the desired result.

2.2.4 Hermitian linearizations

Higham et al. in [37] show that

$$\mathbb{H}(P) := \left\{ L(\lambda) = \lambda A - B \in \mathbb{L}_1(P) : A^* = A, B^* = B \right\}$$
$$= \left\{ \sum_{j=1}^m v_j L_j(\lambda), v \in \mathbb{R}^m \right\},$$

where $L_j(\lambda) = \lambda B_j - B_{j-1}$ and the matrices B_j are a direct sum of block Hankel matrices (see [37, (3.6)–(3.8)]). The pencil $\lambda B_j - B_{j-1} \in \mathbb{H}(P)$ with vector $v = e_j$, where e_j denotes the *j*th column of the identity matrix, is referred to as the *j*th basis pencil. When A_m is nonsingular,

$$\lambda B_m - B_{m-1} \coloneqq \lambda \begin{bmatrix} & A_m \\ & \ddots & A_{m-1} \\ & \ddots & \ddots & \vdots \\ & \ddots & \ddots & A_2 \\ A_m A_{m-1} \dots & A_2 & A_1 \end{bmatrix} - \begin{bmatrix} & A_m \\ & \ddots & A_{m-1} \\ & \ddots & \ddots & \vdots \\ A_m A_{m-1} \dots & A_2 & \\ & & -A_0 \end{bmatrix}$$
(2.11)

is always a linearization of P by the Eigenvalue Exclusion Theorem, Theorem 1.2.2. Note that $\mathbb{H}(P) \subset \mathbb{DL}(P)$.

Any linearization $L(\lambda) \in \mathbb{H}(P)$ with vector v has the property that x is a right eigenvector of P associated to the eigenvalue λ if and only if $\Lambda \otimes x$ (if λ is finite) or $e_1 \otimes x$ (if $\lambda = \infty$) is a right eigenvector for L with eigenvalue λ .

The following result relates the type of a real eigenvalue λ of P to its type as an eigenvalue of a linearization $L \in \mathbb{H}(P)$ and shows that linearizations in $\mathbb{H}(P)$ preserve definite type.

Lemma 2.2.7 Let (λ_0, x) be an eigenpair of P with $\lambda_0 \in \mathbb{R}$ and let $L(\lambda) \in \mathbb{H}(P)$ with vector v be a linearization of P. Then,

$$z^*L'(\lambda_0)z = \mathsf{p}(\lambda_0; v) \cdot x^*P'(\lambda_0)x, \qquad (2.12)$$

where $z = \Lambda_0 \otimes x$. Hence a real eigenvalue λ_0 of L is of definite type if and only if λ_0 as an eigenvalue of P is of definite type. Moreover, if $P(\lambda) = \sum_{j=0}^m \lambda^j A_j$ with A_m nonsingular then $\lambda B_m - B_{m-1}$ in (2.11) is a linearization of P that preserves the type of the real eigenvalues.

Proof. How to obtain (2.12) can be found in [39, Section 3]. Now if $L \in \mathbb{H}(P)$ is a linearization of P then by Theorem 1.2.2, $\mathbf{p}(\lambda_0; v) \neq 0$. Hence $z^*L'(\lambda_0)z \neq 0$ if and only if $x^*P'(\lambda_0)x \neq 0$. The pencil in (2.11) is in $\mathbb{H}(P)$ with vector $v = e_m$ so $\mathbf{p}(\lambda_0; e_m) = 1$. It is a linearization of P when A_m is nonsingular.

2.3 Definite matrix polynomials

The class of definite matrix polynomials (defined in Section 2.3.4) has recently been introduced and investigated by Higham, Mackey and Tisseur [38]. It includes definite pencils, hyperbolic matrix polynomials, and overdamped quadratics. We review these subclasses in the following way: for each subclass we provide a list of equivalent properties, named consistently according to

- (P1) the distribution of the eigenvalue type on the real line,
- (P2) certain definiteness properties,

(P3) the roots of the scalar equations $x^*P(\lambda)x = 0$ (or $x^*L(\lambda)x = 0$ for pencils).

Each subclass has extra equivalent properties listed, either because the property is commonly used to define the subclass or because the property is relevant to the rest of the chapter. We do not claim to provide a full list of characterizations.

2.3.1 Definite pencils

We start with definite pencils, whose occurrence is frequent in applications in science and engineering (see [21, Chap. 9], [46] for examples).

Definition 2.3.1 (definite pencils) An $n \times n$ Hermitian pencil is definite if it satisfies any one (and hence all) of the equivalent properties of Theorem 2.3.2.

Theorem 2.3.2 For an $n \times n$ Hermitian pencil $L(\lambda) = \lambda A - B$ the following are equivalent:

- (P1) $\sigma(L) \subset \mathbb{R} \cup \{\infty\}$ with all eigenvalues of definite type and where the eigenvalues of positive type are separated from the eigenvalues of negative type (see Figure 2.3).
- (P2) $L(\mu)$ is a definite matrix for some $\mu \in \mathbb{R} \cup \{\infty\}$, or equivalently $L(\alpha, \beta) > 0$ for some (α, β) on the unit circle.
- (P3) For every nonzero $x \in \mathbb{C}^n$, the scalar equation $x^*L(\lambda)x = 0$ has exactly one zero in $\mathbb{R} \cup \{\infty\}$.
- (P4) $(x^*Ax, x^*Bx) \neq 0$ for all nonzero $x \in \mathbb{C}^n$.
- (D) There exists a nonsingular $X \in \mathbb{C}^{n \times n}$ such that

$$X^*L(\lambda)X = \begin{bmatrix} L_+(\lambda) & 0\\ 0 & L_-(\lambda) \end{bmatrix},$$

where $L_{+}(\lambda) = \lambda D_{+} - J_{+}$ and $L_{-}(\lambda) = \lambda D_{-} - J_{-}$ are real diagonal pencils, such that $[\lambda_{\min}(L_{+}), \lambda_{\max}(L_{+})] \cap [\lambda_{\min}(L_{-}), \lambda_{\max}(L_{-})] = \emptyset$, D_{+} has nonnegative entries, D_{-} has nonpositive entries and if $(D_{+})_{ii} = 0$ then $(J_{+})_{ii} > 0$ or if $(D_{-})_{ii} = 0$ then $(J_{-})_{ii} < 0$.



Figure 2.3: Distribution of the eigenvalue types for three types of definite pencils $L(\lambda) = \lambda A - B$. On the shaded intervals L is indefinite.

Proof. The proof of (P2) \Leftrightarrow (P4) can be found in [88, Thm. 6.1.18] and the equivalence (P3) \Leftrightarrow (P4) is immediate. We show that (P2) \Rightarrow (P1) \Rightarrow (D) \Rightarrow (P2).

(P2) \Rightarrow (P1): Suppose L(c, s) > 0 for some $c, s \in \mathbb{R}$, $c^2 + s^2 = 1$. If s = 0 then A is definite so that all the eigenvalues belong to one interval of either positive type if A > 0 or negative type if A < 0 since for all eigenpairs (λ, x) , $x^*L'(\lambda)x = x^*Ax$ is either positive or negative. Assume without loss of generality that s > 0 and homogeneously rotate L into $\widetilde{L}(\widetilde{\lambda}) = \widetilde{\lambda}\widetilde{A} - \widetilde{B}$ as in Example 2.2.5 so that $\widetilde{A} > 0$. Hence all the eigenvalues of \widetilde{L} are real and of positive type. Let λ_j be an eigenvalue of L rotated to $\widetilde{\lambda}_j$. By Lemma 2.2.6, their types are related by

$$x^* \widetilde{L}'_{\widetilde{\lambda}}(\widetilde{\lambda}_j) x = \begin{cases} (c - \lambda_j s) x^* L'_{\lambda}(\lambda_j) x & \text{if } \lambda_j \text{ is finite,} \\ -s x^* \big(-(\operatorname{rev} L)'_{\lambda}(0) \big) x & \text{if } \lambda_j = \infty. \end{cases}$$
(2.13)

Note that

$$c - \lambda_j s = \det \begin{bmatrix} c & \lambda_j \\ s & 1 \end{bmatrix}$$
(2.14)

and the sign of these determinants is positive for any $\lambda_j = (\lambda_j, 1)$ that lies counterclockwise from (c, s), and negative for any that lies clockwise from (c, s); see Figure 2.4. Hence it follows from (2.13) that eigenvalues of L lying clockwise from (c, s)(including $+\infty$) are of negative type and eigenvalues of L lying counterclockwise from (c, s) are of positive type. Also, there is a gap between the two types because c/s is not an eigenvalue of L.

 $(P1) \Rightarrow (D)$: Recall that a Hermitian pencil is diagonalizable by congruence if and only if its eigenvalues belong to $\mathbb{R} \cup \{\infty\}$ and are semisimple [56]. Since eigenvalues of definite type are semisimple, there exists X nonsingular such that $X^*L(\lambda)X = \lambda D - J$, with D and J both real and diagonal. Their diagonal entries can be reordered so that $D = D_+ \oplus D_-$ and $J = J_+ \oplus J_-$, where the eigenvalues of $\lambda D_+ - J_+ = L_+(\lambda)$



Figure 2.4: Pictorial representation of (2.14).

are of positive type and that of $\lambda D_- - J_- = L_-(\lambda)$ are of negative type. If A is singular then one of D_+ or D_- (but not both otherwise $\lambda = \infty$ would be of mixed type) must be singular. Hence D_+ and $-D_-$ have nonnegative entries. Each zero entry on D corresponds to an infinite eigenvalue. By Definition 2.2.4 when $(D_+)_{ii} = 0$ we must have $(J_+)_{ii} > 0$ for $\lambda = \infty$ to be of positive type and when $(D_-)_{ii} = 0$ then $(J_-)_{ii} < 0$. Finally because the eigenvalues of positive type are separated from the eigenvalues of negative type, the intersection between $[\lambda_{\min}(L_+), \lambda_{\max}(L_+)]$ and $[\lambda_{\min}(L_-), \lambda_{\max}(L_-)]$ must be empty.

 $(D) \Rightarrow (P2)$: It follows from (D) that

$$\begin{split} L_{+}(\mu) &< 0 \text{ for } \mu < \lambda_{\min}(L_{+}), \qquad L_{+}(\mu) > 0 \text{ for } \mu > \lambda_{\max}(L_{+}) \text{ if } D_{+} \text{ is nonsingular,} \\ L_{-}(\mu) &> 0 \text{ for } \mu < \lambda_{\min}(L_{-}), \qquad L_{-}(\mu) < 0 \text{ for } \mu > \lambda_{\max}(L_{-}) \text{ if } D_{-} \text{ is nonsingular.} \end{split}$$

Hence

- (i) if L_{-} is void then $L(\mu) < 0$ for $\mu < \lambda_{\min}(L_{+})$,
- (ii) if L_+ is void then $L(\mu) > 0$ for $\mu < \lambda_{\min}(L_-)$,
- (iii) if $\lambda_{\max}(L_+) < \lambda_{\min}(L_-)$ then $L(\mu) > 0$ for $\lambda_{\max}(L_+) < \mu < \lambda_{\min}(L_-)$
- (iv) if $\lambda_{\max}(L_{-}) < \lambda_{\min}(L_{+})$ then $L(\mu) < 0$ for $\lambda_{\max}(L_{-}) < \mu < \lambda_{\min}(L_{+})$,

which completes the proof. \blacksquare

Characterizations (P2) and (P4) in Theorem 2.3.2 are commonly used as definitions of definite pencils. In (P2), $\mu = \infty$ is allowed and $L(\infty)$ definite means that A is definite. Note that (P4) is equivalent to saying that 0 is not in the field of values
of A + iB or that the Crawford number

$$\gamma(A,B) = \min_{\substack{z \in \mathbb{C}^n \\ z^*z=1}} \sqrt{(z^*Az)^2 + (z^*Bz)^2}$$
(2.15)

is strictly positive. Finally we remark that in property (D) all the eigenvalues of L_+ are of positive type and those of L_- are of negative type.

Pencils $L(\lambda) = \lambda A - B$ with A > 0 have computational advantages: the eigenvalues can be computed by methods that exploit the definiteness of A [22]. When A and B are both indefinite, characterization (P1) offers an easy way to check definiteness, but it is computationally unattractive since it requires all the eigenpairs. As an alternative, the recently improved arc algorithm of Crawford and Moon [19], [34] efficiently detects whether $\lambda A - B$ is definite and determines μ such that $L(\mu) > 0$ at the cost of just a few Cholesky factorizations. The pencil can then be rotated to a pencil with positive definite leading term as in Example 2.2.5. An earlier level set algorithm of Higham, Tisseur, and Van Dooren [40, Alg. 2.3] is more expensive as it requires the solution of a complex QEP with a zero linear term for the eigenvalues and, possibly, eigenvectors.

2.3.2 Hyperbolic matrix polynomials

Hyperbolic matrix polynomials generalize definite pencils $\lambda A - B$ with A > 0.

Definition 2.3.3 (hyperbolic matrix polynomial) A Hermitian matrix polynomial is hyperbolic if it satisfies any one (and hence all) of the equivalent properties of Theorem 2.3.4.

Theorem 2.3.4 For an $n \times n$ Hermitian matrix polynomial $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j$ the following are equivalent:

(P1) All eigenvalues of P are real and finite, of definite type, and such that

$$\underbrace{\lambda_{mn} \leq \cdots \leq \lambda_{(m-1)n+1}}_{(-1)^{m-1} \ type} < \cdots < \underbrace{\lambda_{2n} \leq \cdots \leq \lambda_{n+1}}_{negative \ type} < \underbrace{\lambda_n \leq \cdots \leq \lambda_1}_{positive \ type},$$

where $\sigma(P) = \{\lambda_j\}_{j=1}^{mn}$ and " $(-1)^{m-1}$ type" denotes positive type for odd m and negative type for even m.



Figure 2.5: Distribution of eigenvalue types of $n \times n$ hyperbolic polynomials of even degree m.

(P2) There exist $\mu_j \in \mathbb{R} \cup \{\infty\}$ such that

$$(-1)^{j} P(\mu_{j}) > 0, \qquad j = 0: m - 1, \qquad \infty = \mu_{0} > \mu_{1} > \mu_{2} > \dots > \mu_{m-1}.$$

- (P3) $A_m > 0$ and for every nonzero $x \in \mathbb{C}^n$, the scalar equation $x^*P(\lambda)x = 0$ has m distinct real and finite zeros.
 - (L) P has a definite linearization $L(\lambda) \in \mathbb{H}(P)$ with vector $v \in \mathbb{R}^m$, where $v_1 \neq 0$, such that $L(\infty) > 0$ if $v_1 > 0$ and $L(\infty) < 0$ if $v_1 < 0$.

Proof. That (P2) \Leftrightarrow (P3) and (P2) \Rightarrow (P1) is due to Markus [72, Section 31]. We show that (P1) \Rightarrow (L) and (L) \Rightarrow (P3).

 $(P1) \Rightarrow (L)$: If m = 1 and $L \in \mathbb{H}(P)$ then $L(\lambda) = vP(\lambda)$ and L is a linearization if $v \neq 0$. By property (P1) of Theorem 2.3.2, $P(\lambda)$ is a definite pencil. Since all the eigenvalues are of positive type, property (D) of Theorem 2.3.2 implies that the leading coefficient of P is positive definite, i.e., $P(\infty) > 0$. Hence property (L) holds.

Now assume that m > 1. Let $v \in \mathbb{R}^m$ be such that the roots μ_j of the v-polynomial in (1.10) satisfy $\lambda_{jn} > \mu_j > \lambda_{jn+1}$, j = 1: m-1. Then by Theorem 1.2.2, $L(\lambda) \in \mathbb{H}(P)$ with vector v is a linearization of P. By construction, $\mathbf{p}(x;v) = v_1 \prod_{j=1}^{m-1} (x - \mu_j)$ with $v_1 \neq 0$ since all roots of $\mathbf{p}(x;v)$ are finite and $\operatorname{sign}(\mathbf{p}(\lambda_k;v)) = (-1)^{j-1}\operatorname{sign}(v_1)$ for $(j-1)n+1 \leq k \leq jn$, j = 1: m-1 (see Figure 2.5). By Lemma 2.2.7 we have that for each eigenpair (λ_k, x_k) of P, $z_k^*L'(\lambda_k)z_k = \mathbf{p}(\lambda_k;v) \cdot x_k^*P'(\lambda_k)x_k$, where $z_k = [\lambda_k^{m-1}, \lambda_k^{m-2}, \dots, 1]^T \otimes x_k$ is an eigenvector of L with eigenvalue λ_k . Hence all eigenvalues of L are of positive type when $v_1 > 0$ and of negative type when $v_1 < 0$. Now properties (P1) and (D) of Theorem 2.3.2 imply that L is a definite pencil with $L(\infty) > 0$ if $v_1 > 0$ and $L(\infty) < 0$ if $v_1 < 0$. (L) \Rightarrow (P3): If P has a definite linearization then by [38, Thm. 4.1], there exists $\mu \in \mathbb{R} \cup \{\infty\}$ such that $P(\mu)$ is definite and for every nonzero $x \in \mathbb{C}^n$ the scalar equation $x^*P(\lambda)x = 0$ has m distinct zeros in $\mathbb{R} \cup \{\infty\}$. By [38, Thm. 4.2], $L(\infty)$ definite implies $P(\infty)$ definite and [38, (4.7)] shows that L(1,0) is congruent to a block diagonal form whose (1,1) block is $v_1P(1,0)$. Now if $v_1 > 0$ and L(1,0) > 0 or $v_1 < 0$ and L(1,0) < 0 then $A_m = P(1,0) > 0$. Hence $x^*A_mx \neq 0$ and $x^*P(\lambda)x = 0$ has m distinct real finite zeros.

Property (P3) is used by Gohberg, Lancaster, and Rodman [30, Section 13.4] as the definition of hyperbolicity. Characterization (P1) is stated for the quadratic case and without proof in [48, Section 9] and [59].

It is shown in [80, Cor. 2] that for a Hermitian triple (A, B, C),

$$(x^*Ax, x^*Bx, x^*Cx) \neq 0$$

for all nonzero $x \in \mathbb{C}^n$ if and only if there exist $\alpha, \beta, \gamma \in \mathbb{R}$ such that $\alpha A + \beta B + \gamma C > 0$. So in view of (P2) in Theorem 2.3.4, the natural extension to quadratics of property (P4) for definite pencils in Theorem 2.3.2, i.e., $(x^*A_2x, x^*A_1x, x^*A_0x) \neq 0$ for all nonzero $x \in \mathbb{C}^n$, is a necessary condition for hyperbolicity (since $P(\mu) > 0$ for sufficiently large μ), but it is not sufficient: the quadratic $Q(\lambda) = \lambda^2 I_n + \lambda I_n + I_n$ is not hyperbolic since it has nonreal eigenvalues but $(x^*x, x^*x, x^*x) \neq 0$ for all nonzero $x \in \mathbb{C}^n$.

As for definite pencils, property (P1) is easy to check once all the eigenpairs of *P* are known. Guo and Lancaster's algorithm [35] tests for hyperbolicity by computing the eigenvalues of the QEP. More efficient approaches to check whether a quadratic matrix polynomial is hyperbolic or not include Guo, Higham and Tisseur's quadratically convergent matrix iteration based on cyclic reduction [33] or Guo, Higham and Tisseur's adaptation of the arc algorithm [34, Section 4.1]. For matrix polynomials of arbitrary degree, Niendorf and Voss [78] propose an algorithm that checks for hyperbolicity and which is based on a minmax and maxmin characterization of the eigenvalues.

Let $Q(\lambda)$ be a Hermitian quadratic matrix polynomial with $n \times n$ coefficient matrices and let $\widehat{Q}(\lambda)$ be the matrix polynomial obtained by deleting the last row and column of $Q(\lambda)$. In what follows we characterize hyperbolic quadratic matrix polynomials according to the distribution of the eigenvalues of both Q and \hat{Q} . Then we recall a recent result by Azizov et al. [9, Thm. 5.1] which implies that hyperbolic quadratics are always *tridiagonalizable*. We make use of this fact later in Chapter 4.

Theorem 2.3.5 Let $Q(\lambda)$ be a Hermitian quadratic $n \times n$ matrix polynomial with positive definite leading coefficient and let $\mu_{2n-2} \leq \cdots \leq \mu_1$, $\lambda_{2n} \leq \cdots \leq \lambda_1$ be the eigenvalues of $\widehat{Q}(\lambda)$ and $Q(\lambda)$, respectively. Then $Q(\lambda)$ is hyperbolic if and only if $\widehat{Q}(\zeta) < 0$ for some $\zeta \in (\lambda_{n+1}, \lambda_n)$ and

1. $\lambda_{j+1} \le \mu_j \le \lambda_j$, for j = 1: n - 1,

2.
$$\lambda_{j+2} \le \mu_j \le \lambda_{j+1}$$
, for $j = n: 2n - 2$,

3. $\lambda_{n+1} < \lambda_n$.

Proof. (\Rightarrow) : If Q is hyperbolic then [79, Lem. 7.1, Thm. 7.3] show that \widehat{Q} is hyperbolic with $\widehat{Q}(\zeta) < 0$ for some $\zeta \in (\lambda_{n+1}, \lambda_n)$. Veselić in [91] shows that the eigenvalues of Q and \widehat{Q} are distributed as in the statement of the theorem.

 (\Leftarrow) : Let $\widehat{Q}(\zeta)$ be negative definite for some $\zeta \in (\lambda_{n+1}, \lambda_n)$ and let the eigenvalues of Q and \widehat{Q} be distributed as in the statement of the theorem. The inclusion principle [42, Thm. 4.3.15] implies that the n-1 smallest eigenvalues of the constant matrix $Q(\zeta)$ are negative. But $\det(Q(\zeta)) = c \prod_{j=1}^{2n} (\zeta - \lambda_j)$, where c is a positive constant since the leading coefficient matrix of Q is positive definite. Since $\zeta \in (\lambda_{n+1}, \lambda_n)$, the sign of $\det(Q(\zeta))$ is $(-1)^n$. Thus, the largest eigenvalue of $Q(\zeta)$ is negative and $Q(\zeta) < 0$ which is equivalent to saying that Q is hyperbolic, see Theorem 2.3.4.



Figure 2.6: The eigenvalues interlacing of a 3×3 quadratic matrix polynomial Q and its leading 2×2 submatrix \hat{Q}

Towards extending Theorem 2.3.5 to higher degrees, let $P(\lambda)$ be a hyperbolic matrix polynomial of the form (1.1). Let $\lambda_1(x) > \lambda_2(x) > \cdots > \lambda_m(x)$ be the roots of $x^*P(\lambda)x$ for some nonzero $x \in \mathbb{C}^n$ and recall the following result. **Theorem 2.3.6** [38] A hyperbolic $P(\lambda)$ that is $n \times n$ of degree $m \ge 2$ with eigenvalues

$$\lambda_{mn} \leq \cdots \leq \lambda_1,$$

has the properties displayed in the following diagram, where $\mathcal{I}_i = \{\lambda_i(x):x \in \mathbb{C}^n, \|x\|_2 = 1\}$ and $P(\lambda)$ is indefinite on these intervals.



Figure 2.7: Definiteness diagram

A direct consequence is what follows.

Corollary 2.3.7 If an $n \times n$ matrix polynomial $P(\lambda)$ of degree $m \ge 2$ is hyperbolic with eigenvalues $\lambda_{mn} \le \cdots \le \lambda_1$, then for each $j, 1 \le j \le m$,

$$\Pi_{(-1)^{j-1}}(P(\mu)) = jn - i, \quad i = 1 + (j-1)n; jn,$$
(2.16)

for all $\mu \in (\lambda_{i+1}, \lambda_i)$, i = 1: mn - 1, where $\Pi_+(P(\mu))$ $(\Pi_-(P(\mu)))$ is the number of positive (negative) eigenvalues of the matrix $P(\mu)$.

Next, we show that the result in [91] extends to hyperbolic matrix polynomials of any degree.

Lemma 2.3.8 If an $n \times n$ matrix polynomial $P(\lambda)$ of degree $m \geq 2$ is hyperbolic and $\widehat{P}(\lambda)$ is the matrix polynomial obtained by deleting the last row and column of $P(\lambda)$, then $\widehat{P}(\lambda)$ is hyperbolic with eigenvalues $\mu_{m(n-1)} \leq \cdots \leq \mu_1$, for which

$$\lambda_{i+j} \le \mu_i \le \lambda_{i+j-1}, \quad i = (j-1)(n-1) + 1; j(n-1), \quad j = 1; m,$$
 (2.17)

where $\lambda_{mn} \leq \cdots \leq \lambda_1$, are the eigenvalues of $P(\lambda)$.

Proof. The proof is exactly the same as the proof of the special case of hyperbolic quadratics [91]. In fact, the first part of that proof can be avoided using the definiteness diagram which proves (2.16). Now, let

$$P(\lambda) = \begin{bmatrix} \widehat{P}(\lambda) & P_{12}(\lambda) \\ P_{12}^*(\lambda) & p_{22}(\lambda) \end{bmatrix},$$
(2.18)

be a partition of $P(\lambda)$, then Markus' characterization of hyperbolicity (i.e., (P2) \Leftrightarrow (P3) in Theorem 2.3.4) implies that $\hat{P}(\lambda)$ is hyperbolic too. To prove (2.17) for j = 1, suppose on the contrary that $\mu_i > \lambda_i$ or $\mu_i < \lambda_{i+1}$ for some i = 1: n - 1. Assume first that there is a $\lambda \notin \sigma(\hat{P})$, with $\mu_i < \lambda < \lambda_{i+1}$. Now

$$P(\lambda) = W \begin{bmatrix} \widehat{P}(\lambda) & 0\\ 0 & p_{22}(\lambda) - P_{12}^*(\lambda)\widehat{P}(\lambda)^{-1}P_{12}(\lambda) \end{bmatrix} W^*,$$

with

$$W = \begin{bmatrix} I_{n-1} & 0\\ P_{12}^*(\lambda)\widehat{P}(\lambda)^{-1} & 1 \end{bmatrix}$$

By Sylvester's law of inertia,

$$\Pi_{+}(\widehat{P}(\lambda)) \le \Pi_{+}(P(\lambda)) \le \Pi_{+}(\widehat{P}(\lambda)) + 1.$$
(2.19)

So, $\mu_i < \lambda < \lambda_{i+1}$ would imply

$$\Pi_+(\widehat{P}(\lambda)) \ge n-i, \quad \Pi_+(P(\lambda)) \le n-i-1,$$

which contradicts the first inequality in (2.19). Similarly, $\lambda_i < \lambda < \mu_i$ implies

$$\Pi_{+}(P(\lambda)) \ge n - i + 1, \quad \Pi_{+}(\widehat{P}(\lambda)) \le n - i - 1,$$

which contradicts the second inequality in (2.19). This proves (2.17) for j = 1. The proof of (2.17) is completely analogous for j > 1.

Now, we are in position to extend Theorem 2.3.5 for degrees greater than two.

Theorem 2.3.9 Let $P(\lambda)$ be a Hermitian $n \times n$ matrix polynomial of degree $m \geq 2$ with positive definite leading coefficient and let $\mu_{m(n-1)} \leq \cdots \leq \mu_1$, $\lambda_{mn} \leq \cdots \leq \lambda_1$ be the eigenvalues of $\widehat{P}(\lambda)$ and $P(\lambda)$, respectively. Then $P(\lambda)$ is hyperbolic if and only if there exist $\zeta_i \in \mathbb{R}$ such that $(-1)^i \widehat{P}(\zeta_i) > 0$, $\zeta_i \in (\lambda_{in+1}, \lambda_{in})$, i = 1: m - 1 and (2.17) holds.

Proof. (\Rightarrow): If P is hyperbolic then Markus characterization of hyperbolicity and partition (2.18) show that \widehat{P} is hyperbolic with $(-1)^i \widehat{P}(\zeta_i) > 0$, $\zeta_i \in (\lambda_{in+1}, \lambda_{in})$, i = 1: m - 1. The eigenvalues interlacing follows from Lemma 2.3.8.

(⇐): A direct extension of the proof of Theorem 2.3.5. To illustrate, consider the cubic case: Let $\widehat{P}(\zeta_1)$ ($\widehat{P}(\zeta_2)$) be negative definite (positive definite) for some $\zeta_1 \in (\lambda_{n+1}, \lambda_n)$ ($\zeta_2 \in (\lambda_{2n+1}, \lambda_{2n})$). Let the eigenvalues of P and \widehat{P} be distributed as in the statement of the theorem. The inclusion principle [42, Thm. 4.3.15] implies that the n-1 smallest (largest) eigenvalues of the constant matrix $P(\zeta_1)$ ($P(\zeta_2)$) are negative (positive). But det($P(\zeta_1)$) = $c \prod_{i=1}^{3n} (\zeta_1 - \lambda_i)$ (det($P(\zeta_2)$) = $c \prod_{i=1}^{3n} (\zeta_2 - \lambda_i)$), where c is a positive constant since the leading coefficient matrix of P is positive definite. As $\zeta_1 \in (\lambda_{n+1}, \lambda_n)$ ($\zeta_2 \in (\lambda_{2n+1}, \lambda_{2n})$), the sign of det($P(\zeta_1)$) (det($P(\zeta_2)$)) is $(-1)^n$ ($(-1)^{2n}$). Thus, the largest (smallest) eigenvalue of $P(\zeta_1)$ ($P(\zeta_2)$) is negative (positive) and $P(\zeta_1) < 0$ ($P(\zeta_1) > 0$) which is equivalent to saying that P is hyperbolic.

The following result helps in solving a particular quadratic Hyperbolic IPEP.

Theorem 2.3.10 [9, Thm. 5.1] Given two sets of real numbers $\{\lambda_j\}_{j=1}^{2n}$ and $\{\mu_j\}_{j=1}^{2n-2}$, $n \geq 2$, that satisfy (1) - (3) of Theorem 2.3.5, there exist $n \times n$ tridiagonal symmetric matrices A_1 and A_0 such that $\sigma(Q) = \{\lambda_j\}_{j=1}^{2n}$ where $Q(\lambda) = \lambda^2 I_n + \lambda A_1 + A_0$ and $\sigma(\widehat{Q}) = \{\mu_j\}_{j=1}^{2n-2}$. Moreover, Q is hyperbolic.

Unfortunately, the proof of the previous result does not extend easily to higher degrees.

2.3.3 Overdamped quadratic matrix polynomials

Overdamped quadratics form a subclass of hyperbolic quadratics. They arise in overdamped systems in structural mechanics [25], [50, Section 7.6] and are defined as follows.

Definition 2.3.11 (overdamped quadratic) A quadratic matrix polynomial is overdamped if it satisfies any one (and hence all) of the equivalent properties of Theorem 2.3.12. **Theorem 2.3.12** For a Hermitian quadratic matrix polynomial $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ the following are equivalent:

- (P1) All eigenvalues of Q are real, finite, nonpositive and distributed in two disjoint closed intervals, the left-most containing n eigenvalues of negative type and the right-most containing n eigenvalues of positive type.
- (P2) $A_2 > 0, A_1 > 0, A_0 \ge 0, and Q(\mu) < 0$ for some $\mu < 0$.
- (P3) $A_2 > 0$ and for every nonzero $x \in \mathbb{C}^n$, the scalar equation $x^*Q(\lambda)x = 0$ has 2 distinct real and finite nonpositive zeros.
- (O) $A_2 > 0, A_1 > 0, A_0 \ge 0$ and

$$(x^*A_1x)^2 > 4(x^*A_2x)(x^*A_0x) \quad for \ all \ nonzero \ x \in \mathbb{C}^n.$$

$$(2.20)$$

Proof. The equivalent characterizations of overdamping (P2), (P3) and (O) can be found in [33, Thm. 2.5]. Note that (P1) \Leftrightarrow (P3) follows from (P1) \Leftrightarrow (P3) in Theorem 2.3.4 to which is added the extra constraint that all the eigenvalues be nonpositive.

Note that property (O) is usually taken as the definition of overdamped quadratics. If equality is allowed in (2.20) for some nonzero x then the quadratic is said to be weakly overdamped. Its 2n eigenvalues are real and when ordered, $\lambda_n = \lambda_{n+1}$ with partial multiplicities² at most 2. Hence λ_n is either of mixed type or if it is not then the property that the eigenvalues are distributed in two disjoint intervals, each interval containing exactly n eigenvalues of one type, is lost.

Converting hyperbolic quadratics to overdamped ones

Hyperbolic quadratic matrix polynomials can be shifted to be overdamped. This shifting idea was discussed in [33], [40], [50]. Let $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ be Hermitian with $A_2 > 0$ and assume that Q is hyperbolic. Now with $\theta \ge \lambda_1 = \lambda_{\max}(Q)$, $\alpha = \lambda - \theta \le 0$ for all $\lambda \in \sigma(Q)$ and

²The partial multiplicities of an eigenvalue of Q are the sizes of the Jordan blocks in which it appears in a Jordan form for Q [31].

$$Q(\lambda) = Q(\alpha + \theta)$$

= $\alpha^2 A_2 + \alpha (A_1 + 2\theta A_2) + (A_0 + \theta A_1 + \theta^2 A_2)$
= $\alpha^2 A_{2\theta} + \alpha A_{1\theta} + A_{0\theta}$
= $\alpha^2 A_2 + \alpha Q'(\theta) + Q(\theta)$
=: $Q_{\theta}(\alpha)$

is overdamped. Thus, quadratic overdamped matrix polynomials are shifted hyperbolic ones and vice versa. In their effective algorithm of detecting and solving hyperbolic QEPs, Guo, Higham and Tisseur [33] use this shifting idea to test hyperbolicity via testing overdamping and they emphasize how large shifts of the spectrum, which can be obtained whenever one uses an upper bound to the modulus of the largest eigenvalue, can be. This is easily observed because we may have a case like $|\lambda_n| \gg |\lambda_1|$. Such a large shift could lead to slow convergence of their iterations.

2.3.4 Definiteness of matrix polynomials

Hyperbolic pencils $L(\lambda) = \lambda A - B$ are definite since their coefficient matrices are Hermitian with A > 0. However definite pairs are not necessarily hyperbolic since A and B can both be indefinite. By relaxing the requirement of definiteness of the leading coefficient, Higham, Mackey, and Tisseur [38] introduced a new class of Hermitian matrix polynomials, the definite matrix polynomials, that extends the notion of hyperbolicity and is consistent with the definition of definite pencils. Definite matrix polynomials that are not hyperbolic arise in acoustic fluid-structure interaction problems [38].

Definition 2.3.13 (definite matrix polynomial) A Hermitian matrix polynomial is definite if it satisfies any one (and hence all) of the equivalent properties of Theorem 2.3.14.

Theorem 2.3.14 For an $n \times n$ Hermitian matrix polynomial $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j$ of degree *m* the following are equivalent:

(P1) All eigenvalues of P are real, of definite type and such that

$\underbrace{\lambda_{mn} \leq \cdots \leq \lambda_{(m-1)n+p+1}}_{\bullet} < \cdots$	$< \underbrace{\lambda_{jn+p} \leq \cdots \leq \lambda_{(j-1)n+p+1}}_{\checkmark} < \cdot$	$\cdots < \underbrace{\lambda_p \leq \cdots \leq \lambda_1}_{p \leq \cdots \leq n} \leq \infty$
n-p eigenvalues	$n \ eigenvalues \ of$	$p \ eigenvalues$
of $(-1)^{m-1}\varepsilon$ type	$(-1)^{j-1}\varepsilon$ type, $1 \le j \le m-1$	$of - \varepsilon \ type$

with $0 \le p < n$, where $\sigma(P) = \{\lambda_j\}_{j=1}^{mn}$, $\varepsilon = \pm 1$ and " $\alpha \varepsilon$ type" denotes positive type when $\alpha \varepsilon > 0$ and negative type otherwise.

- (P2) There exist $\mu_j \in \mathbb{R} \cup \{\infty\}$ with $\mu_0 > \mu_1 > \mu_2 > \cdots > \mu_{m-1}$ ($\mu_0 = \infty$ being possible) such that $P(\mu_0)$, $P(\mu_1)$, ..., $P(\mu_{m-1})$ are definite matrices with alternating parity.
- (P3) There exists μ ∈ ℝ ∪ {∞} such that the matrix P(μ) is definite and for every nonzero x ∈ ℂⁿ the scalar equation x*P(λ)x = 0 has m distinct zeros in ℝ ∪ {∞}.
 - (L) P has a definite linearization $L(\lambda) \in \mathbb{H}(P)$.

Note that when P is a definite matrix polynomial with A_m definite one of P or -P is hyperbolic and by Theorem 2.3.4, p = 0 in property (P1) of Theorem 2.3.14.

Proof. The characterizations (P2), (P3) and (L) and their equivalence can be found in [38, Thms. 2.6 and 4.1].

(P1) \Rightarrow (P3): Suppose neither P nor -P is hyperbolic, i.e. $p \neq 0$. Let μ be such that $\lambda_{p+1} < \mu < \lambda_p$. Then homogeneously rotate P into \tilde{P} so that μ corresponds to $\tilde{\mu} = \infty$. The rotation moves the p largest eigenvalues of P to the n-p smallest ones to form a single group of n eigenvalues (see Figures 2.2 and 2.4) which, by Lemma 2.2.6, are all of $(-1)^{m-1}\varepsilon$ type. The types of the remaining m-1 groups of n eigenvalues remain unchanged. Hence by property (P1) of Theorem 2.3.4, \tilde{P} or $-\tilde{P}$ is hyperbolic. By property (P3) of Theorem 2.3.4, $x^*\tilde{P}(\tilde{\lambda})x = 0$ has real distinct roots for all nonzero $x \in \mathbb{C}^n$ and therefore $x^*P(\lambda)x = 0$ has distinct roots in $\mathbb{R} \cup \{\infty\}$ $(x^*A_mx = 0 \text{ is possible})$. Also by [38, Lem. 2.1], $P(\mu)$ is definite.

(P3) \Rightarrow (P1): Homogeneously rotate P into \tilde{P} so that μ corresponds to $\tilde{\mu} = \infty$. Then $\tilde{P}(\infty) = \tilde{A}_m$ is definite, say $\tilde{A}_m > 0$. Now if $x^*P(\lambda)x = 0$ has distinct roots in $\mathbb{R} \cup \{\infty\}$ then $x^*\tilde{P}(\tilde{\lambda})x = 0$ has real distinct roots (and no infinite root since $\tilde{A}_m > 0$). Thus $\widetilde{P}(\widetilde{\lambda})$ is hyperbolic. Then property (P1) for *P* follows from property (P1) of Theorem 2.3.4, Lemma 2.2.6 and Figure 2.4.

The following result follows from (P2) in Theorem 2.3.14 and from counting sign changes in eigenvalues of the matrix $P(\mu)$ (see [38, Thm. 2.4 and its proof]).

Theorem 2.3.15 For a definite matrix polynomial $P(\lambda)$ of degree m with eigenvalues as in property (P1) of Theorem 2.3.14, let

$$\mathcal{I}_j = (\lambda_{jn+p+1}, \lambda_{jn+p}), \quad j = 1: m-1$$

and

$$\mathcal{I}_0 = \begin{cases} (\lambda_{p+1}, \lambda_p) & \text{if } p \neq 0, \\ (\lambda_1, +\infty) & \text{if } p = 0, \end{cases} \qquad \mathcal{I}_m = \begin{cases} \emptyset & \text{if } p \neq 0, \\ (-\infty, \lambda_{mn}) & \text{if } p = 0. \end{cases}$$

Then $P(\mu)$ is definite for any $\mu \in \mathcal{I}_j$, j = 0: m and if $\mu_j \in \mathcal{I}_j$, $\mu_{j+1} \in \mathcal{I}_{j+1}$ then $P(\mu_j)$ and $P(\mu_{j+1})$ have opposite parity.

Niendorf and Voss's algorithm [78] can be used to detect whether a Hermitian matrix polynomial is definite or not. For definite polynomials it also returns the μ_j of property (P2) so that a definite linearization can be built as shown in [38, Thm. 4.2].

2.4 Quasidefinite matrix polynomials

We have just seen that definite matrix polynomials are characterized by the fact that all their eigenvalues are real and of definite type and with a particular distribution of the eigenvalue types. We now consider a wider class of Hermitian matrix polynomials with real eigenvalues of definite type for which no assumption is made on the distribution of the eigenvalues types.

2.4.1 Definitizable pencils

Definite pencils form only a small subclass of Hermitian pencils with real and semisimple eigenvalues. We now consider a larger subclass of such pencils. **Definition 2.4.1 (definitizable pencils)** A Hermitian pencil $\lambda A - B$ is definitizable if it satisfies any one (and hence all) of the equivalent properties of Theorem 2.4.2.

Theorem 2.4.2 For an $n \times n$ Hermitian pencil $L(\lambda) = \lambda A - B$ the following are equivalent:

- (P1) All the eigenvalues of L are real, finite, and of definite type.
- (P2) A is nonsingular and there exists a real polynomial q such that $Aq(A^{-1}B) > 0$.
- (P3) A is nonsingular and the scalar equation $x^*L(\lambda)x = 0$ has one zero in \mathbb{R} for all eigenvectors $x \in \mathbb{C}^n$ of L.
- (P4) A is nonsingular and $(x^*Ax, x^*Bx, \dots, x^*A(A^{-1}B)^{n-1}x) \neq 0$ for all nonzero $x \in \mathbb{C}^n$.
- (D) There exists a nonsingular $X \in \mathbb{C}^{n \times n}$ such that

$$X^*AX = \begin{bmatrix} I_k & 0\\ 0 & -I_{n-k} \end{bmatrix}, \quad X^*BX = \begin{bmatrix} J_+ & 0\\ 0 & -J_- \end{bmatrix}, \quad (2.21)$$

where $J_+ \in \mathbb{R}^{k \times k}$ and $J_- \in \mathbb{R}^{(n-k) \times (n-k)}$ are diagonal and $\sigma(J_+) \cap \sigma(J_-) = \emptyset$.

Proof. The equivalence of the characterizations (P1), (P2) and (P4) can be found in [59, Thm. 1.3]. (D) \Rightarrow (P3) is immediate. We show that (P3) \Rightarrow (P1) and (P1) \Rightarrow (D).

 $(P3) \Rightarrow (P1)$ Suppose one eigenvalue is not real or is of mixed type then by [59, Lem. 2.2] there exists a corresponding eigenvector x such that $x^*Ax = 0$ and hence (P3) does not hold.

 $(P1) \Rightarrow (D) A$ is nonsingular since all eigenvalues are finite and $\lambda A - B$ is simultaneously diagonalizable by congruence since all the eigenvalues are real and semisimple. Hence there exists X nonsingular such that $X^*(\lambda A - B)X = \lambda D - J$ is real diagonal. Since D is nonsingular, we can choose X such that $D = \begin{bmatrix} I_k & 0\\ 0 & -I_{n-k} \end{bmatrix}$ and partition J as $\begin{bmatrix} J_+ & 0\\ 0 & -J_- \end{bmatrix}$ conformably with D. Then the property that all the eigenvalues are of definite type implies that $\sigma(J_+) \cap \sigma(J_-) = \emptyset$.

Lancaster and Ye [59] define definitizable pencils by property (P2) and add the adverb "strongly" to definitizable to emphasize the strict inequality in (P2). Note

that the real polynomial q in (P2) is not unique and that J_+ in (2.21) contains the eigenvalues of positive type and J_- those of negative type. Now if $L(\lambda) = \lambda A - B$ with nonsingular A is definite then by property (P2) of Theorem 2.3.2, there exists $\mu \in \mathbb{R} \cup \{\infty\}$ such that the matrix $L(\mu)$ is definite. Then $Aq(A^{-1}B) > 0$ for $q(x) = -\varepsilon$ if $\mu = \infty$ and $q(x) = \varepsilon(x - \mu)$ otherwise, where $\varepsilon = 1$ if $L(\mu) < 0$ and $\varepsilon = -1$ if $L(\mu) > 0$. Hence definite pencils with nonsingular leading coefficient matrix are definitizable.

Though not necessarily computationally efficient, property (P1) provides an easy way to check whether a Hermitian pencil $\lambda A - B$ is definite or definitizable or none of these.

As a by-product of the proof of [59, Thm. 1.3], a real polynomial q of minimal degree such that $Aq(A^{-1}B) > 0$ can easily be constructed once all the eigenvalues of a definitizable pencil $\lambda A - B$ are known together with their types, as shown in the next theorem. The knowledge of q can be useful when constructing conjugate gradient iterations for solving saddle point problems [63].

Theorem 2.4.3 For an $n \times n$ definitizable pencil $\lambda A - B$ with eigenvalues $\lambda_n \leq \cdots \leq \lambda_1$, let k_j , $j = 1: \ell - 1$ be the set of increasing integers such that

$$\begin{array}{ll} \underbrace{\lambda_n \leq \cdots \leq \lambda_{k_{\ell-1}+1}}_{n-k_{\ell-1} \ eigenvalues} & < \cdots < \underbrace{\lambda_{k_{j+1}} \leq \cdots \leq \lambda_{k_j+1}}_{k_{j+1}-k_j \ eigenvalues} & < \cdots < \underbrace{\lambda_{k_1} \leq \cdots \leq \lambda_1}_{k_1 \ eigenvalues} \\ of \ (-1)^{\ell-1} \varepsilon \ type & of \ (-1)^j \varepsilon \ type & of \ \varepsilon \ type \end{array}$$

Then $p(x) = \varepsilon \prod_{j=1}^{\ell-1} (x - \mu_j)$ with $\lambda_{k_j+1} < \mu_j < \lambda_{k_j}$ is a real polynomial of minimal degree $\ell - 1$ such that $Ap(A^{-1}B) > 0$.

Example 2.4.4 The pencils

$$L_1(\lambda) = \lambda \begin{bmatrix} 1 & 0 & 0 \\ 0 & -1 & 0 \\ 0 & 0 & 1 \end{bmatrix} - \begin{bmatrix} 1 & 0 & 0 \\ 0 & -2 & 0 \\ 0 & 0 & 3 \end{bmatrix}, \quad L_2(\lambda) = \lambda \begin{bmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & -1 \end{bmatrix} - \begin{bmatrix} 1 & 0 & 0 \\ 0 & 2 & 0 \\ 0 & 0 & -3 \end{bmatrix}$$

have the same eigenvalues, i.e., $\sigma(L_1) = \sigma(L_2) = \{1, 2, 3\}$. Both pencils are definitizable because the eigenvalues are real, distinct and hence of definite type but only L_2 is definite since the eigenvalue 3, of negative type, is separated from the eigenvalues 1 and 2 of positive type as in the right-most depiction of Figure 2.3. Furthermore, by Theorem 2.3.15, $L_2(\mu) > 0$ for all $\mu \in (2,3)$. Also, by Theorem 2.4.3 any polynomial of the form $p(x) = (x - \mu_1)(x - \mu_2)$ with $\mu_1 \in (2,3)$ and $\mu_2 \in (1,2)$ satisfies $A_1p(A_1^{-1}B_1) > 0$, where $L_1(\lambda) =: \lambda A_1 - B_1$.

2.4.2 Quasihyperbolic matrix polynomials

The notion of definitizable pencils extends to matrix polynomials of degree higher than one.

Definition 2.4.5 (quasihyperbolic matrix polynomial) A Hermitian matrix polynomial is quasihyperbolic if it satisfies either (and hence both) of the equivalent properties of Theorem 2.4.6.

Theorem 2.4.6 For an $n \times n$ Hermitian matrix polynomial $P(\lambda)$ the following are equivalent:

- (P1) All the eigenvalues of P are real, finite and of definite type.
- (L) Any linearization $L(\lambda) \in \mathbb{H}(P)$ is definitizable.

Proof. We note that this result was proved in [53, Thm. 7.1] for a particular linearization in $\mathbb{H}(P)$.

A matrix polynomial has only finite eigenvalues if and only if its leading coefficient matrix is nonsingular. Let $L \in \mathbb{H}(P)$ be a linearization of P. Then P has only finite eigenvalues if and only if L has only finite eigenvalues or equivalently L has nonsingular leading matrix coefficient. Moreover $\sigma(P) \subseteq \mathbb{R}$ if and only if $\sigma(L) \subseteq \mathbb{R}$. By Lemma 2.2.7, the eigenvalues of P are of definite type if and only if those of Lare of definite type. Hence by (P1) of Theorem 2.4.2, (P1) is equivalent to (L).

There is no obvious extension of properties (P2) and (P4) of Theorem 2.4.2 to quasihyperbolic matrix polynomials at the nonlinear level but by property (L) of Theorem 2.4.6 and property (P2) of Theorem 2.4.2, we have that P is quasihyperbolic if and only if there exists a real polynomial q such that $B_m q(B_m^{-1}B_{m-1}) > 0$, where $\lambda B_m - B_{m-1}$ is the $mn \times mn$ pencil (2.11). Property (P3) of Theorem 2.4.2 extends to quadratic matrix polynomials but not to higher degrees as shown by the next theorem and the following example. **Theorem 2.4.7** Let P be a Hermitian matrix polynomial of degree m with nonsingular leading matrix coefficient. If the scalar equation $x^*P(\lambda)x = 0$ has m real distinct zeros for every eigenvector x of P then P is quasihyperbolic. The converse is also true when $m \leq 2$.

Proof. Distinct real roots of $x^*P(\lambda)x = 0$ for all eigenvectors x of P implies that $\sigma(P) \subset \mathbb{R}$ and $x^*P'(\lambda_0)x \neq 0$ for each eigenvalue $\lambda_0 \in \sigma(P)$. Hence all eigenvalues are real, finite and of definite type, so P is quasihyperbolic by Theorem 2.4.6.

The converse is clearly true for linear P (see Theorem 2.4.2). Now for quadratic P, suppose the scalar quadratic $x^*P(\lambda)x = 0$, where x is an eigenvector, has a real double root. Then this double root is necessarily an eigenvalue of P, say λ_0 associated with x, and since it is a double root, $x^*P'(\lambda_0)x = 0$, so that λ_0 is of mixed type. Hence P is not quasihyperbolic.

Here is an example to show that the converse of Theorem 2.4.7 does not hold for polynomials of degree 3. The cubic polynomial

$$P(\lambda) = \lambda^3 \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix} + \lambda^2 \begin{bmatrix} 9 & 0 \\ 0 & -6 \end{bmatrix} + \lambda \begin{bmatrix} -10 & 0 \\ 0 & 11 \end{bmatrix} + \begin{bmatrix} 0 & 0 \\ 0 & -6 \end{bmatrix}$$

is quasihyperbolic. Any nonzero $x \in \mathbb{C}^2$ is an eigenvector associated with the eigenvalue $\lambda = 1$. It is easily checked that with $x = \begin{bmatrix} 1 \\ 1 \end{bmatrix}$, the scalar polynomial $x^*P(\lambda)x$ has one real root and two complex conjugate roots.

We remark that a definite pencil $\lambda A - B$ is not necessarily definitizable because A can be singular, and a definite matrix polynomial is not necessarily quasihyperbolic because its leading term can be singular. We therefore extend the definition of quasihyperbolic matrix polynomials to allow singular leading term and call such polynomials quasidefinite.

Definition 2.4.8 A Hermitian matrix polynomial $P(\lambda)$ is quasidefinite if $\sigma(P) \subset \mathbb{R} \cup \{\infty\}$ and each eigenvalue is of definite type.

Take a quasidefinite matrix polynomial P. Since P is regular, there exists $\mu \in \mathbb{R} \cup \{\infty\}$ such that $P(\mu)$ is nonsingular. Homogeneously rotate P into \tilde{P} so that μ corresponds to ∞ and $\tilde{A}_m = \tilde{P}(\mu)$ is nonsingular. Then by Lemma 2.2.6 the eigenvalues of \tilde{P} are all of definite type and \tilde{P} is quasihyperbolic. Hence any quasidefinite matrix polynomial is a "homogeneously rotated" quasihyperbolic one. Note that amongst the properties (P1), (P2) and (P3) we started with in Section 2.3, only a property of type (P1) remains for quasihyperbolic and quasidefinite matrix polynomials.

2.4.3 Gyroscopically stabilized systems

Quadratic matrix polynomials associated with gyroscopic systems have the form

$$G(\lambda) = \lambda^2 M + \lambda C + K,$$

where M, K are Hermitian and C is skew-Hermitian [90]. As $G(\lambda)^* = G(-\overline{\lambda})$, the spectrum of $G(\lambda)$ is symmetric with respect to the imaginary axis. The quadratic $G(\lambda)$ is not Hermitian but

$$Q(\lambda) = -G(-i\lambda) = \lambda^2 M + \lambda(iC) - K =: \lambda^2 A_2 + \lambda A_1 + A_0$$

is. The gyroscopic system is said to be weakly stable if all the eigenvalues of G lie on the imaginary axis or equivalently, if the eigenvalues of Q are all real. The following definition appears in [12]. For a Hermitian B we write $|B| = (B^2)^{1/2}$, where the square root is the principal square root [36, Prob. 1.27].

Definition 2.4.9 A Hermitian $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ is gyroscopically stabilized if $A_2 > 0$, $A_0 > 0$ and A_1 is indefinite and nonsingular with $|A_1| > \mu A_2 + \mu^{-1} A_0$ for some positive μ .

Barkwell, Lancaster, and Markus [12] prove that gyroscopically stabilized quadratics have real eigenvalues of definite type that belong to at most four distinct intervals of alternating types, with the number of eigenvalues in each interval depending on the number of positive eigenvalues p of A_1 (see Figure 2.8). Hence gyroscopically stabilized quadratics are quasihyperbolic. They are overdamped when $A_1 > 0$.

Note that the eigenvalue type distribution in Figure 2.8 is just a property of gyroscopically stabilized quadratics.



Figure 2.8: Eigenvalue distribution of gyroscopically stabilized quadratics $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$, where p is the number of positive eigenvalues of A_1 .

Example 2.4.10 Let $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ with

$$A_2 = I_3, \quad A_1 = \begin{bmatrix} 1.7122 & -0.0865 & 0 \\ -0.0865 & 1.3770 & 0 \\ 0 & 0 & -5 \end{bmatrix}, \quad A_0 = \begin{bmatrix} 0.7227 & -0.0865 & 0 \\ -0.0865 & 0.4164 & 0 \\ 0 & 0 & 6 \end{bmatrix}.$$

Note that $A_2, A_0 > 0$ and A_1 is indefinite and nonsingular. Since $\lambda^2 A_2 + \lambda |A_1| + A_0$ is not overdamped, $Q(\lambda)$ is not gyroscopically stabilized while its eigenvalues are distributed as in Figure 2.8 with p = 2.

2.5 Diagonalizable quasidefinite matrix polynomials

Recall that a Hermitian pencil is diagonalizable by congruence if and only if its eigenvalues belong to $\mathbb{R} \cup \{\infty\}$ and are semisimple [56], a property shared by both definite and definitizable pencils. We now investigate how this property extends to (quasi)hyperbolic and definite matrix polynomials, thereby extending the simultaneous diagonalization property (D) in Theorems 2.3.2 and 2.4.2.

Matrix polynomials cannot in general be simultaneously diagonalized by a strict equivalence transformation. However, Lancaster and Zaballa [61] have recently characterized a class of quadratic matrix polynomials that can be diagonalized by applying strict equivalence transformations or congruences to a linearization of the quadratic while preserving the structure of the linearization. Along the same line, we identify amongst all quasidefinite matrix polynomials of arbitrary degree those that can be diagonalized by a congruence transformation applied to a Hermitian linearization Lof the matrix polynomial P while maintaining the block structure of L. In particular, we show that all hyperbolic matrix polynomials can be transformed to a diagonal form in this way.

Two matrix polynomials are *isospectral* if they have the same eigenvalues with the same partial multiplicities. If furthermore they share the same sign characteristic then these two matrix polynomials are *strictly isospectral* [54]. For example any linearization $L(\lambda) \in \mathbb{H}(P)$ is isospectral to P but not necessarily strictly isospectral as shown by Lemma 2.2.7.

In [28] the authors focus on the theory and practice of diagonalizing $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ by the application of simple λ -dependent transformations which they call *filters*. This can be done without using linearizations. They show that, in general, if systems

$$Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0, \quad \widetilde{Q}(\lambda) = \lambda^2 \widetilde{A}_2 + \lambda \widetilde{A}_1 + \widetilde{A}_0$$

are isospectral, then there exist pencils $F(\lambda)$ and $\widetilde{F}(\lambda)$ such that

$$\widetilde{F}(\lambda)Q(\lambda) = \widetilde{Q}(\lambda)F(\lambda).$$

The interest, then, is in finding filters for which $\tilde{Q}(\lambda)$ is diagonal, which is useful in engineering and control theory as there are mechanisms admitting physical implementation of such filters.

Now suppose that two $n \times n$ quasihyperbolic matrix polynomials P_1 and P_2 of degree m > 1 are strictly isospectral. Let L_{1m} and L_{2m} be the *m*th basis pencils of $\mathbb{H}(P_1)$ and $\mathbb{H}(P_2)$, respectively (see (2.11)). By Lemma 2.2.7 L_{1m} and L_{2m} are strictly isospectral and by Theorem 2.4.6 they are also definitizable. It follows from property (D) of Theorem 2.4.2, that there exist nonsingular matrices $X_1, X_2 \in \mathbb{C}^{nm \times nm}$ such that

$$X_1 L_{1_m}(\lambda) X_1^* = \lambda \begin{bmatrix} I_k & 0\\ 0 & -I_{n-k} \end{bmatrix} - \begin{bmatrix} J_+ & 0\\ 0 & -J_- \end{bmatrix} = X_2 L_{2_m}(\lambda) X_2^*.$$

The matrix $X_2^{-1}X_1$ defines a structure preserving congruence [54], [61], [89] since it preserves the block structure of $L_{1_m}(\lambda)$: $(X_2^{-1}X_1)L_{1_m}(\lambda)(X_2^{-1}X_1)^* = L_{2_m}(\lambda)$. Thus if there exists a diagonal quasihyperbolic matrix polynomial $D(\lambda)$ of degree m strictly isospectral to $P(\lambda)$ then there exits a congruence transformation that preserves the block structure of the mth basis pencils of $\mathbb{H}(P)$ but also diagonalizes each block. If such structure preserving congruence exists then P is said to be diagonalizable. In what follows $D(\lambda)$ has the form

$$D(\lambda) = \operatorname{diag}(d_1(\lambda), \dots, d_n(\lambda)), \qquad (2.22a)$$

$$d_i(\lambda) = \delta_i(\lambda - \lambda_{i_m}) \cdots (\lambda - \lambda_{i_2})(\lambda - \lambda_{i_1}), \quad \delta_i \neq 0, \quad i = 1:n \quad (2.22b)$$

with $\bigcup_{i=1}^{n} \{i_1, \ldots, i_m\} = \{1, \ldots, mn\}$ and $\delta_i = \pm 1$. The scalars $\lambda_{i_j}, j = 1; m, i = 1; n$ are the eigenvalues of $D(\lambda)$ and $P(\lambda)$ and are therefore real. We assume that they are ordered as

$$\lambda_{mn} \leq \cdots \leq \lambda_{j+1} \leq \lambda_j \leq \cdots \leq \lambda_1.$$

Theorem 2.5.1 An $n \times n$ quasihyperbolic matrix polynomial of degree m with eigenvalues $\lambda_{mn} \leq \cdots \leq \lambda_1$ is strictly isospectral to an $n \times n$ diagonal matrix polynomial of degree m if and only if there is a grouping of its eigenvalues into n subsets of m distinct eigenvalues $\{\lambda_{i_j} : j = 1: m\}_{i=1}^n$ such that with the ordering $\lambda_{i_m} < \cdots < \lambda_{i_2} < \lambda_{i_1}, i = 1:n$, the eigenvalue λ_{i_j} is of $\delta_i(-1)^{j-1}$ type, where λ_{i_1} is of δ_i type.

Proof. Let $P(\lambda)$ denote the $n \times n$ quasihyperbolic matrix polynomial of degree m.

 (\Rightarrow) Suppose $P(\lambda)$ is strictly isospectral to an $n \times n$ diagonal matrix $D(\lambda)$ of degree m as in (2.22). The scalar polynomials $d_i(\lambda)$ must have distinct roots since otherwise $0 = d'_i(\lambda_{i_j}) = e_i^* D'(\lambda_{i_j}) e_i$ for some eigenvalue λ_{i_j} , which implies that λ_{i_j} is not of definite type, a contradiction. Here e_i , the *i*th column of I_n , is a corresponding eigenvector. Consider the grouping $\{\lambda_{i_j}, j = 1: m\}_{i=1}^n$ of the eigenvalues. With the ordering $\lambda_{i_m} < \cdots < \lambda_{i_2} < \lambda_{i_1}$, i = 1: n, it is easily seen that this grouping must be such that, in each group, the eigenvalue λ_{i_j} is of $\delta_i(-1)^{j-1}$ type and the sign of δ_i is determined by the type of λ_{i_1} .

 (\Leftarrow) Let $\{\lambda_{i_j}, j = 1: m\}_{i=1}^n$ be a grouping of the eigenvalues of P into n subsets of m distinct eigenvalues, such that with the ordering $\lambda_{i_m} < \cdots < \lambda_{i_2} < \lambda_{i_1}$, i = 1: n, the eigenvalue λ_{i_j} is of $\delta_i(-1)^{j-1}$ type, where λ_{i_1} is of δ_i type. Let $D(\lambda)$ and $d_i(\lambda)$ be as in (2.22). Then by construction $D(\lambda)$ is quasihyperbolic and its eigenvalues and their types are the same as the eigenvalues of P and their types. Hence D is strictly isospectral to P.

Example 2.5.2 If there is a 2×2 cubic quasihyperbolic matrix polynomial $P(\lambda)$ with real eigenvalues $\lambda_1 > \lambda_2 > \lambda_3 = \lambda_4 > \lambda_5 > \lambda_6$ and associated types {+, -, +, +, -},

where + means positive type and - denotes negative type, then this polynomial is not diagonalizable by structure preserving congruence because there is no sorting of the eigenvalues into two groups of three distinct eigenvalues, which when ordered have alternating types. Note that if the sign characteristic had been $\{+, -, +, +, -, +\}$ then P would have been diagonalizable by structure preserving congruence.

Quasihyperbolic matrix polynomials of degree m strictly isospectral to diagonal matrix polynomials of degree m form a new subclass of Hermitian polynomials with eigenvalues all real and of definite type. Note that $n \times n$ Hermitian quasihyperbolic quadratics have n eigenvalues of positive type and n eigenvalues of negative type [29, Thm. 1.3]. So there is always a sorting of the eigenvalues into n groups of two distinct eigenvalues with opposite types. By Theorem 2.3.4, the eigenvalues of an $n \times n$ hyperbolic matrix polynomial of degree m are distributed in m disjoint intervals each of which contains n eigenvalues and, the types of the intervals alternate. So we can always sort the eigenvalues in n subsets of m distinct eigenvalues, which when ordered have alternating types. Hence by Theorem 2.5.1, quasihyperbolic quadratics and hyperbolic matrix polynomials of arbitrary degree, say m, are strictly isospectral to diagonal matrix polynomials of degree m. This result also applies to quasidefinite quadratic matrix polynomials and definite matrix polynomials.

- **Corollary 2.5.3** (a) A quasidefinite quadratic matrix polynomial is always strictly isospectral to a quasidefinite diagonal quadratic matrix polynomial.
 - (b) A definite matrix polynomial is always strictly isospectral to a definite diagonal matrix polynomial.

Proof. The proofs of (a) and (b) are similar so we just provide that for (b). A definite matrix polynomial P of degree m is a homogeneously rotated hyperbolic matrix polynomial \tilde{P} of degree m. From the comments preceding Corollary 2.5.3, \tilde{P} is strictly isospectral to a diagonal matrix polynomial \tilde{D} of degree m. Applying back the homogeneous rotation to \tilde{D} produces a diagonal matrix polynomial of degree m, which is strictly isospectral to P.

The following result is a direct consequence of [9, Thm. 5.1].

Corollary 2.5.4 Every $n \times n$ hyperbolic quadratic matrix polynomial is strictly isospectral to a real $n \times n$ tridiagonal hyperbolic matrix polynomial.

Applying [1, Thm. 6.1] for 2n, if $Q(\lambda)$ is a given $n \times n$ quasihyperbolic quadratic matrix polynomial, we can construct an $n \times n$ tridiagonal symmetric quadratic matrix polynomial, say Q_{tri} , for which $\sigma(Q_{tri}) = \sigma(Q)$ by assigning 2n - 2 distinct real numbers to be the eigenvalues of $\widehat{Q}_{tri}(\lambda)$, which is the leading $(n - 1) \times (n - 1)$ submatrix of $Q_{tri}(\lambda)$, such that $\sigma(Q_{tri}) \cap \sigma(\widehat{Q}_{tri})$ is empty.

2.6 Summary and further discussion

In this chapter we have described the diverse subclasses of Hermitian matrix polynomials with real eigenvalues of definite type scattered in the literature in a consistent way. We have pointed out their differences and similarities in particular with respect to the distribution of their eigenvalue types on the extended real line. This latter property allows to produce the diagram in Figure 2.1. Among these subclasses we have identified a new class of diagonalizable matrix polynomials. One of the main tools used in this study is the Hermitian *m*th basis pencil $\lambda B_m - B_{m-1}$ in (2.11) which preserves the eigenvalue type. For this particular linearization, as shown in Lemma 2.2.7, the v-polynomial equals 1. This may explain why this linearization is the most frequent one appearing in the literature concerning Hermitian matrix polynomials. Though, other Hermitian linearizations of quasidefinite matrix polynomials possess the property of preserving the eigenvalue type. For example, Fiedler-like Hermitian linearizations defined in [7] have this property. We realized this non-obvious fact during Fernando De Terán's visit to Manchester in February–May 2010. He generously shared a lemma showing that if y and x are, respectively, left and right eigenvectors of a given Fiedler-like linearization F of P associated with an eigenvalue λ , then

$$y^*F'(\lambda)x = w^*P'(\lambda)v,$$

where w and v are, respectively, left and right eigenvectors of P associated with λ . Thus F preserves the eigenvalue type.

Chapter 3

Standard Triples of Structured Matrix Polynomials

3.1 Introduction

Standard and Jordan triples for matrix polynomials were introduced and developed by Gohberg, Lancaster and Rodman (see for example [29], [30], [31]). Jordan triples extend to matrix polynomials of degree m

$$P(\lambda) = \sum_{j=0}^{m} \lambda^{j} A_{j}, \quad A_{j} \in \mathbb{F}^{n \times n}, \quad \det(A_{m}) \neq 0,$$
(3.1)

the notion of Jordan pair (X, J) for a single matrix $A \in \mathbb{C}^{n \times n}$, where $X \in \mathbb{C}^{n \times n}$ is nonsingular, J is a Jordan canonical form for A, and $A = XJX^{-1}$. The matrix Xin a Jordan triple (X, J, Y) for $P(\lambda)$ is $n \times mn$, and as for the single matrix case, it contains the right eigenvectors and generalized eigenvectors of $P(\lambda)$. The matrix $J \in \mathbb{C}^{mn \times mn}$ is in Jordan canonical form, displaying the elementary divisors of $P(\lambda)$, and the matrix $Y \in \mathbb{C}^{mn \times n}$ plays the role of X^{-1} for a single matrix, i.e., the columns of Y^* determine left eigenvectors and generalized eigenvectors of $P(\lambda)$. A Jordan triple is a particular standard triple (U, T, V) in which the matrix T is in canonical form. Precise definitions of standard and Jordan triples are given in Section 3.2.2.

Our objective in this chapter is to study the standard and Jordan triples of structured matrix polynomials $P(\lambda)$ of the types listed in Table 1.1. Indeed the eigentriples' pairing of such matrix polynomials given in Table 3.1 indicates that when all

Table 3.1: Eigentriple pairings (if any) for $P(\lambda) \in \mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$ with structure $\mathcal{S} \in \mathbb{S}$. When $\mathbb{F} = \mathbb{R}$, the eigenvalues of *T*-even and *T*-odd $P(\lambda)$ occur in quadruples $(\lambda, \overline{\lambda}, -\lambda, -\overline{\lambda})$ and those of *T*-palindromic and *T*-antipalindromic $P(\lambda)$ occur in quadruples $(\lambda, \overline{\lambda}, 1/\lambda, 1/\overline{\lambda})$.

Herm	Symm	*-even, *-odd	T-even, T -odd	*-(anti)palindromic	T-(anti)palindromic
(λ, x, y)	$(\lambda, x, \overline{x})$	(λ, x, y)	(λ, x, y)	(λ, x, y)	(λ, x, y)
$(\overline{\lambda},y,x)$		$(-\overline{\lambda},y,x)$	$(-\lambda, \overline{y}, \overline{x})$	$(rac{1}{\overline{\lambda}},y,x)$	$(rac{1}{\lambda}, \overline{y}, \overline{x})$

eigenvalues are semisimple, there exists a nonsingular matrix S that depends on the structure and connects Y with X^* . For example, consider a 2×2 Hermitian quadratic with Jordan form $J = \text{diag}(\lambda_1, \lambda_2, \lambda, \bar{\lambda})$ and corresponding right eigenvector matrix $X = \begin{bmatrix} x_1 & x_2 & x & y \end{bmatrix}$, where λ_1, λ_2 are real and λ is nonreal. The first column of Table 3.1 implies that $Y^* = \begin{bmatrix} \bar{s}_1 x_1 & \bar{s}_2 x_2 & \bar{s}_4 y & \bar{s}_3 x \end{bmatrix} = XS^*$ for some nonsingular matrix S of the form

$$S = \begin{bmatrix} s_1 & 0 & 0 & 0 \\ 0 & s_2 & 0 & 0 \\ 0 & 0 & 0 & s_4 \\ 0 & 0 & s_3 & 0 \end{bmatrix}$$

The structure of standard and Jordan triples are well understood for Hermitian matrix polynomials [29], [30] and more recently real symmetric matrix polynomials [18], [60]. With no assumption on the sizes of the Jordan blocks, GLR [29] show that if (X, J, Y) is a Jordan triple for a Hermitian matrix polynomial then $Y = SX^*$ for some nonsingular $mn \times mn$ matrix S such that $S = S^*$ and $JS = (JS)^*$. We show in Section 3.3 that results of this type also hold for the structures in \mathbb{S} , where

 $S = \{\text{Hermitian, symmetric, *-even, *-odd, } T\text{-even, } T\text{-odd,}$ (3.2) *-palindromic, *-antipalindromic, T-palindromic, T-antipalindromic}.

For $S \in S$, we introduce the notion of S-structured standard triples. With the exception of T-(anti)palindromic matrix polynomials of even degree with both -1 and 1 as eigenvalues, we show that $P(\lambda)$ has structure S if and only if $P(\lambda)$ admits an S-structured standard triple, and that for any $P(\lambda)$ with structure S, all standard triples for $P(\lambda)$ are S-structured. Finally, we study in Section 3.4 the special case of S-structured Jordan triples.

Two features of this work are (a) a distinction, when necessary, between triples and matrix polynomials defined over the complex (\mathbb{C}) or real (\mathbb{R}) fields, and (b) a unified presentation of the results, except in Section 3.5, where we provide explicit expressions for the *S*-matrix of *S*-structured Jordan triples, which are structuredependent.

3.2 Preliminaries

The set of all matrix polynomials with coefficient matrices in $\mathbb{F}^{n \times n}$ is denoted by $\mathcal{P}(\mathbb{F}^n)$. When the polynomials are structured with structure \mathcal{S} , the corresponding space is denoted by $\mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$ (see Table 1.1). Throughout this chapter we assume that $P(\lambda)$ has a nonsingular leading coefficient matrix as in (3.1).

Recall that linearizations play a major role in the development of the theory of matrix polynomials and note that both $P(\lambda)$ and its linearization $L(\lambda)$ share the same Jordan form. The construction of linearizations that respect the structure of a given matrix polynomial has been an important area of research in the last decade (see for example [8], [24], [37], [69]). We give in Section 3.2.1 explicit expressions for the structured linearizations, which are important for the derivation of some results in Section 3.3 and all the results in Section 3.4

3.2.1 Structured linearizations

Let $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j \in \mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$ with $\det(A_m) \neq 0$ and $\mathcal{S} \in \mathbb{S}$. Recall that the vector space of pencils $\mathbb{L}_1(P)$ in (1.9) provides a rich source for structured linearizations. It is shown in [37], [65], [69] that for some $v \in \mathbb{F}^m$ satisfying the additional constraint

- (i) $v \in \mathbb{R}^m$ if $\mathcal{S} =$ Hermitian,
- (ii) $v = \Sigma_m v$ if $S \in \{T \text{-even}, T \text{-odd}\}$ or $v = \Sigma_m \overline{v}$ if $S \in \{* \text{-even}, * \text{-odd}\}$,
- (iii) $v = F_m v$ if $S \in \{T\text{-palindromic}, T\text{-antipalindromic}\}$ or $v = F_m \bar{v}$ if $S \in \{*\text{-palindromic}, *\text{-antipalindromic}\},$

where

$$\Sigma_m = \operatorname{diag}((-1)^{m-1}, \dots, (-1)^0), \qquad F_m = \begin{bmatrix} & 1 \\ & \ddots & \\ 1 & & \end{bmatrix}_{m \times m},$$

there exists a unique pencil $\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}} \in \mathbb{L}_1(P)$ with structure $\mathcal{S} \in \mathbb{S}$. This pencil is a linearization of $P(\lambda)$ if the roots of the v-polynomial are not eigenvalues of P, see Theorem 1.2.2. The vector $v = e_m$ is an admissible vector for $\mathcal{S} \in \{\text{Hermitian}, \text{symmetric}, \star\text{-even}, \star\text{-odd}\}$ since $e_m \in \mathbb{R}^m$ and $\Sigma_m e_m = e_m$. Also, the roots of $\mathbf{p}(x; e_m)$ are all equal to ∞ and since $\det(A_m) \neq 0$ then $\infty \notin \sigma(P)$. Hence the structured pencils $\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}} \in \mathbb{L}_1(P)$ with vector e_m are linearizations of P. They are given by

$$\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}} = \begin{cases} \lambda \mathcal{A}(1) + \mathcal{B}(1) \text{ when } \mathcal{S} \in \{\text{Hermitian, symmetric}\}, \\ \lambda \mathcal{A}(-1) + \mathcal{B}(-1) \text{ when } \mathcal{S} \in \{\star\text{-even}, \star\text{-odd}\}, \end{cases}$$
(3.3)

where

$$\mathcal{A}(\varepsilon) = \begin{bmatrix} 0 & \cdots & 0 & \varepsilon^{m-1}A_m \\ \vdots & \ddots & \varepsilon^{m-2}A_{m-1} \\ \vdots & \ddots & \ddots & \vdots \\ \varepsilon^0 A_m & \varepsilon^0 A_{m-1} & \cdots & \varepsilon^0 A_1 \end{bmatrix},$$

and

$$\mathcal{B}(\varepsilon) = - \begin{bmatrix} 0 & \dots & 0 & \varepsilon^{m-1}A_m & 0 \\ \vdots & \ddots & \varepsilon^{m-2}A_m & \varepsilon^{m-2}A_{m-1} & \vdots \\ 0 & \ddots & \ddots & \vdots & \vdots \\ \varepsilon A_m & \varepsilon A_{m-1} & \cdots & \varepsilon A_2 & 0 \\ 0 & \dots & \dots & 0 & -A_0 \end{bmatrix}$$

Note that for $S \in \{\text{Hermitian, symmetric}\}, \lambda \mathcal{A}_{S} + \mathcal{B}_{S} = \lambda B_{m} - B_{m-1} \text{ that is given in}$ (2.11) and for *-(anti)palindromic $P(\lambda)$, we have $0 \notin \sigma(P)$ since $\infty \notin \sigma(P)$. When $m = 2k + 1, v = e_{k+1}$ satisfies $v = F_m v = F_m \overline{v}$ and $0, \infty$ are the only roots of the v-polynomial. The corresponding *-(anti)palindromic pencils in $\mathbb{L}_1(P)$, given by

$$\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}} = \begin{cases} \lambda \mathcal{A}^{odd} + (\mathcal{A}^{odd})^{\star} & \text{when } \mathcal{S} = \star \text{-palindromic with } m = 2k + 1, \\ \lambda \mathcal{A}^{odd} - (\mathcal{A}^{odd})^{\star} & \text{when } \mathcal{S} = \star \text{-antipalindromic with } m = 2k + 1, \end{cases}$$
(3.4)

where

$$\mathcal{A}^{odd} = \begin{bmatrix} \mathcal{A}_{11}^{odd} & \mathcal{A}_{12}^{odd} \\ \mathcal{A}_{21}^{odd} & \mathcal{A}_{22}^{odd} \end{bmatrix},\tag{3.5}$$

with $\mathcal{A}_{11}^{odd} = (\mathcal{A}_{22}^{odd})^T = \mathbf{0}_{nk \times n(k+1)}$ and

$$\mathcal{A}_{12}^{odd} = \begin{bmatrix} -A_m^{\star} & 0 & \dots & 0 \\ -A_{m-1}^{\star} & \ddots & & \vdots \\ \vdots & \ddots & \ddots & 0 \\ -A_{k+2}^{\star} & \dots & -A_{m-1}^{\star} & -A_m^{\star} \end{bmatrix}, \quad \mathcal{A}_{21}^{odd} = \begin{bmatrix} A_m & A_{m-1} & \dots & A_{k+1} \\ 0 & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & A_{m-1} \\ 0 & \dots & 0 & A_m \end{bmatrix},$$

are linearizations of $P(\lambda)$.

For \star -(anti)palindromic polynomials of even degree m = 2k, the simplest nonzero vector v satisfying $F_m v = v$ when $\star = T$ or $F_m v = \bar{v}$ when $\star = *$ is of the form

$$v = \begin{bmatrix} 0 & \cdots & 0 & z & z^{\star} & 0 & \cdots & 0 \end{bmatrix}^T,$$

where z and z^* are in positions k and k + 1, respectively. The corresponding \star -(anti)palindromic pencil in $\mathbb{L}_1(P)$ is a linearization of $P(\lambda)$ if $-z/z^*$ is not an eigenvalue of P and is given by

$$\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}} = \begin{cases} \lambda \mathcal{A}_{-}^{even}(z) + (\mathcal{A}_{-}^{even}(z))^{\star} & \text{when } \mathcal{S} = \star \text{-palindromic with } m = 2k, \\ \lambda \mathcal{A}_{-}^{even}(z) - (\mathcal{A}_{-}^{even}(z))^{\star} & \text{where } \mathcal{S} = \star \text{-antipalindromic with } m = 2k, \end{cases}$$
(3.6)

where

$$\mathcal{A}_{-}^{even}(z) = \begin{bmatrix} \mathcal{A}_{11}^{even}(z) & \mathcal{A}_{12}^{even}(z) \\ \mathcal{A}_{21}^{even}(z) & \mathcal{A}_{22}^{even}(z) \end{bmatrix},$$

with

$$\mathcal{A}_{11}^{even}(z) = z \begin{bmatrix} 0 & 0 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \dots & 0 \\ A_m & A_{m-1} & \dots & A_{k+1} \end{bmatrix}, \quad \mathcal{A}_{22}^{even}(z) = z \begin{bmatrix} A_{k+1} & 0 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ A_{m-1} & 0 & \dots & 0 \\ A_m & 0 & \dots & 0 \end{bmatrix},$$
$$\mathcal{A}_{12}^{even}(z) = - \begin{bmatrix} z^*A_0 & zA_0 & 0 & \dots & \dots & 0 \\ z^*A_1 & z^*A_0 + zA_1 & \ddots & \ddots & \ddots & \vdots \\ \vdots & \vdots & \ddots & \ddots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & \ddots & 0 \\ z^*A_{k-2} & z^*A_{k-2} + zA_{k-1} & \dots & z^*A_1 + zA_2 & z^*A_0 + zA_1 & zA_0 \\ -zA_k + z^*A_{k-1} & z^*A_{k-2} & \dots & z^*A_1 & z^*A_0 \end{bmatrix},$$

$$\mathcal{A}_{21}^{even}(z) = \begin{bmatrix} z^{\star}A_m & zA_m + z^{\star}A_{m-1} & zA_{m-1} + z^{\star}A_{m-2} & \dots & zA_{k+2} + z^{\star}A_{k+1} \\ 0 & \ddots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ \vdots & & \ddots & \ddots & \ddots & zA_{m-1} + z^{\star}A_{m-2} \\ \vdots & & & \ddots & z^{\star}A_m & zA_m + z^{\star}A_{m-1} \\ 0 & \dots & \dots & 0 & z^{\star}A_m \end{bmatrix}$$

Note that when $\star = *$, we can always pick a $z \in \mathbb{F}$ such that $-z/z^{\star} \notin \sigma(P)$. But when $\star = T$, $-z/z^{\star} = -1$ so if $-1 \in \sigma(P)$, the corresponding \star -(anti)palindromic pencil in $\mathbb{L}_1(P)$ is not a linearization of $P(\lambda)$. In fact it is shown in [69] that some T-(anti)palindromic matrix polynomials of even degree do not have T-(anti)palindromic linearizations. Instead, we allow a linearization with "anti" structure: palindromic becomes antipalindromic and vice versa. For this, let $v = \begin{bmatrix} 0 & \cdots & 0 & 1 & -1 & 0 & \cdots & 0 \end{bmatrix}^T$ satisfying $v = -F_m v$. If $P(\lambda)$ is T-palindromic then there is a unique T-antipalindromic pencil in $\mathbb{L}_1(P)$ with vector v. Similarly if $P(\lambda)$ is T-antipalindromic then there is unique T-palindromic pencil in $\mathbb{L}_1(P)$ with vector v. Such pencils are linearizations of P if $1 \notin \sigma(P)$ and are given by

$$\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}} = \begin{cases} \lambda \mathcal{A}_{+}^{even} - (\mathcal{A}_{+}^{even})^{T} & \text{when } \mathcal{S} = T\text{-palindromic with } m = 2k, \\ \lambda \mathcal{A}_{+}^{even} + (\mathcal{A}_{+}^{even})^{T} & \text{when } \mathcal{S} = T\text{-antipalindromic with } m = 2k, \end{cases}$$
(3.7)

where $\mathcal{A}^{even}_{+}(z)$ has a block structure similar to that of $\mathcal{A}^{even}_{-}(z)$ in (3.6) with z replaced by 1 and z^{\star} replaced by -1. In particular, when m = 2,

$$\mathcal{A}^{even}_{+} = \begin{bmatrix} A_2 & A_1 + A_0 \\ -A_2 & A_2 \end{bmatrix}.$$

The next result, useful later, shows that the linearizations (3.3)-(3.7) share a property.

Lemma 3.2.1 Let $S \in \mathbb{S}$ and $P(\lambda) \in \mathcal{P}_{S}(\mathbb{F}^{n})$ with nonsingular leading coefficient. If $\lambda \mathcal{A}_{S} + \mathcal{B}_{S}$ is a structured linearization of $P(\lambda)$ as in (3.3)–(3.7) then $\mathcal{C} = -\mathcal{A}_{S}^{-1}\mathcal{B}_{S}$, where \mathcal{C} is the companion form of $P(\lambda)$ given in (1.8).

Proof. Some easy calculations show that $-\mathcal{A}_{\mathcal{S}}\mathcal{C} = \mathcal{B}_{\mathcal{S}}$.

Hence, with the exception of T-(anti)palindromic matrix polynomials of even degree with both -1 and 1 as eigenvalues, the companion form of $P(\lambda)$ can be factorized as $\mathcal{C} = -\mathcal{A}_{\mathcal{S}}^{-1}\mathcal{B}_{\mathcal{S}}$, where $\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}} = \mathcal{A}_{\mathcal{S}}(\lambda I - \mathcal{C})$ is a structured linearization of $P(\lambda)$.

3.2.2 Standard triples

Recall that (U,T) is an (m,n)-standard pair over \mathbb{F} if $T \in \mathbb{F}^{mn \times mn}$ and $U \in \mathbb{F}^{n \times mn}$ are such that

$$Q = Q(U,T) := \begin{bmatrix} UT^{m-1} \\ \vdots \\ UT \\ U \end{bmatrix}$$
(3.8)

is nonsingular [62, Def. 2.1]. The triple (U, T, V) forms an (m, n)-standard triple over \mathbb{F} if (U, T) is an (m, n)-standard pair over \mathbb{F} and $V \in \mathbb{F}^{mn \times n}$ is such that $UT^{m-1}V$ is nonsingular and, if $m \geq 2$,

$$UT^{j}V = 0, \quad j = 0: m - 2,$$
 (3.9)

or equivalently,

$$QV = e_1 \otimes N \tag{3.10}$$

for some nonsingular $n \times n$ matrix N, and where e_1 is the first column of the $m \times m$ identity matrix [62, Def. 2.3]. Note that the definitions of standard pairs and triples make no reference to matrix polynomials.

An (m, n)-standard pair (U, T) over \mathbb{F} is a standard pair for $P(\lambda) = \sum_{j=0}^{m} \lambda^{j} A_{j}$ if

$$A_m U T^m + A_{m-1} U T^{m-1} + \dots + A_1 U T + A_0 U = 0, \qquad (3.11)$$

[31, p. 46]. A standard triple (U, T, V) is a standard triple for $P(\lambda)$ if (3.11) holds and $A_m = (UT^{m-1}V)^{-1}$. Any $P(\lambda) \in \mathcal{P}(\mathbb{F}^n)$ with nonsingular leading coefficient admits a standard triple. For example, it is easy to check that

$$(e_m^T \otimes I_n, \mathcal{C}, e_1 \otimes A_m^{-1}) \tag{3.12}$$

with C as in (1.8) is a standard triple for $P(\lambda)$.

Let $U_i \in \mathbb{F}^{n \times mn}$, $T_i \in \mathbb{F}^{mn \times mn}$ and $V_i \in \mathbb{F}^{mn \times n}$, i = 1, 2. Then (U_1, T_1, V_1) is

similar to (U_2, T_2, V_2) if there exists a nonsingular $G \in \mathbb{F}^{mn \times mn}$ such that

$$U_2 = U_1 G, \qquad T_2 = G^{-1} T_1 G, \qquad V_2 = G^{-1} V_1.$$

Moreover if (U_1, T_1, V_1) is a standard triple so is (U_2, T_2, V_2) [30, Prop. 12.1.3]. Note that if (U, T, V) is a standard triple for $P(\lambda)$ then

$$(e_m^T \otimes I_n)Q = U, \quad Q^{-1}\mathcal{C}Q = T, \quad e_1 \otimes A_m^{-1} = QV,$$

with Q as in (3.8). Hence any standard triple (U, T, V) for $P(\lambda)$ is similar to $(e_m^T \otimes I_n, \mathcal{C}, e_1 \otimes A_m^{-1})$. Note that because T is similar to $\mathcal{C}, \lambda I - T$ is a linearization of $P(\lambda)$ and $\sigma(P) = \sigma(T)$. The following result [30, Thm. 12.1.4] will be useful.

Lemma 3.2.2 Let $U \in \mathbb{F}^{n \times mn}$, $T \in \mathbb{F}^{mn \times mn}$, $V \in \mathbb{F}^{mn \times n}$ and let $P(\lambda) \in \mathcal{P}(\mathbb{F}^n)$ be of degree m with nonsingular leading coefficient. Then (U, T, V) is a standard triple for $P(\lambda)$ if and only if $P(\lambda)^{-1} = U(\lambda I - T)^{-1}V$ for $\lambda \in \mathbb{C} \setminus \sigma(P)$.

A Jordan triple (X, J, Y) over \mathbb{F} for $P(\lambda)$ is a standard triple for $P(\lambda)$ for which the matrix J is in Jordan form or real Jordan form if $\mathbb{F} = \mathbb{R}$. By (3.11) and [31, Prop. 2.1], we have that $\sum_{j=0}^{m} A_j X J^j = 0$ and $\sum_{j=0}^{m} J^j Y A_j = 0$. The columns of Xand Y^* determine right and left eigenvectors and generalized eigenvectors of $P(\lambda)$. The matrix J is the Jordan form of the companion form \mathcal{C} of $P(\lambda)$. In a similar way that a Jordan pair (X, J) with X nonsingular generates $A - \lambda I$ via $A = XJX^{-1}$, $I = XX^{-1}$, an (m, n)-Jordan triple (X, J, Y) over \mathbb{F} generates a matrix polynomial $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j \in \mathcal{P}(\mathbb{F}^n)$ uniquely, where

$$A_m = (XJ^{m-1}Y)^{-1}, (3.13)$$

and for $1 \leq j \leq m$,

$$A_{m-j} = -A_m \sum_{i=m-j+1}^m X J^{i+j-1} Y A_i = -\sum_{i=m-j+1}^m A_i X J^{i+j-1} Y A_m.$$
(3.14)

These formulae follow directly from [58, Thm. 14.7.1] on using [58, Thm. 14.2.5], and a similar argument to that in the proof of [49, Thm. 1] can be used to prove the uniqueness of $P(\lambda)$ (see also [62, Thm. 2.4]).

3.3 *S*-structured standard triples

We now consider standard triples in the context of structured matrix polynomials. We start by listing two assumptions used in our analysis. Let $S \in \mathbb{S}$, $P(\lambda) \in \mathcal{P}_{S}(\mathbb{F}^{n})$ have degree m with nonsingular leading coefficient and let $T \in \mathbb{F}^{mn \times mn}$.

Assumption (a): if $S \in \{T\text{-palindromic}, T\text{-antipalindromic}\}$ and $P(\lambda)$ has degree m = 2k then either $-1 \notin \sigma(P)$ or $1 \notin \sigma(P)$.

Assumption (b): if $S \in \{T\text{-palindromic}, T\text{-antipalindromic}\}$ and m = 2k then either $-1 \notin \sigma(T)$ or $1 \notin \sigma(T)$.

These two assumptions are equivalent when $\lambda I - T$ is a linearization of $P(\lambda)$. For some T satisfying assumption (b) we define $u_{\mathcal{S}}(T), t_{\mathcal{S}}(T), v_{\mathcal{S}}(T)$ as in Table 3.2. Note that assumption (b) ensures the existence of $\alpha \in \mathbb{F}$ such that $\alpha^* \alpha = 1$ and $-\alpha \notin \sigma(T)$. Also, for *-(anti)palindromic structures, the eigenvalues of T come in pairs (λ, λ^{-*}) . Hence $0 \notin \sigma(T)$ since $\infty \notin \sigma(T)$ and T^{-*} is well defined.

Before stating our main result in Theorem 3.3.5, we provide a few lemmas and introduce the notion of S-structured standard triple. The first lemma of this section extends to all structures in S a result in [58, Cor. 14.2.1] for Hermitian structure.

Table 3.2: Definition of $u_{\mathcal{S}}(T), t_{\mathcal{S}}(T), v_{\mathcal{S}}(T)$ for some $T \in \mathbb{F}^{mn \times mn}$ satisfying assumption (b), where α is some scalar in \mathbb{F} such that $\alpha^* \alpha = 1$ and $-\alpha \notin \sigma(T)$.

Structure \mathcal{S}	$u_{\mathcal{S}}(T)$	$t_{\mathcal{S}}(T)$	$v_{\mathcal{S}}(T)$
Hermitian/symmetric	Ι	T^{\star}	Ι
★ -even	-I	$-T^{\star}$	Ι
*-odd	Ι	$-T^{\star}$	Ι
*-palindromic, $m = 2k + 1$	$-T^{\star(k-1)}$	$T^{-\star}$	$T^{\star k}$
*-palindromic, $m = 2k$	$-T^{\star(k-1)}(I+\alpha T^{\star})^{-1}$	$T^{-\star}$	$(I + \alpha T^{\star})T^{\star(k-1)}$
\star -antipalindromic, $m = 2k + 1$	$T^{\star(k-1)}$	$T^{-\star}$	$T^{\star k}$
$\star\text{-antipalindromic}m=2k$	$T^{\star(k-1)}(I + \alpha T^{\star})^{-1}$	$T^{-\star}$	$(I + \alpha T^{\star})T^{\star(k-1)}$

Lemma 3.3.1 Let (U, T, V) be an (m, n)-standard triple for $P(\lambda) \in \mathcal{P}(\mathbb{F}^n)$ with nonsingular leading coefficient and let $\mathcal{S} \in \mathbb{S}$. Assume that T satisfies assumption (b). Then $P(\lambda)$ has structure \mathcal{S} if and only if $(V^{\star}u_{\mathcal{S}}(T), t_{\mathcal{S}}(T), v_{\mathcal{S}}(T)U^{\star})$ is a standard triple for $P(\lambda)$. **Proof.** The proof for $S \in \{\text{Hermitian, symmetric, }\star\text{-even, }\star\text{-odd}\}$ is easy to obtain using the resolvent form for $P(\lambda)$ given in Lemma 3.2.2 and the definition of the structures in Table 1.1. See also [58, Cor. 14.2.1] for the Hermitian structure.

Now suppose that $P(\lambda)$ is *-palindromic. Since any standard triple for $P(\lambda)$ is similar to $(e_m^T \otimes I_n, \mathcal{C}, e_1 \otimes A_m^{-1}) =: (\widetilde{U}, \mathcal{C}, \widetilde{V})$, it suffices to show that this standard triple is similar to $(\widetilde{V}^* u_{\mathcal{S}}(\mathcal{C}), \mathcal{C}^{-*}, v_{\mathcal{S}}(\mathcal{C})\widetilde{U}^*)$. We need to consider three cases:

(i) m = 2k + 1. The pencil $\lambda \mathcal{A}^{odd} + (\mathcal{A}^{odd})^*$ with \mathcal{A}^{odd} as in (3.5) is a linearization of $P(\lambda)$. By Lemma 3.2.1, $\mathcal{C} = -(\mathcal{A}^{odd})^{-1}(\mathcal{A}^{odd})^*$. So if we let $G^{-1} = \mathcal{A}^{odd}$ then

$$\widetilde{V}^{\star}u_{\mathcal{S}}(\mathcal{C}) = -(e_1 \otimes A_m^{-1})^{\star}(\mathcal{C}^{\star})^{k-1} = \widetilde{U}G, \quad G^{-1}\mathcal{C}G = \mathcal{C}^{-\star} = t_{\mathcal{S}}(\mathcal{C}),$$

and

$$G^{-1}\widetilde{V} = G^{-1}(e_1 \otimes A_m^{-1}) = e_{2k} \otimes I = (\mathcal{C}^{\star})^k (e_m \otimes I) = v_{\mathcal{S}}(\mathcal{C})\widetilde{U}^{\star}.$$

(ii) $m = 2k, \star = T$ and $-1 \in \sigma(T)$. From assumption (b) it follows that $1 \notin \sigma(T)$ so we can take $\alpha = -1$ in the definition of $u_{\mathcal{S}}$ and $v_{\mathcal{S}}$. The pencil $\lambda \mathcal{A}^{even}_{+} - (\mathcal{A}^{even}_{+})^{\star}$ with \mathcal{A}^{even}_{+} as in (3.7) is a linearization of $P(\lambda)$. By Lemma 3.2.1, $\mathcal{C} = -(\mathcal{A}^{even}_{+})^{-1}(\mathcal{A}^{even}_{+})^{\star}$. If we let $G^{-1} = \mathcal{A}^{even}_{+}$ then as in (i), $G^{-1}\mathcal{C}G = t_{\mathcal{S}}(\mathcal{C})$. Also,

$$v_{\mathcal{S}}(\mathcal{C})U^{T} = (I - \mathcal{C}^{T})\mathcal{C}^{T(k-1)}(e_{m} \otimes I_{n})$$
$$= e_{k+1} \otimes I - e_{k} \otimes I$$
$$= -G^{-1}(e_{1} \otimes I)A_{m}^{-1} = -G^{-1}\widetilde{V}$$

From $\widetilde{V} = -Gv_{\mathcal{S}}(\mathcal{C})\widetilde{U}^T$ it follows that $\widetilde{V}^T = -\widetilde{U}\mathcal{C}^{(k-1)}(I-\mathcal{C})G^T$ so that

$$\begin{split} \widetilde{V}^T u_{\mathcal{S}}(\mathcal{C}) &= \widetilde{U}\mathcal{C}^{(k-1)}(I-\mathcal{C})G^T\mathcal{C}^{T(k-1)}(I-\mathcal{C}^T)^{-1} \\ &= \widetilde{U}\mathcal{C}^{(k-1)}(I-\mathcal{C})\mathcal{C}^{(1-k)}G^T(I-\mathcal{C}^T)^{-1} \\ &= -\widetilde{U}G(I-\mathcal{C}^T)(I-\mathcal{C}^T)^{-1} = -\widetilde{U}G, \end{split}$$

where we used $G^T \mathcal{C}^{T(k-1)} G^{-T} = \mathcal{C}^{-(k-1)}$ and $\mathcal{C} G^T = -G$.

(iii) $m = 2k, \star = *, T$ and $-1 \notin \sigma(T)$. In this case Lemma 3.2.1 says that $\mathcal{C} =$

$$-(\mathcal{A}^{even}_{-}(z))^{-1}(\mathcal{A}^{even}_{-}(z))^{\star}$$
 with $\mathcal{A}^{even}_{-}(z)$ as in (3.6). The proof is similar to that in (ii) with $\alpha = z/z^{\star}$.

Conversely, suppose that (U, T, V) and $(V^* u_{\mathcal{S}}(T), T^{-*}, v_{\mathcal{S}}(T)U^*)$ are standard triples for $P(\lambda)$. Assumption (b) guarantees the existence of $\alpha \in \mathbb{F}$ such that $\alpha^* \alpha = 1$ and $I + \alpha T^*$ is nonsingular. Hence $u_{\mathcal{S}}$ and $v_{\mathcal{S}}$ are well defined. Using the resolvent form for $P(\lambda)$, $P(\lambda)^{-1} = U(\lambda I - T)^{-1}V$, we obtain

$$\lambda^{-m}(P(\lambda^{-\star}))^{-\star} = \lambda^{-m}(U(\lambda^{-\star}I - T)^{-1}V)^{\star} = \lambda^{1-m}V^{\star}(I - \lambda T^{\star})^{-1}U^{\star}$$

If $\rho(\lambda T^{\star}) < 1$ then

$$(I - \lambda T^{\star})^{-1} = I + \lambda T^{\star} + \lambda^2 T^{\star 2} + \cdots .$$
(3.15)

Using (3.15) and the fact that $V^{\star}T^{\star j}U^{\star} = 0, \ j = 0; \ m - 2$ (see (3.9)), we obtain

$$\lambda^{-m}(P(\lambda^{-\star}))^{-\star} = V^{\star}T^{\star(m-1)}(I + \lambda T^{\star} + \lambda^{2}T^{\star 2} + \cdots)U^{\star}$$

$$= V^{\star}T^{\star(k-1)}(I - \lambda T^{\star})^{-1}T^{\star(m-k)}U^{\star}$$

$$= -V^{\star}T^{\star(k-1)}(\lambda I - T^{-\star})^{-1}T^{\star(m-k-1)}U^{\star} \qquad (3.16)$$

for all $|\lambda| < (\rho(T))^{-1}$. When m = 2k + 1, (3.16) reads as

$$\lambda^{-m}(P(\lambda^{-\star}))^{-\star} = V^{\star}u_{\mathcal{S}}(T)(\lambda I - T^{-\star})^{-1}v_{\mathcal{S}}(T)U^{\star} = P(\lambda)^{-1}$$

Note that $(\lambda I - T^{-\star})^{-1}$ commutes with $T^{\star k-1}$, $(I + \alpha T^{\star})$ and $(I + \alpha T^{\star})^{-1}$ so when m = 2k, (3.16) can be rewritten as

$$\lambda^{-m}(P(\lambda^{-\star}))^{-\star} = V^{\star}u_{\mathcal{S}}(T)(\lambda I - T^{-\star})^{-1}v_{\mathcal{S}}(T)U^{\star} = P(\lambda)^{-1}$$

for all $|\lambda| < (\rho(T))^{-1}$. Since $\lambda^{-m}(P(\lambda^{-\star}))^{-\star} = P(\lambda)^{-1}$ holds for many values of λ , $P(\lambda) = \lambda^m P^{\star}(\lambda^{-1})$ for all λ , that is, $P(\lambda)$ is \star -palindromic.

The results for the *-antipalindromic structure are proved in a similar way.

In the proof of Lemma 3.3.1 we use the fact that if (U, T, V) is a standard triple for a structured $P(\lambda)$ then there exists a nonsingular S such that

$$US = V^{\star} u_{\mathcal{S}}(T), \quad S^{-1}TS = t_{\mathcal{S}}(T), \quad S^{-1}V = v_{\mathcal{S}}(T)U^{\star}.$$
 (3.17)

These relations imply certain properties of S, which we use in our definition of S-structured standard triples.

Definition 3.3.2 (S-structured standard triple) Let $S \in S$. An (m, n)-standard triple (U, T, V) with T satisfying assumption (b) is said to be S-structured if $V = Sv_{S}(T)U^{\star}$ for some nonsingular $S \in \mathbb{F}^{mn \times mn}$ having the following properties:

- $S = S^*, TS = (TS)^*$ when $S \in \{Hermitian, symmetric\},\$
- $S = -S^*$, $TS = (TS)^*$ when $S = \star$ -even,
- $S = S^*$, $TS = -(TS)^*$ when $S = \star \text{-odd}$,
- $TS^{\star} = -S$ when $S = \star$ -palindromic and m = 2k + 1 or $TS^{\star} = -\alpha S$ when $S = \star$ -palindromic and m = 2k,
- $TS^{\star} = S$ when $S = \star$ -antipalindromic and m = 2k + 1 or $TS^{\star} = \alpha S$ when $S = \star$ -antipalindromic and m = 2k,

for some $\alpha \in \mathbb{F}$ such that $\alpha^* \alpha = 1$ and $-\alpha \notin \sigma(T)$.

We refer to the matrix S in Definition 3.3.2 as the S-matrix of the S-structured standard triple (U, T, V). We point out that Hermitian and symmetric structured standard triples are called self-adjoint standard triples in the literature (see for example [30, p. 244]). For \star -(anti)palindromic structures, the matrix T in Definition 3.3.2 is S^{-1} -unitary, that is, $T^{\star}S^{-1}T = S^{-1}$, and with additional constraints on T's structure, Lancaster, Prells and Rodman refer to (U, T, V) as a *unitary standard triple* [55, Def. 4]. Hence a unitary standard triple is S-structured but the converse is not true in general.

Our definition of \mathcal{S} -structured standard triples is justified by the next lemma.

Lemma 3.3.3 Let $S \in S$. An (m, n)-standard triple (U, T, V) with T satisfying assumption (b) is S-structured if and only if it is similar to $(V^{\star}u_{\mathcal{S}}(T), t_{\mathcal{S}}(T), v_{\mathcal{S}}(T)U^{\star})$. **Proof.** The proof of this lemma appears in [60, Thm. 3.4] for the symmetric structure and the proof there extends easily to structures $S \in \{\text{Hermitian}, \star\text{-even}, \star\text{-odd}\}$.

Suppose $S = \star$ -palindromic (the proof for \star -antipalindromic structure is similar and so we omit it). If m = 2k + 1 and (U, T, V) is S-structured, then there exists Snonsingular such that $TS^{\star} = -S$ and $V = ST^{\star k}U^{\star}$. Hence $S^{-1}TS = T^{-\star}$ and

$$US = V^{\star}S^{-\star}T^{-k}S = -V^{\star}S^{-\star}T^{-(k-1)}S^{\star} = -V^{\star}T^{\star(k-1)}$$

Hence (U, T, V) is similar to $(-V^{\star}T^{\star k-1}, T^{-\star}, T^{\star m-k-1}U^{\star})$.

When m = 2k, $TS^* = -\alpha S$ and $V = S(I + \alpha T^*)T^{*(k-1)}U^*$ for some $\alpha \in \mathbb{F}$ such that $\alpha^* \alpha = 1$ and $-\alpha \notin \sigma(T)$. The first equality implies that $S^{-1}TS = T^{-*}$ while the first and second equality yield

$$US = V^* S^{-*} T^{-(k-1)} (I + \alpha^* T)^{-1} S$$

= $V^* T^{*(k-1)} S^{-*} (I + \alpha^* T)^{-1} S$
= $-V^* T^{*(k-1)} (I + \alpha T^*)^{-1} = V^* u_{\mathcal{S}}(T) S$

Conversely, if (U, T, V) is similar to $(V^* u_{\mathcal{S}}(T), T^{-*}, v_{\mathcal{S}}(T)U^*)$ then there exists S nonsingular such that (3.17) holds with $t_{\mathcal{S}}(T) = T^{-*}$. It remains to show that $TS^* = -S$ when m = 2k + 1 and $TS^* = -\alpha S$ when m = 2k. If m = 2k + 1, the first and last equalities in (3.17) imply that $V = -ST^{*k}S^{-*}T^{k-1}V$ and since V has full rank, $-ST^{*k}S^{-*}T^{k-1} = I$, which on using the second equality in (3.17) yields $TS^* = -S$. When m = 2k, the first and last equalities in (3.17) and the fact that V has full rank imply that

$$-ST^{\star(k-1)}(I + \alpha T^{\star})S^{-\star}(I + \alpha^{\star}T)^{-1}T^{(k-1)} = I$$

$$\Leftrightarrow -T^{-(k-1)}(I + \alpha T^{-1})SS^{-\star}(I + \alpha^{\star}T)^{-1}T^{(k-1)} = I$$

$$\Leftrightarrow -T^{-(k-1)}SS^{-\star}(\alpha T^{-1})T^{(k-1)} = I$$

$$\Leftrightarrow -\alpha SS^{-\star} = T^{(k-1)}T^{-(k-2)} = T,$$

which completes the proof. \blacksquare

The next lemma shows that any standard triple that is similar to an \mathcal{S} -structured standard triple is itself \mathcal{S} -structured.

Lemma 3.3.4 Let (U, T, V) be a standard triple similar to (U_1, T_1, V_1) , that is,

$$(U_1, T_1, V_1) = (UG, G^{-1}TG, G^{-1}V)$$

for some nonsingular matrix G. Let $S \in S$ and assume that T satisfies assumption (b). If (U, T, V) is S-structured with matrix S then (U_1, T_1, V_1) is S-structured with matrix $S_1 = G^{-1}SG^{-\star}$.

Proof. It is easy to check that $V_1 = S_1 v_{\mathcal{S}}(T_1) U_1^*$ and since the properties of S are preserved by \star -congruence, (U_1, T_1, V_1) is \mathcal{S} -structured with matrix S_1 .

We can now state our main result, which is a direct consequence of Lemma 3.3.1, Lemma 3.3.3 and Lemma 3.3.4. It extends a result for Hermitian structure [30, Thm. 12.2.2] to all structures in S.

Theorem 3.3.5 Let $S \in S$ and $P(\lambda) \in \mathcal{P}(\mathbb{F}^n)$ with nonsingular leading coefficient satisfying assumption (a). Then $P(\lambda)$ has structure S if and only if $P(\lambda)$ admits an S-structured standard triple, in which case every standard triple for $P(\lambda)$ is Sstructured.

The structure of the matrix S in an S-structured standard triple is uniquely determined by the triple, as shown by the next result.

Proposition 3.3.6 Let $S \in S$ and (U, T, V) be an S-structured standard triple with matrix S. Then

$$S = Q(U,T)^{-1}Q(V^{\star}u_{\mathcal{S}}(T), t_{\mathcal{S}}(T)).$$

with Q(U,T) as in (3.8).

Proof. Using Definition 3.3.2 we check that $Q(U,T)S = Q(V^{\star}u_{\mathcal{S}}(T), t_{\mathcal{S}}(T))$.

The matrix S is also easy to construct when the matrix coefficients of $P(\lambda)$ are known.

Proposition 3.3.7 Let $S \in \mathbb{S}$ and $P(\lambda) \in \mathcal{P}_{S}(\mathbb{F}^{n})$ of degree m with nonsingular leading coefficient be satisfying assumption (a). If (U,T) is a standard pair for $P(\lambda)$ then $(U,T, Sv_{S}(T)U^{\star})$ is an S-structured standard triple for $P(\lambda)$ with matrix S given by

$$S^{-1} = \begin{cases} Q^T \mathcal{A}^{even}_+ Q & \text{if } P \text{ is } T\text{-}(anti) palindromic, \ m = 2k \text{ and } -1 \in \sigma(P), \\ z^{-\star} Q^{\star} \mathcal{A}^{even}_-(z) Q & \text{if } P \text{ is } \star\text{-}(anti) palindromic, \ m = 2k, \ -z/z^{\star} \notin \sigma(P), \\ Q^{\star} \mathcal{A}_{S} Q & \text{otherwise,} \end{cases}$$

where Q := Q(U,T) is as in (3.8), and $\mathcal{A}_{\mathcal{S}}$, $\mathcal{A}^{even}_{-}(z)$ and \mathcal{A}^{even}_{+} are as in (3.3)–(3.7).

Proof. We first show that the matrix S in the proposition has the properties listed in Definition 3.3.2. Note that under assumption (a), $P(\lambda)$ has a structured linearization $\lambda \mathcal{A}_{S} + \mathcal{B}_{S}$, which is one of (3.3)–(3.7). The pair (Q, T) is a standard pair for $\lambda \mathcal{A}_{S} + \mathcal{B}_{S}$, and hence

$$Q^{\star}\mathcal{A}_{\mathcal{S}}QT = -Q^{\star}\mathcal{B}_{\mathcal{S}}Q \Leftrightarrow Q^{\star}\mathcal{B}_{\mathcal{S}}Q = -z^{\star}S^{-1}T,$$

where z = 1 except when $\mathcal{A}_{\mathcal{S}} = \mathcal{A}^{even}_{-}(z)$, in which case z is such that $-z/z^* \notin \sigma(P)$. Since *-congruence preserves any structure in S, the pencil

$$Q^{\star}(\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}})Q = z^{\star}(\lambda S^{-1} - S^{-1}T)$$

has the same structure as $\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}}$, and hence S satisfies the appropriate properties.

It remains to show that V in (3.10), for which (U, T, V) is a standard triple for $P(\lambda) \in \mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$, has the form $V = Sv_{\mathcal{S}}(T)U^{\star}$. For $\mathcal{S} \in \{\text{Hermitian, symmetric, } \star\text{-even, } \star\text{-odd}\}, v_{\mathcal{S}}(T) = I$. Since $U^{\star} = Q^{\star}(e_m \otimes I_n)$ and $S^{-1} = Q^{\star}\mathcal{A}_{\mathcal{S}}Q$, we find that

$$SU^{\star} = Q^{-1}\mathcal{A}_{\mathcal{S}}^{-1}Q^{-\star}Q^{\star}(e_m \otimes I_n) = Q^{-1}\mathcal{A}_{\mathcal{S}}^{-1}(e_m \otimes I_n).$$

From the block structure of $\mathcal{A}_{\mathcal{S}}$ in (3.3) we see that $\mathcal{A}_{\mathcal{S}}(e_1 \otimes I_n) = (e_m \otimes I)A_m$, or equivalently, $\mathcal{A}_{\mathcal{S}}^{-1}(e_m \otimes I) = (e_1 \otimes I_n)A_m^{-1}$ since $\mathcal{A}_{\mathcal{S}}$ and A_m are both nonsingular. Hence $SU^{\star} = Q^{-1}(e_1 \otimes I_n)A_m^{-1} = V$.

When $P(\lambda)$ is \star -(anti)palindromic of odd degree then the definition of V in (3.10), the expression for S in the proposition and the structure of $\mathcal{A}_{\mathcal{S}} = \mathcal{A}^{odd}$ in (3.4) yield

$$S^{-1}V = Q^{\star} \mathcal{A}_{\mathcal{S}} Q Q^{-1} (e_1 \otimes I_n) A_m^{-1} = T^{\star k} U^{\star}.$$

For a \star -palindromic P of even degree, we have shown that $TS^{\star} = -\alpha S$, where $\alpha = -z/z^{\star}$. From the definition of Q in (3.8), $T^{\star(k-1)}U^{\star} = Q^{\star}(e_{k+1} \otimes I_n)$. Hence on
using the definition of V in (3.10), the expression for S in the proposition and the structure of $\mathcal{A}_{\mathcal{S}} = \mathcal{A}^{even}_{-}(z)$ in (3.6) we have that

$$QS(I + \alpha T^{\star})T^{\star(k-1)}U^{\star} = Q(S - S^{\star})Q^{\star}(e_{k+1} \otimes I_n)$$

= $z^{\star} (\mathcal{A}^{even}_{-}(z))^{-1}(e_{k+1} \otimes I_n) - z (\mathcal{A}^{even}_{-}(z))^{-\star}(e_{k+1} \otimes I_n).$

From the definition of $\mathcal{A}^{even}_{-}(z)$ in (3.6), we find that

$$\mathcal{A}_{-}^{even}(z) \begin{bmatrix} z^{-1}A_{m}^{-1} - z^{m-1}P(z, -z^{\star})^{-1} \\ (-z)^{m-2}z^{\star}P(z, -z^{\star})^{-1} \\ (-z)^{m-3}z^{\star 2}P(z, -z^{\star})^{-1} \\ \vdots \\ (-z)^{0}z^{\star(m-1)}P(z, -z^{\star})^{-1} \end{bmatrix} = -\alpha^{-1}e_{k+1} \otimes I_{n},$$
$$\left(\mathcal{A}_{-}^{even}(z)\right)^{\star} \begin{bmatrix} -z^{m-1}(z^{\star})^{0}P(z, -z^{\star})^{-1} \\ (-z)^{m-2}z^{\star}P(z, -z^{\star})^{-1} \\ (-z)^{m-3}z^{\star 2}P(z, -z^{\star})^{-1} \\ \vdots \\ (-z)^{0}z^{\star(m-1)}P(z, -z^{\star})^{-1} \end{bmatrix} = e_{k+1} \otimes I_{n},$$

where $P(z, -z^{\star}) = \sum_{j=0}^{m} z^{j} (-z^{\star})^{m-j} A_{j}$. Hence,

$$z^{\star} \big(\mathcal{A}^{even}_{-}(z) \big)^{-1} (e_{k+1} \otimes I_n) - z \big(\mathcal{A}^{even}_{-}(z) \big)^{-\star} (e_{k+1} \otimes I_n) = (e_1 \otimes I_n) \mathcal{A}^{-1}_m,$$

that is, $S(I+\alpha T^{\star})T^{\star(k-1)}U^{\star} = Q^{-1}(e_1 \otimes I_n)A_m^{-1} = V$. The proof for \star -antipalindromic P is along the same lines.

It follows from Theorem 3.3.5 and Proposition 3.3.7 that if $P(\lambda)$ has structure S then the standard triple (3.12) is S-structured with matrix $S = \mathcal{A}_{S}^{-1}$ except when $\mathcal{A}_{S} = \mathcal{A}_{-}^{even}(z)$ in which case $S = z^{\star}(\mathcal{A}_{-}^{even}(z))^{-1}$.

3.4 S-structured Jordan triples

We now explain how to obtain explicit expressions for the Jordan matrix and S-matrix of S-structured Jordan triples $(X, J, S_J v_S(J)X^*)$ for $P(\lambda) \in \mathcal{P}_S(\mathbb{F}^n)$. We note that the matrix S_J displays the sign characteristic of $P(\lambda)$. Indeed, if $(U, T, S_T v_S(T)U^*)$ is a standard triple for $P(\lambda) \in \mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$, then the sign characteristic of $P(\lambda)$ can be defined as the sign characteristic of the pair (T, S_T^{-1}) , which is a list of signs, with a sign (+1 or -1) attached to each partial multiplicity of

- real eigenvalues of Hermitian or real symmetric matrix polynomials,
- purely imaginary eigenvalues of *-even, *-odd, real *T*-even and real *T*-odd matrix polynomials, and
- eigenvalues with unit modulus of *-(anti)palindromic and real *T*-(anti)palindromic matrix polynomials.

These signs can be read off the canonical decomposition of $\lambda S_T^{-1} - S_T^{-1}T$ via \star congruence (see [30, Section 12.4] for Hermitian structure). Note that the definition
of the sign characteristic for $P(\lambda)$ is independent of the choice of standard triple.
Indeed if $(U_i, T_i, S_{T_i}v_S(T_i)U_i^{\star})$, i = 1, 2 are *S*-structured standard triples for $P(\lambda)$,
then by Lemma 3.3.4 there exists a nonsingular *G* such that $T_2 = G^{-1}T_1G$ and $S_{T_2} = G^{-1}S_{T_1}G^{-\star}$. Hence, $\lambda S_{T_2}^{-1} - S_{T_2}^{-1}T_2 = G^{\star}(\lambda S_{T_1}^{-1} - S_{T_1}^{-1}T_1)G$, that is, the pencils $\lambda S_{T_i}^{-1} - S_{T_i}^{-1}T_i$, i = 1, 2 are \star -congruent. They share the same canonical decomposition
via \star -congruence and therefore the same sign characteristic.

We know that the triple $((e_m^T \otimes I_n), \mathcal{C}, (e_1 \otimes A_m^{-1}))$ is a standard triple for $P(\lambda)$ and by Theorem 3.3.5, it is \mathcal{S} -structured with S-matrix as in Proposition 3.3.7 with $Q = I_{mn}$, say $S_{\mathcal{C}} = z^* \mathcal{A}_{\mathcal{S}}^{-1}$. Hence, on using Lemma 3.2.1, we find that

$$\lambda S_{\mathcal{C}}^{-1} - S_{\mathcal{C}}^{-1} \mathcal{C} = \lambda z^{-\star} \mathcal{A}_{\mathcal{S}} + z^{-\star} \mathcal{B}_{\mathcal{S}},$$

where $\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}}$ is a structured linearization of $P(\lambda)$ as in (3.3)–(3.7), and z = 1except when $\mathcal{A}_{\mathcal{S}} = \mathcal{A}^{even}_{-}(z)$, in which case $z \in \mathbb{F}$ is chosen such that $-z/z^* \notin \sigma(P)$. So what we need is a canonical decomposition of $\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}}$ via \star -congruence,

$$Z^{\star}(\lambda \mathcal{A}_{\mathcal{S}} + \mathcal{B}_{\mathcal{S}})Z = \lambda(Z^{\star}\mathcal{A}_{\mathcal{S}}Z) - (Z^{\star}\mathcal{A}_{\mathcal{S}}Z)(Z^{-1}\mathcal{C}Z) = z^{\star}(\lambda S_{J}^{-1} - S_{J}^{-1}J),$$

where $J = Z^{-1}CZ$ is the Jordan form of C. Fortunately, such decompositions are available in the literature for all the structures in S. We use these canonical decompositions to provide explicit expressions for J and S_J in Section 3.5. These expressions show that S_J and J have the same block structure and that we can read the sign characteristic of $P(\lambda)$ from certain diagonal blocks of S_J .

3.5 Explicit expressions for J and S_J

Using the canonical decompositions of structured pencils via \star -congruences, we provide in this section an explicit expression for the Jordan matrix and S-matrix of S-structured Jordan triples $(X, J, S_J v_S(J) X^{\star})$ of $P(\lambda) \in \mathcal{P}_S(\mathbb{F}^n)$ for each $S \in \mathbb{S}$. We assume that $P(\lambda)$ is of degree m with nonsingular leading coefficient matrix. To facilitate the description of J and S_J , we introduce the matrices $F_1 = E_1 = [1]$ and for integers k > 1

$$E_{k} = \begin{bmatrix} & & 1 \\ & -1 & \\ & \ddots & \\ & 1 & \\ (-1)^{k-1} & & \end{bmatrix}_{k \times k} = (-1)^{k-1} E_{k}^{T}, \qquad F_{k} = \begin{bmatrix} & & 1 \\ & \ddots & \\ & 1 & & \end{bmatrix}_{k \times k}.$$

We denote by

$$J_{\ell_k}(\lambda_k) = \begin{bmatrix} \lambda_k & 1 & & \\ & \lambda_k & \ddots & \\ & & \ddots & 1 \\ & & & \lambda_k \end{bmatrix} \in \mathbb{C}^{\ell_k \times \ell_k}, \tag{3.18}$$

the Jordan block of size ℓ_k associated with λ_k , and by

$$K_{2m_k}(\lambda_k, \bar{\lambda}_k) = K_{2m_k}(\Lambda_k) = \begin{bmatrix} \Lambda_k & I & & \\ & \Lambda_k & \ddots & \\ & & \ddots & I \\ & & & \Lambda_k \end{bmatrix} \in \mathbb{R}^{2m_k \times 2m_k}, \quad \Lambda_k = \begin{bmatrix} \alpha_k & \beta_k \\ -\beta_k & \alpha_k \end{bmatrix},$$
(3.19)

the $2m_k \times 2m_k$ real Jordan block associated with the pair of complex conjugate eigenvalues $(\lambda_k, \bar{\lambda}_k)$, where $\lambda_k = \alpha_k + i\beta_k$ with $\alpha_k, \beta_k \in \mathbb{R}, \ \beta_k \neq 0$. We use the notation $\bigoplus_{j=1}^r F_j$ to denote the direct sum of the matrices F_1, \ldots, F_r .

Note that there are restrictions on the Jordan structure of P. For instance, a regular $n \times n$ matrix polynomial cannot have more than n elementary divisors associated with the same eigenvalue [64]. Also, the elementary divisors have certain pairing, which depends on the structure $\mathcal{S} \in \mathbb{S}$ and the eigenvalue. Hence we describe for each $\mathcal{S} \in \mathbb{S}$ the elementary divisors arising from $P(\lambda) \in \mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$ and then provide an expression for J and S_J .

3.5.1 Hermitian structure

Suppose $P(\lambda)$ is Hermitian with

- r real elementary divisors $(\lambda \lambda_j)^{\ell_j}$, j = 1: r, and
- s pairs of nonreal conjugate elementary divisors $(\lambda \mu_j)^{m_j}$, $(\lambda \overline{\mu}_j)^{m_j}$, j = 1: s,

with ℓ_j, m_j such that $\sum_{j=1}^r \ell_j + 2 \sum_{j=1}^s m_j = mn$. It follows from [56, Thm. 6.1] that

$$J = \bigoplus_{j=1}^r J_{\ell_j}(\lambda_j) \oplus \bigoplus_{j=1}^s (J_{m_j}(\bar{\mu}_j) \oplus J_{m_j}(\mu_j)), \quad S_J = S_J^{-1} = \bigoplus_{j=1}^r \varepsilon_j F_{\ell_j} \oplus \bigoplus_{j=1}^s F_{2m_j}.$$

Here $\{\varepsilon_1, \ldots, \varepsilon_r\}$ with $\varepsilon_j = \pm 1$ is the sign characteristic associated with the real eigenvalues λ_j , j = 1: r of $P(\lambda)$. We easily check that $S_J = S_J^*$ and $JS_J = (JS_J)^*$.

3.5.2 Real symmetric structure

Suppose $P(\lambda)$ is real symmetric with

- r real elementary divisors $(\lambda \lambda_j)^{\ell_j}$, j = 1: r, and
- s pairs of nonreal conjugate elementary divisors $(\lambda \mu_j)^{m_j}$, $(\lambda \overline{\mu}_j)^{m_j}$, j = 1: s,

with ℓ_j, m_j such that $\sum_{j=1}^r \ell_j + 2 \sum_{j=1}^s m_j = mn$. On using [56, Thm. 9.2] we find that

$$J = \bigoplus_{j=1}^r J_{\ell_j}(\lambda_j) \oplus \bigoplus_{j=1}^s K_{2m_j}(\mu_j, \bar{\mu}_j), \qquad S_J = S_J^{-1} = \bigoplus_{j=1}^r \varepsilon_j F_{\ell_j} \oplus \bigoplus_{j=1}^s F_{2m_j},$$

where the scalars $\varepsilon_j = \pm 1$ form the sign characteristic associated with the real eigenvalues of $P(\lambda)$. Note that $S_J = S_J^T$ and $JS_J = (JS_J)^T$.

3.5.3 Complex symmetric structure

Suppose $P(\lambda)$ is complex symmetric with q elementary divisors $(\lambda - \lambda_j)^{m_j}$, $\lambda_j \in \mathbb{C}$, j = 1; q, with m_j such that $\sum_{j=1}^q m_j = mn$. Then [83, Prop. 4.3] leads to

$$J = \bigoplus_{j=1}^{q} J_{m_j}(\lambda_j), \qquad S_J = S_J^{-1} = \bigoplus_{j=1}^{q} F_{m_j},$$

which satisfy $S_J = S_J^T$ and $JS_J = (JS_J)^T$.

3.5.4 *-even structure

Suppose $P(\lambda)$ is *-even with

- r purely imaginary (including 0) elementary divisors $(\lambda i\beta_j)^{\ell_j}$, j = 1: r, and
- s pairs of nonzero and non-purely imaginary elementary divisors $(\lambda i\mu_j)^{m_j}$, $(\lambda - i\overline{\mu}_j)^{m_j}$, j = 1: s,

with ℓ_j, m_j such that $\sum_{j=1}^r \ell_j + 2 \sum_{j=1}^s m_j = mn$. With the change of eigenvalue parameter $\lambda = -i\mu$, the *-even linearization of $P(\lambda), \lambda \mathcal{A}_S + \mathcal{B}_S = \mu(-i\mathcal{A}_S) + \mathcal{B}_S$ becomes a Hermitian pencil in μ . Using Section 3.5.1 we obtain that

$$J = -i \left(\bigoplus_{j=1}^{r} J_{\ell_j}(-\beta_j) \oplus \bigoplus_{j=1}^{s} (J_{m_j}(-\bar{\mu}_j) \oplus J_{m_j}(-\mu_j)) \right),$$
$$S_J = -i \left(\bigoplus_{j=1}^{r} \varepsilon_j F_{\ell_j} \oplus \bigoplus_{j=1}^{s} F_{2m_j} \right).$$

Here $\{\varepsilon_1, \ldots, \varepsilon_r\}$ with $\varepsilon_j = \pm 1$ is the sign characteristic associated with the zero and purely imaginary eigenvalues of $P(\lambda)$. Note that $S_J = -S_J^*$ and $JS_J = (JS_J)^*$.

3.5.5 Real *T*-even structure

Suppose $P(\lambda)$ is real *T*-even with (see [66])

- t zero elementary divisors λ^{n_j} with n_j even, j = 1: t,
- r pairs of real elementary divisors $(\lambda + \alpha_j)^{p_j}$, $(\lambda \alpha_j)^{p_j}$ with p_j odd if $\alpha_j = 0$, j = 1: r,

- s pairs of purely imaginary elementary divisors $(\lambda + i\beta_j)^{k_j}$, $(\lambda i\beta_j)^{k_j}$ with $\beta_j > 0, j = 1:s$, and
- q quadruples of nonreal and non-purely imaginary elementary divisors $(\lambda + \mu_j)^{m_j}, (\lambda \mu_j)^{m_j}, (\lambda + \overline{\mu}_j)^{m_j}, (\lambda \overline{\mu}_j)^{m_j}, j = 1; q,$

with n_j, p_j, k_j, m_j such that $\sum_{j=1}^t n_j + 2 \sum_{j=1}^r p_j + 2 \sum_{j=1}^s k_j + 4 \sum_{j=1}^q m_j = mn$. Using [57, Thm. 16.1], we find that

$$J = \bigoplus_{j=1}^{t} J_{n_j}(0) \oplus \bigoplus_{j=1}^{r} \left(J_{p_j}(\alpha_j) \oplus -J_{p_j}(\alpha_j)^T \right)$$
$$\oplus \bigoplus_{j=1}^{s} K_{2k_j}(i\beta_j, -i\beta_j) \oplus \bigoplus_{j=1}^{q} \left(K_{2m_j}(\mu_j, \bar{\mu}_j) \oplus -K_{2m_j}(\mu_j, \bar{\mu}_j)^T \right),$$

$$S_J = \bigoplus_{j=1}^t \varepsilon_j E_{n_j} \oplus \bigoplus_{j=1}^r \begin{bmatrix} 0 & -I_{p_j} \\ I_{p_j} & 0 \end{bmatrix} \oplus \bigoplus_{j=1}^s \varepsilon_j (E_{k_j} \otimes E_2^{k_j}) \oplus \bigoplus_{j=1}^q \begin{bmatrix} 0 & -I_{2m_j} \\ I_{2m_j} & 0 \end{bmatrix},$$

where the scalars $\varepsilon_j = \pm 1$ form the sign characteristic associated with the purely imaginary eigenvalues and zero eigenvalues of even partial multiplicities (see [77]). We easily check that $S_J = -S_J^T$ and $JS_J = (JS_J)^T$.

3.5.6 Complex *T*-even structure

Let $\lambda_j \in \mathbb{C} \setminus \{0\}$ and suppose $P(\lambda)$ is complex *T*-even with (see [66])

- t zero elementary divisors λ^{m_j} with m_j even, j = 1: t, and
- q pairs of elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda + \lambda_j)^{k_j}$ with k_j odd if $\lambda_j = 0$, j = 1; q,

with m_j, k_j such that $\sum_{j=1}^t m_j + 2 \sum_{j=1}^q k_j = mn$. Then, by [83, Prop. 4.7 (b)], we obtain that

$$J = \bigoplus_{j=1}^{t} J_{m_j}(0) \oplus \bigoplus_{j=1}^{q} (J_{k_j}(\lambda_j) \oplus J_{k_j}(-\lambda_j)),$$
$$S_J = \bigoplus_{j=1}^{t} \begin{bmatrix} 0 & -F_{\frac{1}{2}m_j} \\ F_{\frac{1}{2}m_j} & 0 \end{bmatrix} \oplus \bigoplus_{j=1}^{q} \begin{bmatrix} 0 & -F_{k_j} \\ F_{k_j} & 0 \end{bmatrix}$$

Note that $S_J = -S_J^T$ and $JS_J = (JS_J)^T$.

3.5.7 *-odd structure

Suppose $P(\lambda)$ is *-odd with

- r purely imaginary (including 0) elementary divisors $(\lambda i\beta_j)^{\ell_j}$, j = 1: r, and
- s pairs of nonzero and non-purely imaginary elementary divisors $(\lambda i\mu_j)^{m_j}$, $(\lambda - i\overline{\mu}_j)^{m_j}$, j = 1: s,

with ℓ_j, m_j such that $\sum_{j=1}^r \ell_j + 2 \sum_{j=1}^s m_j = mn$. Note that for the *-odd linearization $\lambda \mathcal{A}_S + \mathcal{B}_S$ of $P(\lambda)$ in (3.3), the pencil $i(\lambda \mathcal{A}_S + \mathcal{B}_S)$ is *-even and the structure for S_J and J follows from Section 3.5.4. We find that

$$J = -i \left(\bigoplus_{j=1}^{r} J_{\ell_j}(-\beta_j) \oplus \bigoplus_{j=1}^{s} \left(J_{m_j}(-\bar{\mu}_j) \oplus J_{m_j}(-\mu_j) \right) \right),$$
$$S_J = S_J^{-1} = \bigoplus_{j=1}^{r} \varepsilon_j F_{\ell_j} \oplus \bigoplus_{j=1}^{s} F_{2m_j},$$

which satisfy $S_J = S_J^*$ and $JS_J = -(JS_J)^*$. Here $\{\varepsilon_1, \ldots, \varepsilon_r\}$ with $\varepsilon_j = \pm 1$ is the sign characteristic associated with the zero and purely imaginary eigenvalues of $P(\lambda)$.

3.5.8 Real *T*-odd structure

Suppose $P(\lambda)$ is real *T*-odd with (see [66])

- t zero elementary divisors λ^{ℓ_j} with ℓ_j odd, j = 1: t,
- r pairs of real elementary divisors $(\lambda + \alpha_j)^{p_j}$, $(\lambda \alpha_j)^{p_j}$ with p_j even if $\alpha_j = 0$, j = 1: r,
- s pairs of purely imaginary elementary divisors $(\lambda + i\beta_j)^{k_j}$, $(\lambda i\beta_j)^{k_j}$ with $\beta_j > 0, j = 1: s$, and
- q quadruples of nonreal and non-purely imaginary elementary divisors $(\lambda + \mu_j)^{m_j}$, $(\lambda \mu_j)^{m_j}$, $(\lambda + \overline{\mu}_j)^{m_j}$, $(\lambda \overline{\mu}_j)^{m_j}$ j = 1: q,

with ℓ_j, p_j, k_j, m_j such that $\sum_{j=1}^t \ell_j + 2 \sum_{j=1}^r p_j + 2 \sum_{j=1}^s k_j + 4 \sum_{j=1}^q m_j = mn$. On using [57, Thm. 17.1] we find that

$$J = \bigoplus_{j=1}^{t} J_{\ell_j}(0) \oplus \bigoplus_{j=1}^{r} \left(J_{p_j}(\alpha_j) \oplus -J_{p_j}(\alpha_j)^T \right)$$
$$\oplus \bigoplus_{j=1}^{s} K_{2k_j}(i\beta_j, -i\beta_j) \oplus \bigoplus_{j=1}^{q} \left(K_{2m_j}(\mu_j, \bar{\mu}_j) \oplus -K_{2m_j}(\mu_j, \bar{\mu}_j)^T \right),$$

$$S_J = S_J^{-1} = \bigoplus_{j=1}^t \varepsilon_j E_{\ell_j} \oplus \bigoplus_{j=1}^r \begin{bmatrix} 0 & I_{p_j} \\ I_{p_j} & 0 \end{bmatrix} \oplus \bigoplus_{j=1}^s \varepsilon_j (E_{k_j} \otimes E_2^{k_j-1}) \oplus \bigoplus_{j=1}^q \begin{bmatrix} 0 & I_{2m_j} \\ I_{2m_j} & 0 \end{bmatrix},$$

where the scalars $\varepsilon_j = \pm 1$ form the sign characteristic associated with the purely imaginary eigenvalues and the zero eigenvalues with odd partial multiplicities. We easily check that $S_J = S_J^T$ and $JS_J = -(JS_J)^T$.

3.5.9 Complex *T*-odd structure

Let $\lambda_j \in \mathbb{C} \setminus \{0\}$ and suppose $P(\lambda)$ is complex *T*-odd with (see [66])

- s zero elementary divisors λ^{ℓ_j} with ℓ_j odd, j = 1: s, and
- q pairs of elementary divisors $(\lambda + \lambda_j)^{k_j}$, $(\lambda \lambda_j)^{k_j}$ with k_j even if $\lambda_j = 0$, j = 1: q,

with ℓ_j, k_j such that $\sum_{j=1}^s \ell_j + 2 \sum_{j=1}^q k_j = mn$. It follows from [83, Prop. 4.7 (b)] that

$$J = \bigoplus_{j=1}^{s} J_{\ell_j}(0) \oplus \bigoplus_{j=1}^{q} \left(-J_{k_j}(\lambda_j) \oplus J_{k_j}(\lambda_j) \right), \qquad S_J = S_J^{-1} = \bigoplus_{j=1}^{s} E_{\ell_j} \oplus \bigoplus_{j=1}^{q} F_{2k_j}.$$

Clearly, $S_J = S_J^T$ and $JS_J = -(JS_J)^T$.

Notice the difference between the zero elementary divisors associated with T-even and T-odd pencils (see [66, Cor. 4.3]).

3.5.10 *-(anti)palindromic structure

Suppose $P(\lambda)$ is complex *-palindromic with $-1 \notin \sigma(P)$ and (see [68])

• q pairs of elementary divisors $(\lambda - \lambda_j)^{k_j}$, $(\lambda - 1/\overline{\lambda}_j)^{k_j}$ with $\lambda_j \in \mathbb{C} \setminus \{0\}$, $|\lambda_j| \neq 1$, j = 1: q,

- t elementary divisors $(\lambda \lambda_j)^{2\ell_j + 1}$ with $\lambda_j \in \mathbb{C}$ such that $|\lambda_j| = 1, j = 1: t$, and
- s elementary divisors $(\lambda \lambda_j)^{2m_j}$ with $\lambda_j \in \mathbb{C}, |\lambda_j| = 1, j = 1: s$,

with k_j, ℓ_j, m_j such that $2 \sum_{j=1}^q k_j + \sum_{j=1}^t (2\ell_j + 1) + 2 \sum_{j=1}^s m_j = mn$. Then using either [86, Thm. 5] or [87, Section 2.2.2] we find that

$$J = -S_J S_J^{-*}$$

with

$$S_{J} = \bigoplus_{j=1}^{q} \begin{bmatrix} 0_{k_{j}} & F_{k_{j}}J_{k_{j}}(-\lambda_{j}) \\ F_{k_{j}} & 0_{k_{j}} \end{bmatrix} \oplus \bigoplus_{j=1}^{t} \varepsilon_{j} \begin{bmatrix} 0 & 0 & F_{\ell_{j}}J_{\ell_{j}}(-\lambda_{j}) \\ 0 & (-\lambda_{j})^{1/2} & e_{1}^{T} \\ F_{\ell_{j}} & 0 & 0 \end{bmatrix}$$
$$\oplus \bigoplus_{j=1}^{s} \varepsilon_{j} \begin{bmatrix} 0_{m_{j}} & F_{m_{j}}J_{m_{j}}(-\lambda_{j}) \\ F_{m_{j}} & e_{1}e_{1}^{T} \end{bmatrix}$$

has the above elementary divisors. The scalars $\varepsilon_j = \pm 1$ form the sign characteristic associated with the eigenvalues of unit modulus of $P(\lambda)$ (see [55]).

For the *-antipalindromic structure, $J = S_J S_J^{-*}$ with S_J as above but with $-\lambda_j$ replaced by λ_j .

3.5.11 Real *T*-(anti)palindromic structure

Suppose $P(\lambda)$ is real *T*-palindromic with $-1 \notin \sigma(P), \lambda_j \in \mathbb{C} \setminus \{0\}$, and (see [68])

- r pairs of real elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda 1/\lambda_j)^{k_j}$ with $\lambda_j \in \mathbb{R}$, $|\lambda_j| \neq 1$, j = 1: r,
- q quadruples of nonreal elementary divisors $(\lambda \lambda_j)^{n_j}$, $(\lambda \overline{\lambda}_j)^{n_j}$, $(\lambda 1/\lambda_j)^{n_j}$, $(\lambda -$
- s elementary divisors $(\lambda 1)^{2m_j}$, j = 1: s,
- t pairs of elementary divisors $(\lambda 1)^{2\ell_j + 1}$, $(\lambda 1)^{2\ell_j + 1}$ j = 1:t,
- *u* pairs of elementary divisors $(\lambda \lambda_j)^{\ell'_j}$, $(\lambda \overline{\lambda}_j)^{\ell'_j}$ with $|\lambda_j| = 1$, $\lambda_j \neq 1$, ℓ'_j odd, j = 1: u, and

• p pairs of elementary divisors $(\lambda - \lambda_j)^{m'_j}$, $(\lambda - \overline{\lambda}_j)^{m'_j}$ with $|\lambda_j| = 1$, $\lambda_j \neq 1$, m'_j even, j = 1: p.

We have that $2\sum_{j=1}^{r} k_j + 4\sum_{j=1}^{q} n_j + 2\sum_{j=1}^{s} m_j + 2\sum_{j=1}^{t} (2\ell_j + 1) + 2\sum_{j=1}^{u} \ell'_j + 2\sum_{j=1}^{p} m'_j = mn.$

On using [87, Thm. 2.8] we find that $J = -S_J S_J^{-T}$ has the above list of elementary divisors, where

$$\begin{split} S_{J} &= \bigoplus_{j=1}^{r} \begin{bmatrix} 0_{k_{j}} & F_{k_{j}}J_{k_{j}}(-\lambda_{j}) \\ F_{k_{j}} & 0_{k_{j}} \end{bmatrix} \oplus \bigoplus_{j=1}^{q} \begin{bmatrix} 0_{2n_{j}} & K_{2n_{j}}(-\Lambda_{j}) \\ F_{n_{j}} \otimes I_{2} & 0_{2n_{j}} \end{bmatrix} \\ &\oplus \bigoplus_{j=1}^{s} \begin{bmatrix} 0 & F_{m_{j}}J_{m_{j}}(-1) \\ F_{m_{j}} & 0 \end{bmatrix} \\ &\oplus \bigoplus_{j=1}^{t} \varepsilon_{j} \begin{bmatrix} 0_{\ell_{j}} & 0 & F_{\ell_{j}}J_{\ell_{j}}(-1) \\ 0 & 1 & e_{1}^{T} \\ F_{\ell_{j}} & 0 & 0_{\ell_{j}} \end{bmatrix} \oplus \bigoplus_{j=1}^{t} \varepsilon_{j} \begin{bmatrix} 0_{\ell_{j}} & 0 & F_{\ell_{j}}J_{\ell_{j}}(-1) \\ 0 & 1 & e_{1}^{T} \\ F_{\ell_{j}} & 0 & 0_{\ell_{j}} \end{bmatrix} \\ &\oplus \bigoplus_{j=1}^{u} \varepsilon_{j} \begin{bmatrix} 0_{\ell_{j}'-1} & 0 & K_{\ell_{j}'-1}(-\Lambda_{j}) \\ 0 & (-\Lambda_{j})^{\frac{1}{2}} & e_{1}^{T} \otimes I_{2} \\ F_{(\ell_{j}'-1)/2} \otimes I_{2} & 0 & 0_{\ell_{j}'-1} \end{bmatrix} \oplus \bigoplus_{j=1}^{p} \varepsilon_{j} \begin{bmatrix} 0_{m_{j}'} & K_{m_{j}'}(-\Lambda_{j}) \\ F_{\frac{1}{2}m_{j}'} \otimes I_{2} & e_{1}e_{1}^{T} \otimes I_{2} \\ F_{\frac{1}{2}m_{j}'} \otimes I_{2} & e_{1}e_{1}^{T} \otimes I_{2} \end{bmatrix} \end{split}$$

Here $(-\Lambda_j)^{\frac{1}{2}}$ is the principal square root of $-\Lambda_j$. The scalars ε_j are signs ± 1 and form the sign characteristic associated with the eigenvalues of unit modulus of $P(\lambda)$ except the eigenvalues 1 with even partial multiplicities (see [55]).

For the *T*-antipalindromic $P(\lambda)$, $J = S_J S_J^{-T}$ where S_J is as above but with $-\lambda_j, -1, -\Lambda_j$ replaced by $\lambda_j, 1, \Lambda_j$, respectively.

3.5.12 Complex *T*-(anti)palindromic structure

Suppose $P(\lambda)$ is complex T-palindromic with $-1 \notin \sigma(P)$ and (see [68])

- t elementary divisors $(\lambda 1)^{m_j}$ with m_j even, j = 1: t,
- q pairs of elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda 1/\lambda_j)^{k_j}$ with k_j odd when $\lambda_j = 1$, $\lambda_j \neq 0, \ j = 1; q$,

with m_j, k_j such that $\sum_{j=1}^t m_j + 2 \sum_{j=1}^q k_j = mn$. On using either [86, Thm. 1] or [87, Thm. 2.6], we find that with

$$S_{J} = \bigoplus_{j=1}^{t} \begin{bmatrix} 0_{m_{j}/2} & F_{m_{j}/2}J_{m_{j}/2}(-1) \\ F_{m_{j}/2} & e_{1}e_{1}^{T} \end{bmatrix} \oplus \bigoplus_{j=1}^{q} \begin{bmatrix} 0_{k_{j}} & F_{k_{j}}J_{k_{j}}(-\lambda_{j}) \\ F_{k_{j}} & 0_{k_{j}} \end{bmatrix}$$

the matrix $J = -S_J S_J^{-T}$ has the above elementary divisors.

Now if $P(\lambda)$ is complex *T*-antipalindromic with $-1 \notin \sigma(P)$ and (see [68])

- t elementary divisors $(\lambda 1)^{\ell_j}$ with ℓ_j odd, j = 1: t,
- q pairs of elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda 1/\lambda_j)^{k_j}$ with k_j even if $\lambda_j = 1$, j = 1: q,

with ℓ_j, k_j such that $\sum_{j=1}^t \ell_j + 2 \sum_{j=1}^q k_j = mn$. On using [87, Thm. 2.6], we find that the matrix $J = S_J S_J^{-T}$ with

$$S_{J} = \bigoplus_{j=1}^{t} \begin{bmatrix} 0_{\ell_{j}} & 0 & F_{\ell_{j}} J_{\ell_{j}}(1) \\ 0 & 1 & e_{1}^{T} \\ F_{\ell_{j}} & 0 & 0_{\ell_{j}} \end{bmatrix} \oplus \bigoplus_{j=1}^{q} \begin{bmatrix} 0_{k_{j}} & F_{k_{j}} J_{k_{j}}(\lambda_{j}) \\ F_{k_{j}} & 0_{k_{j}} \end{bmatrix}$$

has the above elementary divisors.

Note that J in Section 3.5.10—Section 3.5.12 is "almost" in Jordan canonical form.

3.6 Applications

The theory developed in this chapter simplifies the structured inverse quadratic eigenvalue problem having the form that given a structure S and 2n eigenvalues with their algebraic, geometric and partial multiplicities plus the attached sign characteristic when necessary, construct the corresponding $Q(\lambda) \in \mathcal{P}_{S}(\mathbb{F}^{n})$. We illustrate with some examples now and offer a deeper discussion in the next chapter.

Example 3.6.1 (complex symmetric) If $J = J_2(1+i) \oplus J_2(2-i)$ is a Jordan matrix containing the eigenvalues of a complex symmetric quadratic matrix polynomial, then we can use Section 3.5.3 to construct $S_J = F_2 \oplus F_2$. A solution X for

$$XS_J X^T = 0 \text{ is}$$

$$X = \begin{bmatrix} 1 & 0 & 1 & 0 \\ 0 & 1 & 0 & -1 \end{bmatrix}.$$
Thus
$$A_2 = (XJS_J X^T)^{-1} = \begin{bmatrix} 0 & -0.2 - 0.4i \\ -0.2 - 0.4i & 0.24 - 0.32i \end{bmatrix}$$

$$A_1 = -A_2 XJ^2 S_J X^T A_2 = \begin{bmatrix} 0 & 0.6 + 1.2i \\ 0.6 + 1.2i & -0.72 + 0.96i \end{bmatrix}$$

and

$$A_0 = -A_2(XJ^2S_JX^TA_1 + XJ^3S_JX^TA_2) = \begin{bmatrix} 0 & -0.2 - 1.4i \\ -0.2 - 1.4i & 0.04 - 0.72i \end{bmatrix},$$

define a complex symmetric $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ with the prescribed spectrum.

Example 3.6.2 (real *T***-even)** If $J = J_1(1) \oplus -J_1(1)^T \oplus K_2(1+i, 1-i) \oplus -K_2(1+i, 1-i)^T \oplus K_2(i, -i)$ is a Jordan matrix containing the eigenvalues of a real *T*-even quadratic matrix polynomial where ε_1 associated with i, -i equals 1. Then we use Section 3.5.5 to construct

$$S_{J} = \begin{bmatrix} 0 & -1 \\ 1 & 0 \end{bmatrix} \oplus \begin{bmatrix} 0 & 0 & -1 & 0 \\ 0 & 0 & 0 & -1 \\ 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \end{bmatrix} \oplus \begin{bmatrix} 0 & 1 \\ -1 & 0 \end{bmatrix}.$$

A solution for $XS_JX^T = 0$ is

$$X = \begin{bmatrix} 0 & 1 & 1 & 0 & 0 & 0 & 0 & 0 \\ 1 & 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 & 0 & -1 \end{bmatrix}.$$

Thus

$$A_{2} = (XJS_{J}X^{T})^{-1} = \begin{bmatrix} -0.2 & -0.2 & -0.4 & -0.6 \\ -0.2 & -0.2 & 0.6 & 0.4 \\ -0.4 & 0.6 & -0.8 & -1.2 \\ -0.6 & 0.4 & -1.2 & -0.8 \end{bmatrix},$$
$$A_{1} = -A_{2}XJ^{2}S_{J}X^{T}A_{2} = \begin{bmatrix} 0 & -0.6 & 1 & 0.2 \\ 0.6 & 0 & 0.2 & 1 \\ -1 & -0.2 & 0 & -2.6 \\ -0.2 & -1 & 2.6 & 0 \end{bmatrix},$$

and

$$A_0 = -A_2(XJ^2S_JX^TA_1 + XJ^3S_JX^TA_2) = \begin{bmatrix} 0.2 & 0.8 & -0.6 & 0.4 \\ 0.8 & 0.2 & -0.4 & 0.6 \\ -0.6 & -0.4 & 1.8 & -1.2 \\ 0.4 & 0.6 & -1.2 & 1.8 \end{bmatrix},$$

define a real T-even $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ with the prescribed spectrum.

Example 3.6.3 (real or complex *-palindromic) If 2, 1/2, i, -i are the simple eigenvalues of either a real or a complex quadratic *-palindromic matrix polynomial where $\varepsilon_{1,2} = 1$. Then we can use Sections 3.5.10, 3.5.12 and 3.5.11 to construct complex $Q_1(\lambda), Q_2(\lambda)$ and a real $Q_3(\lambda)$ with the prescribed eigenvalues and the desirable structure. For example,

$$Q_{1}(\lambda) = \lambda^{2}A_{2} + \lambda A_{1} + A_{2}^{*} := \\ \lambda^{2} \begin{bmatrix} 0.2879 & -0.1111 - 0.1768i \\ -0.1111 + 0.1768i & 0.2879 \end{bmatrix} \\ + \lambda \begin{bmatrix} -0.2778 & 0.2778 \\ 0.2778 & -0.2778 \end{bmatrix} + A_{2}^{*},$$

$$Q_{2}(\lambda) = \lambda^{2}A_{2} + \lambda A_{1} + A_{2}^{T} := \lambda^{2} \begin{bmatrix} 0 & 0.1333 - 0.2667i \\ -0.1333 - 0.2667i & 0 \end{bmatrix}$$

+
$$\lambda \begin{bmatrix} 0 & 0.2000 + 0.2667i \\ 0.2000 + 0.2667i & 0 \end{bmatrix} + A_2^T$$
,

and

$$Q_3(\lambda) = \lambda^2 \begin{bmatrix} 0.3536 & 0.1667 \\ -0.6667 & 0.3143 \end{bmatrix} + \lambda \begin{bmatrix} -0.8839 & 0 \\ 0 & -0.7857 \end{bmatrix} + \begin{bmatrix} 0.3536 & 0.1667 \\ -0.6667 & 0.3143 \end{bmatrix}^T$$

3.7 Concluding remarks

The results in this chapter represent a first step towards the solution of the SIPEP: given a list of admissible elementary divisors for the structure, and possibly, corresponding right eigenvectors and generalized eigenvectors, construct a structured matrix polynomial having these elementary divisors and eigenvectors/generalized eigenvectors. Indeed, using the results in Sections 3.3 and 3.4 we show in Chapter 4 how to construct an S-structured (2, n)-Jordan triple (X, J, Y) from a given list of 2n prescribed eigenvalues and n linearly independent eigenvectors and generalized eigenvectors, and use the fact that an S-structured (2, n)-Jordan triple defines a unique structured quadratic $Q(\lambda) \in \mathcal{P}_{S}(\mathbb{F}^{n})$.

Also, in the next chapter, we provide a detailed treatment of the associated quadratic and cubic quasidefinite IPEP using self-adjoint Jordan triples.

Finally, we note that standard triples have been useful to describe structure preserving transformations (SPTs) for matrix polynomials, and in particular quadratic matrix polynomials [28]. We believe that the notion of S-structured standard triples will further our understanding of SPTs for structured matrix polynomials.

Chapter 4

Structured Inverse Polynomial Eigenvalue Problems

4.1 Introduction

The IPEP is to determine the coefficient matrices of $P(\lambda)$ from its prescribed spectral data (i.e., eigenvalues and eigenvectors). Standard, generalized and quadratic inverse eigenvalue problems have been widely investigated, see [15], [16], [17], [20], [27] and the references therein. For example, suppose that a $2n \times 2n$ Jordan matrix J is given together with an $n \times 2n$ matrix X whose columns contain the corresponding eigenvectors and generalized eigenvectors. Then if $\begin{bmatrix} XJ\\ X \end{bmatrix}$ is nonsingular, the $n \times n$ quadratic $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ defined by

$$A_2 = I_n, \quad \begin{bmatrix} A_1 & A_0 \end{bmatrix} = -XJ^2 \begin{bmatrix} XJ \\ X \end{bmatrix}^{-1}$$
(4.1)

has the prescribed Jordan form J and eigenvector matrix X [31, p. 3].

In applications, the coefficients of $P(\lambda)$ often have a specific structure such as for example symmetry or skew-symmetry, which corresponds to the physics of the problem. This leads to SIPEPs. Note that apart from A_2 , the other coefficient matrices in (4.1) have no particular structure in general. The inverse problem for Hermitian and symmetric quadratics has been the topic of many papers such as [18], [48], [49] to cite just a few.

The interest in this chapter is in constructing, from spectral data, complex and

real quadratics with structures S listed in Table 1.1 as well as generating quadratic and cubic quasidefinite matrix polynomials displayed in Figure 2.1.

Our solution process makes use of the recent concept of S-structured Jordan triples introduced in Chapter 3. The structure of $P(\lambda)$ imposes certain constraints on the Jordan form of S-structured Jordan triples. For example, the spectrum of Hermitian matrix polynomials is symmetric with respect to the real line. So for each $S \in S$, we carefully describe in Section 4.3 the structure of the Jordan matrix, or equivalently, the list of elementary divisors admissible for the structure S of a quadratic $Q(\lambda) \in \mathcal{P}_S(\mathbb{F}^n)$. Given such a list, we show how to construct families of Sstructured Jordan triples that generates structured quadratics having the given list of elementary divisors. Our construction shows that a solution to the quadratic inverse problem also exists when half of the eigenvectors and generalized eigenvectors are also provided. Since the solution process we provide allows freedom on how to choose the sign characteristic, which is an invariant for certain structured λ -matrices, an appropriate choice for the sign characteristic will facilitate generating quasihyperbolic quadratics and cubics.

We also show how to use Theorem 2.3.10 along with [81, Alg. 2.1] to generate tridiagonal hyperbolic quadratics.

Some of our constructions form the basis of a MATLAB function, gen_hyper2, in the NLEVP collection [13].

4.2 Preliminaries

Throughout this chapter, $J \in \mathbb{F}^{mn \times mn}$ satisfies Assumption (b) of Section 3.3.

Recall that an $\ell \times \ell$ Jordan matrix over \mathbb{C} is a block diagonal matrix displaying the algebraic, geometric and partial multiplicities of the eigenvalues, that is,

$$J = \bigoplus_{i=1}^{s} \bigoplus_{j=1}^{m_i} J_{\ell_{ij}}(\lambda_i) \in \mathbb{C}^{\ell \times \ell}, \quad \lambda_i \neq \lambda_k \text{ if } i \neq k,$$

$$(4.2)$$

where

$$J_{\ell_{ij}}(\lambda_i) := \begin{bmatrix} \lambda_i & 1 & & \\ & \lambda_i & \ddots & \\ & & \ddots & 1 \\ & & & \lambda_i \end{bmatrix} \in \mathbb{C}^{\ell_{ij} \times \ell_{ij}}$$

is a Jordan block of size ℓ_{ij} associated with λ_i . To each $J_{\ell_{ij}}(\lambda_i)$ corresponds an elementary divisor $(\lambda - \lambda_i)^{\ell_{ij}}$ of $\lambda I - J$, or respectively, $P(\lambda)$. The integer ℓ_{ij} is a partial multiplicity of λ_i and λ_i can have several partial multiplicities. The integer m_i , which is the number of partial multiplicities associated with λ_i is called the geometric multiplicity of λ_i . The algebraic multiplicity of λ_i is the sum of its partial multiplicities. Note that $\sum_{i=1}^{s} \sum_{j=1}^{m_i} \ell_{ij} = \ell$.

An $\ell \times \ell$ Jordan matrix J over $\mathbb R$ is a block diagonal matrix

$$J = \bigoplus_{i=1}^{r} \bigoplus_{j=1}^{m_i} J_{\ell_{ij}}(\lambda_i) \bigoplus_{i=1}^{t} \bigoplus_{j=1}^{m'_i} K_{2\ell'_{ij}}(\mu_i, \bar{\mu}_i) \in \mathbb{C}^{\ell \times \ell}, \quad \lambda_i \neq \lambda_k, \ \mu_i \neq \mu_k \text{ if } i \neq k, \quad (4.3)$$

where $\lambda_i \in \mathbb{R}$, i = 1: r and $\mu_i \in \mathbb{C}$ with $\text{Im}(\mu_i) > 0$, i = 1: t, and

$$K_{2\ell'_{ij}}(\mu_i,\bar{\mu}_i) = \begin{bmatrix} \Lambda_i & I_2 & & \\ & \Lambda_i & \ddots & \\ & & \ddots & I_2 \\ & & & & \Lambda_i \end{bmatrix} \in \mathbb{R}^{2\ell'_{ij} \times 2\ell'_{ij}}, \quad \Lambda_i = \begin{bmatrix} \operatorname{Re}(\mu_i) & \operatorname{Im}(\mu_i) \\ -\operatorname{Im}(\mu_i) & \operatorname{Re}(\mu_i) \end{bmatrix} \in \mathbb{R}^{2 \times 2},$$

is a real Jordan block associated with $(\mu_i, \bar{\mu}_i)$. To each $K_{2\ell_{ij}}(\mu_i, \bar{\mu}_i)$ corresponds a pair of elementary divisors $(\lambda - \mu_i)^{\ell'_{ij}}$, $(\lambda - \bar{\mu}_i)^{\ell'_{ij}}$ of $\lambda I - J$. Note that $\sum_{i=1}^r \sum_{j=1}^{m_i} \ell_{ij} + 2\sum_{i=1}^t \sum_{j=1}^{m'_i} \ell'_{ij} = \ell$.

A quadratic $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0 \in \mathcal{P}(\mathbb{F}^n)$ with A_2 nonsingular has 2n finite eigenvalues so its Jordan form J over \mathbb{F} is $2n \times 2n$. Note that the geometric multiplicity of λ_i is the number of linearly independent eigenvectors of $Q(\lambda)$ corresponding to λ_i and since $Q(\lambda)$ is $n \times n$, we have that m_i and m'_i in (4.2) and (4.3) cannot be larger than n. Recall that, as a direct consequence of Theorem 3.3.5, if (X, J, Y) is an \mathcal{S} -structured Jordan triple then the matrices in (3.13)–(3.14) with m = 2 define a unique structured quadratic.

If $Q(\lambda)$ is a quadratic matrix polynomial with structure $S \in S$ and S-structured Jordan triple $(X, J, Sv_{\mathcal{S}}(J)X^{\star})$ then the sign characteristic is defined as the sign characteristic of the pair (J, S^{-1}) (see Section 3.4). This is a list of signs, with a sign (+1 or -1). These signs can be read off the block diagonal entries of S. They form an invariant for $Q(\lambda)$ and they play an essential role in the construction of Hermitian and symmetric quadratics with special properties (see Chapter 2 and Section 4.3.1).

4.3 Building S-structured Jordan triples

As explained in the introduction, our main objective is to construct structured quadratics, both real and complex, from given spectral data. To be more specific, we solve the following problem:

(P) Given a structure $S \in \mathbb{S}$ and a list of elementary divisors

$$\mathcal{L} = \{ (\lambda - \lambda_i)^{\ell_{ij}}, \ i = 1: s, \ j = 1: m_i \le n, \ \lambda_i \ne \lambda_k \ \text{if } i \ne k, \ \sum_{i=1}^s \sum_{j=1}^{m_i} \ell_{ij} = 2n \}$$
(4.4)

admissible for the structure, construct an $n \times n$ quadratic $Q(\lambda)$ with structure S having the prescribed spectral data.

One way to solve problem (**P**) is to construct an S-structured Jordan triple $(X, J, S_J v_S(J)X^*)$, where J displays the given list \mathcal{L} of elementary divisors. To do so, we use the explicit expressions for J and S_J provided for each $S \in S$ in Section 3.5. Note that the structure of J imposes constraints on the structure of the list \mathcal{L} . The main difficulty lies in constructing $X \in \mathbb{F}^{n \times 2n}$ such that

$$XS_J v_{\mathcal{S}}(J) X^{\star} = 0 \tag{4.5}$$

and

$$\det \begin{bmatrix} XJ\\ X \end{bmatrix} \neq 0, \quad \det(XJS_Jv_{\mathcal{S}}(J)X^{\star}) \neq 0.$$
(4.6)

Note that the latter constraints are more likely to hold than the former constraint. We consider each structure separately.

4.3.1 Hermitian and real symmetric quadratics

It follows from [71] that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r real elementary divisors $(\lambda \lambda_j)^{\ell_j}$, j = 1: r, and
- t pairs of nonreal conjugate elementary divisors $(\lambda \mu_j)^{m_j}$, $(\lambda \overline{\mu}_j)^{m_j}$, j = 1: t,

with ℓ_j, m_j such that $\sum_{j=1}^r \ell_j + 2 \sum_{j=1}^t m_j = 2n$. The corresponding complex and real Jordan matrices are given by Section 3.5.1, Section 3.5.2

$$J = \bigoplus_{j=1}^r J_{\ell_j}(\lambda_j) \oplus \bigoplus_{j=1}^t (J_{m_j}(\bar{\mu}_j) \oplus J_{m_j}(\mu_j)), \quad J = \bigoplus_{j=1}^r J_{\ell_j}(\lambda_j) \oplus \bigoplus_{j=1}^t K_{2m_j}(\mu_j, \bar{\mu}_j),$$

respectively, and

$$S_J = S_J^{-1} = \bigoplus_{j=1}^r \varepsilon_j F_{\ell_j} \oplus \bigoplus_{j=1}^t F_{2m_j}$$

where the $\varepsilon_j = \pm 1$, j = 1:r define the sign characteristic associated with the real eigenvalues λ_j and must be such that [31, Prop. 10.12]

$$\operatorname{sig}(S_J) = \sum_{j=1}^r \frac{1}{2} (1 - (-1)^{\ell_j}) \varepsilon_j = 0, \qquad (4.7)$$

where $sig(S_J)$ is the signature of S_J , that is, the difference between the number of positive eigenvalues and the number of negative eigenvalues of S_J . We easily check that $S_J = S_J^*$ and $JS_J = (JS_J)^*$.

Then on using (4.7), we see that there exists an orthogonal matrix $W \in \mathbb{F}^{2n \times 2n}$ such that $W^T S_J W = \begin{bmatrix} I_n & 0 \\ 0 & -I_n \end{bmatrix}$. If we let $\widetilde{X} = XW = \begin{bmatrix} \widetilde{X}_1 & \widetilde{X}_2 \end{bmatrix}$ with $\widetilde{X}_1 \in \mathbb{F}^{n \times n}$ of full rank and write $\widetilde{X}_2 = \widetilde{X}_1 \Theta$ for some unitary $\Theta \in \mathbb{F}^{n \times n}$ then

$$XS_J v_{\mathcal{S}}(J) X^{\star} = \widetilde{X}_1 \widetilde{X}_1^{\star} - \widetilde{X}_2 \widetilde{X}_2^{\star} = 0,$$

Hence we can build X as

$$X = \widetilde{X}W^T = \begin{bmatrix} \widetilde{X}_1 & \widetilde{X}_1\Theta \end{bmatrix} W^T, \tag{4.8}$$

which is completely determined by $\widetilde{X}_1, \Theta \in \mathbb{F}^{n \times n}$ with Θ unitary. Note that the conditions in (4.6) hold for almost all \widetilde{X}_1 and Θ .

In the case where some of the 2n given eigenvalues are infinite, we start by applying a homogeneous rotation $G := \begin{bmatrix} c & -s \\ s & c \end{bmatrix}$ to the given spectrum as in (2.5) so that the rotated eigenvalues are all finite (this can be done by setting c and s so that c/s is not one of the given eigenvalues). Then we build a (2, n)-self-adjoint Jordan triple and the unique Hermitian/symmetric quadratic $\widetilde{Q}(\widetilde{\lambda}) = \widetilde{\lambda}^2 \widetilde{A}_2 + \widetilde{\lambda} \widetilde{A}_1 + \widetilde{A}_0$ that it generates. Next, we homogeneously rotate back $\widetilde{Q}(\widetilde{\lambda})$ into $Q(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$, where $A_2 = \widetilde{Q}(c, s)$ is singular, $A_1 = -2cs\widetilde{A}_2 + (c^2 - s^2)\widetilde{A}_1 + 2cs\widetilde{A}_0$ and $A_0 = \widetilde{Q}(-s, c)$.

Example 4.3.1 Consider the symmetric structure and $\mathcal{L} = \{(\lambda - 2)^3, (\lambda - 2), (\lambda - i), (\lambda + i)\}$ with sign characteristic $\{+1, -1\}$, i.e., a '+' sign associated with $(\lambda - 2)^3$ and a '-' sign associated with $(\lambda - 2)$, so that (4.7) holds with $(\ell_1, \varepsilon_1) = (3, 1)$ and $(\ell_2, \varepsilon_2) = (1, -1)$. Then

$$J = \begin{bmatrix} 2 & 1 \\ 2 & 1 \\ & 2 \end{bmatrix} \oplus \begin{bmatrix} 2 \end{bmatrix} \oplus \begin{bmatrix} 0 & 1 \\ -1 & 0 \end{bmatrix}, \quad S_J = \begin{bmatrix} 1 & 1 \\ 1 & 1 \\ 1 & -1 \end{bmatrix} \oplus (-1) \begin{bmatrix} 1 \end{bmatrix} \oplus \begin{bmatrix} 0 & 1 \\ 1 & 0 \end{bmatrix}$$

With the choice $\widetilde{X}_1 = \Theta = I_3$, we obtain

$$X = \begin{bmatrix} \sqrt{2} & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 1/\sqrt{2} & 1/\sqrt{2} \\ 0 & -1 & 0 & 0 & -1/\sqrt{2} & 1/\sqrt{2} \end{bmatrix}.$$

We easily check that $\begin{bmatrix} XJ \\ X \end{bmatrix}$ is nonsingular. The second and third columns of X are generalized eigenvectors and for matrix polynomials, they can be equal to the zero vector. Then (3.13)–(3.14) with $Y = S_J X^T$ and m = 2 yield

$$A_{2} = \frac{1}{4} \begin{bmatrix} -5 & \sqrt{2} & -2\sqrt{2} \\ \sqrt{2} & -2 & 0 \\ -2\sqrt{2} & 0 & 0 \end{bmatrix}, \quad A_{1} = \begin{bmatrix} 5 & -\sqrt{2} & \frac{3}{2\sqrt{2}} \\ -\sqrt{2} & 1 & \frac{1}{2} \\ \frac{3}{2\sqrt{2}} & \frac{1}{2} & -1 \end{bmatrix}, \quad A_{0} = \begin{bmatrix} -5 & \sqrt{2} & \frac{1}{\sqrt{2}} \\ \sqrt{2} & 0 & -1 \\ \frac{1}{\sqrt{2}} & -1 & -\frac{1}{2} \end{bmatrix}$$

Note that our construction of Hermitian and real symmetric quadratics extends the approach in [18], [48] which solves Problem (**P**) for real symmetric quadratics under the assumption that \mathcal{L} consists entirely of linear elementary divisors, i.e., $\ell_{ij} =$ 1 for all $i = 1: s, j = 1: m_i$ (or equivalently the eigenvalues are all semisimple). Lancaster in [48] writes

$$X = \begin{bmatrix} X_{r_1} & X_{r_2} & X_c & \overline{X}_c \end{bmatrix},$$

where X_c is an $n \times (n-r)$ matrix corresponding to complex eigenvalues in the upper half plane, $X_{r_1}(X_{r_2})$ is a real $n \times r$ matrix corresponding to the real eigenvalues for which $\varepsilon_j = 1$ ($\varepsilon_j = -1$), and 0 < r < n. Assuming X_{r_1} and X_{r_2} are given together with \mathcal{L} ,

$$XS_J X^* = 0 \Leftrightarrow X_c X_c^T + \overline{X_c X_c^T} = -X_{r_1} X_{r_1}^T + X_{r_2} X_{r_2}^T.$$

Thus, $X_c X_c^T = R_1 - iR$ where $R_1 = \frac{1}{2}(-X_{r_1}X_{r_1}^T + X_{r_2}X_{r_2}^T)$ and R is a real symmetric matrix which is chosen so that $R_1 - iR$ has rank n - r. Takagi's factorization is then the tool to recover X_c , see [48, App. A] or [42, Section 4.4.4].

Generating quasidefinite quadratics

Recall that an eigenvalue λ_0 is of definite type if its partial multiplicities are all equal to 1 and the sign (in the sign characteristic) attached to each partial multiplicity is the same (either positive or negative) and a definite type eigenvalue with positive(negative) sign is said to be of positive(negative) type. Note that the eigenvalue 2 in Example 4.3.1 is not of definite type.

The task of constructing an appropriate self-adjoint Jordan triple defined by the three matrices X, J and S_J to produce quasidefinite quadratics follows directly from Section 4.3.1. So for a quasihyperbolic quadratic,

$$J = \operatorname{diag}(\lambda_1, \ldots, \lambda_{2n}), \quad S_J = \operatorname{diag}(\varepsilon_1, \ldots, \varepsilon_{2n}),$$

with $\varepsilon_j = \pm 1$ such that $\varepsilon_j = \varepsilon_k$ if $\lambda_j = \lambda_k$ and $\sum_{j=1}^{2n} \varepsilon_j = 0$ (see (4.7)). The matrix X such that $(X, J, S_J X^*)$ forms an \mathcal{S} -structured Jordan triple can be constructed as in (4.8).

Example 4.3.2 Let J = diag(-1, -2, -3, -4) and let $\widetilde{X}_1 = \Theta = I_2$ in (4.8).

If $S_J = \text{diag}(1, -1, 1, -1)$ then $W = \begin{bmatrix} e_1 & e_3 & e_2 & e_4 \end{bmatrix}$, where e_j is the *j*th column of I_4 , $X = \begin{bmatrix} 1 & 1 & 0 & 0 \\ 0 & 0 & 1 & 1 \end{bmatrix}$ and on using (3.13)–(3.14) with m = 2 and $Y = S_J X^T$, $\begin{bmatrix} 3 & 0 \end{bmatrix} \begin{bmatrix} 2 & 0 \end{bmatrix}$

$$A_2 = I_2, \quad A_1 = \begin{bmatrix} 3 & 0 \\ 0 & 7 \end{bmatrix}, \quad A_0 = \begin{bmatrix} 2 & 0 \\ 0 & 12 \end{bmatrix},$$
 (4.9)

define a quasihyperbolic quadratic.

Applying the homogeneous rotation $G = -\sqrt{2}/2 \begin{bmatrix} 1 & -1 \\ 1 & 1 \end{bmatrix}$ on J provides the quasidefinite $\widetilde{Q}(\widetilde{\lambda}) = \widetilde{\lambda}^2 \widetilde{A}_2 + \widetilde{\lambda} \widetilde{A}_1 + \widetilde{A}_0$ with $\sigma(\widetilde{Q}) = \{5/3, 2, 3, \infty\}$ where

$$\widetilde{A}_2 = \frac{1}{2}(A_2 - A_1 + A_0) = \begin{bmatrix} 0 & 0 \\ 0 & 3 \end{bmatrix},$$

$$\widetilde{A}_1 = A_2 - A_0 = \begin{bmatrix} -1 & 0 \\ 0 & -11 \end{bmatrix}, \quad \widetilde{A}_0 = \frac{1}{2}(A_2 + A_1 + A_0) = \begin{bmatrix} 3 & 0 \\ 0 & 10 \end{bmatrix}$$

If $S_J = \text{diag}(1, 1, -1, -1)$ then $W = I_4$ and (3.13)–(3.14) with m = 2 and $Y = S_J X^T$ yield

$$A_{2} = \frac{1}{2}I_{2}, \quad A_{1} = \begin{bmatrix} 2 & 0 \\ 0 & 3 \end{bmatrix}, \quad A_{0} = \begin{bmatrix} 3/2 & 0 \\ 0 & 4 \end{bmatrix}, \quad (4.10)$$

which defines a hyperbolic quadratic that is overdamped since all of its eigenvalues are negative. The two quadratics defined in (4.9) and (4.10) have the same eigenvalues but different properties. The quadratic defined in (4.9) cannot be linearized into a definite pencil whereas the quadratic defined in (4.10) can.

4.3.2 Complex symmetric quadratics

There are no additional constraints on the list \mathcal{L} of elementary divisors in (4.4). If \mathcal{L} consists of $(\lambda - \lambda_j)^{\ell_j}$, $\lambda_j \in \mathbb{C}$, j = 1:t with $\sum_{j=1}^t \ell_j = 2n$ then the corresponding complex Jordan matrix is $J = \bigoplus_{j=1}^t J_{\ell_j}(\lambda_j)$. The matrix S_J , using Section 3.5.3, takes the form

$$S_J = S_J^{-1} = \bigoplus_{j=1}^{\iota} F_{\ell_j}.$$

Since $S_J = S_J^T \in \mathbb{R}^{2n \times 2n}$, $S_J^2 = I$, there exists an orthogonal matrix W such that $W^T S_J W = \begin{bmatrix} I_n & 0 \\ 0 & \Sigma^2 \end{bmatrix}$, where $\Sigma = \begin{bmatrix} I_{n-p} & 0 \\ 0 & iI_p \end{bmatrix}$ and $p \leq n$ depends on the number of odd partial multiplicities ℓ_j (p = 0 when all eigenvalues are semisimple). If we let $\widetilde{X} = XW = \begin{bmatrix} \widetilde{X}_1 & \widetilde{X}_2 \end{bmatrix}$ with \widetilde{X}_1 nonsingular and write $\widetilde{X}_2 = i\widetilde{X}_1\Theta\Sigma$ for some complex orthogonal $\Theta \in \mathbb{C}^{n \times n}$ then since $v_S(J) = I$,

$$XS_J X^T = \widetilde{X}_1 \widetilde{X}_1^T + \widetilde{X}_2 \Sigma^2 \widetilde{X}_2^T = 0.$$

Hence we can build X as $X = \widetilde{X}W^T = [\widetilde{X}_1 \ i\widetilde{X}_1\Theta\Sigma]W^T$ which is completely determined by \widetilde{X}_1, Σ and Θ .

Example 4.3.3 Consider the complex symmetric structure and $\mathcal{L} = \{(\lambda - 2)^2, (\lambda - 2), (\lambda - i)\}$. Then

$$J = \begin{bmatrix} 2 & 1 \\ & 2 \end{bmatrix} \oplus \begin{bmatrix} 2 \end{bmatrix} \oplus \begin{bmatrix} i \end{bmatrix}, \quad S_J = \begin{bmatrix} 1 \\ 1 \end{bmatrix} \oplus \begin{bmatrix} 1 \end{bmatrix} \oplus \begin{bmatrix} 1 \end{bmatrix} \oplus \begin{bmatrix} 1 \end{bmatrix}.$$

With the choice $\widetilde{X}_1 = \Theta = I_3$, we obtain

$$X = \begin{bmatrix} -\sqrt{2}i & \sqrt{2}i & -i & 0\\ \sqrt{2} & \sqrt{2} & 0 & -i \end{bmatrix}.$$

Since $\begin{bmatrix} XJ \\ X \end{bmatrix}$ is nonsingular, (3.13)–(3.14) with m = 2 and $Y = S_J X^T$ yield

$$A_2 = \begin{bmatrix} -2.4 - 0.2i & 0.2 - 0.4i \\ 0.2 - 0.4i & 0.4 + 0.2i \end{bmatrix},$$

$$A_{1} = \begin{bmatrix} 8.6 + 0.8i & -0.8 + 0.6i \\ -0.8 + 0.6i & -0.6 - 0.8i \end{bmatrix}, \quad A_{0} = \begin{bmatrix} -7.6 - 0.8i & 0.8 + 0.4i \\ 0.8 + 0.4i & -0.4 + 0.8i \end{bmatrix}.$$

4.3.3 *-even quadratics

It follows from [66] and Section 3.5.4 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r purely imaginary (including 0) elementary divisors $(\lambda i\beta_j)^{\ell_j}$, j = 1: r, and
- t pairs of nonzero and non-purely imaginary elementary divisors $(\lambda i\mu_j)^{m_j}$, $(\lambda - i\overline{\mu}_j)^{m_j}$, j = 1:t,

with $\sum_{j=1}^{r} \ell_j + 2 \sum_{j=1}^{t} m_j = 2n$. The corresponding complex Jordan matrix is

$$J = -i\left(\bigoplus_{j=1}^r J_{\ell_j}(-\beta_j) \oplus \bigoplus_{j=1}^t (J_{m_j}(-\bar{\mu}_j) \oplus J_{m_j}(-\mu_j))\right).$$

The matrix S_J given in Section 3.5.4 satisfies $S_J = -S_J^*$, $JS_J = (JS_J)^*$ and must be such that

$$\operatorname{sig}(iS_J) = \sum_{j=1}^r \frac{1}{2} (1 - (-1)^{\ell_j}) \varepsilon_j = 0.$$
(4.11)

Note that iS_J is real symmetric and on using (4.11) we see that there exists an orthogonal matrix W such that $W^T S_J W = -i \begin{bmatrix} I_n & 0 \\ 0 & -I_n \end{bmatrix}$. As for the Hermitian case in Section 4.3.1 we can choose X to be of the form

$$X = \begin{bmatrix} \widetilde{X}_1 & \widetilde{X}_1 \Theta \end{bmatrix} W^T,$$

where $\widetilde{X}_1, \Theta \in \mathbb{C}^{n \times n}$ with \widetilde{X}_1 nonsingular and Θ unitary.

Example 4.3.4 Consider the complex *-even structure and $\mathcal{L} = \{\lambda, (\lambda - 2i), (\lambda - i)^2, (\lambda - (1+i)), (\lambda - (-1+i))\}$ with sign characteristic $\{-1, +1, -1\}$ so that (4.11) holds with $(\ell_1, \varepsilon_1) = (1, -1), (\ell_2, \varepsilon_2) = (1, 1), (\ell_3, \varepsilon_3) = (2, -1)$. Then

$$J = -i \left(\begin{bmatrix} 0 \end{bmatrix} \oplus \begin{bmatrix} -2 \end{bmatrix} \oplus \begin{bmatrix} -1 & 1 \\ & -1 \end{bmatrix} \oplus \begin{bmatrix} -1 - i \end{bmatrix} \oplus \begin{bmatrix} i - 1 \end{bmatrix} \right),$$
$$S_J = -i \left((-1) \begin{bmatrix} 1 \end{bmatrix} \oplus \begin{bmatrix} 1 \end{bmatrix} \oplus (-1) \begin{bmatrix} 1 \\ 1 \end{bmatrix} \oplus \begin{bmatrix} 1 \\ 1 \end{bmatrix} \right).$$

With the choice $\widetilde{X}_1 = \Theta = I_3$, we obtain

$$X = \begin{bmatrix} 0 & 0 & \sqrt{2}i & -\sqrt{2}i & \sqrt{2}i \\ -1 & 0 & 0 & 0 & \sqrt{2} & \sqrt{2} \\ 0 & -i & -\sqrt{2}i & -\sqrt{2}i & 0 & 0 \end{bmatrix},$$

where $\begin{bmatrix} XJ \\ X \end{bmatrix}$ is nonsingular. Now, (3.13)–(3.14) with m = 2 and $Y = S_J X^*$ yield

$$A_{2} = \begin{bmatrix} -1.5 & 1.5 & -0.5 \\ 1.5 & -0.5 & 0.5 \\ -0.5 & 0.5 & 0.5 \end{bmatrix}, \quad A_{1} = \begin{bmatrix} 5i & -2i & 2i \\ -2i & -i & -i \\ 2i & -i & -i \end{bmatrix}, \quad A_{0} = \begin{bmatrix} 4 & 0 & 2 \\ 0 & 0 & 0 \\ 2 & 0 & 0 \end{bmatrix}.$$

4.3.4 Real *T*-even quadratics

It follows from [66, Thm. 4.2] and Section 3.5.5 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r zero elementary divisors λ^{n_j} with n_j even, j = 1: r,
- t pairs of real elementary divisors $(\lambda + \alpha_j)^{\ell_j}$, $(\lambda \alpha_j)^{\ell_j}$ with ℓ_j odd if $\alpha_j = 0$, j = 1: t,
- q pairs of purely imaginary elementary divisors $(\lambda + i\beta_j)^{k_j}$, $(\lambda i\beta_j)^{k_j}$ with $\beta_j > 0, \ j = 1; q$,
- p quadruples of nonreal and non-purely imaginary elementary divisors $(\lambda + \mu_j)^{m_j}, (\lambda \mu_j)^{m_j}, (\lambda + \overline{\mu}_j)^{m_j}, (\lambda \overline{\mu}_j)^{m_j}, j = 1; p,$

with $\sum_{j=1}^{r} n_j + 2 \sum_{j=1}^{t} \ell_j + 2 \sum_{j=1}^{q} k_j + 4 \sum_{j=1}^{p} m_j = 2n$. The corresponding real Jordan matrix is

$$J = \bigoplus_{j=1}^{r} J_{n_j}(0) \oplus \bigoplus_{j=1}^{t} \left(J_{\ell_j}(\alpha_j) \oplus -J_{\ell_j}(\alpha_j)^T \right)$$
$$\oplus \bigoplus_{j=1}^{q} K_{2k_j}(i\beta_j, -i\beta_j) \oplus \bigoplus_{j=1}^{p} \left(K_{2m_j}(\mu_j, \bar{\mu}_j) \oplus -K_{2m_j}(\mu_j, \bar{\mu}_j)^T \right).$$

The matrix S_J given in Section 3.5.5 satisfies $S_J = -S_J^T$ and $JS_J = (JS_J)^T$.

Now, from the Takagi factorization of real skew-symmetric matrices (see [42, Section 4.4, problems 25-26]) and the form of S_J , we notice that there exists an orthogonal matrix W such that $W^T S_J W = \begin{bmatrix} 0 & I_n \\ -I_n & 0 \end{bmatrix}$. If we let $\widetilde{X} = XW = \begin{bmatrix} \widetilde{X}_1 & \widetilde{X}_2 \end{bmatrix}$ with $\widetilde{X}_1 \in \mathbb{R}^{n \times n}$ of full rank and write $\widetilde{X}_2 = \widetilde{X}_1 H$ for some real symmetric matrix H then

$$XS_J X^T = -\widetilde{X}_2 \widetilde{X}_1^T + \widetilde{X}_1 \widetilde{X}_2^T = -\widetilde{X}_1 H \widetilde{X}_1^T + \widetilde{X}_1 H \widetilde{X}_1^T = 0.$$

Hence we can build X as $X = \widetilde{X}W^T = \begin{bmatrix} \widetilde{X}_1 & \widetilde{X}_1H \end{bmatrix} W^T$ for some $\widetilde{X}_1, H \in \mathbb{R}^{n \times n}$ with H symmetric.

Note that H can be the zero matrix. A simple 1×1 example is $Q(\lambda) = 1 - \lambda^2$ which is built by

$$J = \begin{bmatrix} 1 \\ -1 \end{bmatrix}, \quad S_J = \begin{bmatrix} -1 \\ 1 \end{bmatrix}, \quad W = \begin{bmatrix} \frac{1-i}{2} & \frac{1-i}{2} \\ \frac{1+i}{2} & \frac{-1-i}{2} \end{bmatrix},$$

with $\widetilde{X}_1 = 1$ and H = 0.

4.3.5 Complex *T*-even quadratics

It follows from [66, Thm. 4.2] and Section 3.5.6 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r zero elementary divisors λ^{m_j} with m_j even, j = 1: r, and
- t pairs of elementary divisors $(\lambda \lambda_j)^{\ell_j}$, $(\lambda + \lambda_j)^{\ell_j}$ with ℓ_j odd if $\lambda_j = 0, j = 1: t$,

with $\sum_{j=1}^{r} m_j + 2 \sum_{j=1}^{t} \ell_j = 2n$. The corresponding complex Jordan matrix is

$$J = \bigoplus_{j=1}^{r} J_{m_j}(0) \oplus \bigoplus_{j=1}^{t} (J_{\ell_j}(\lambda_j) \oplus J_{\ell_j}(-\lambda_j)).$$

From the structure of the matrix S_J that is given in Section 3.5.6, there exists an orthogonal matrix W such that $W^T S_J W = \begin{bmatrix} 0 & I_n \\ -I_n & 0 \end{bmatrix}$. The rest of the construction is similar to that for real T-even quadratics in Section 4.3.4, except that H must now be complex symmetric.

4.3.6 *-odd quadratics

It follows from [66] and Section 3.5.7 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r purely imaginary (including 0) elementary divisors $(\lambda i\beta_j)^{\ell_j}$, j = 1:r, and
- t pairs of nonzero and non-purely imaginary elementary divisors $(\lambda i\mu_j)^{m_j}$, $(\lambda - i\overline{\mu}_j)^{m_j}$, j = 1: t,

with $\sum_{j=1}^{r} \ell_j + 2 \sum_{j=1}^{t} m_j = 2n$. The corresponding complex Jordan matrix is

$$J = -i\left(\bigoplus_{j=1}^{r} J_{\ell_j}(-\beta_j) \oplus \bigoplus_{j=1}^{t} \left(J_{m_j}(-\bar{\mu}_j) \oplus J_{m_j}(-\mu_j)\right)\right)$$

The matrix S_J given in Section 3.5.7 satisfies $S_J = S_J^{-1} = S_J^*$, $JS_J = -(JS_J)^*$ and must be such that

$$sig(S_J) = \sum_{j=1}^r \frac{1}{2} (1 - (-1)^{\ell_j}) \varepsilon_j = 0.$$

The construction of S-structured Jordan triples follows that of Hermitian quadratics in Section 4.3.1. In this case, for the leading matrix coefficient of $Q(\lambda)$ to be nonsingular, n must be even.

4.3.7 Real *T*-odd quadratics

It follows from [66, Thm. 4.2] and Section 3.5.8 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r zero elementary divisors λ^{ℓ_j} with ℓ_j odd, j = 1: r,
- t pairs of real elementary divisors $(\lambda + \alpha_j)^{n_j}$, $(\lambda \alpha_j)^{n_j}$ with n_j even if $\alpha_j = 0$, j = 1: t,
- q pairs of purely imaginary elementary divisors $(\lambda + i\beta_j)^{k_j}$, $(\lambda i\beta_j)^{k_j}$ with $\beta_j > 0, j = 1; q$, and
- p quadruples of nonreal and non-purely elementary divisors $(\lambda + \mu_j)^{m_j}$, $(\lambda \mu_j)^{m_j}$, $(\lambda + \overline{\mu}_j)^{m_j}$, $(\lambda \overline{\mu}_j)^{m_j}$ j = 1: p.

with $\sum_{j=1}^{r} \ell_j + 2 \sum_{j=1}^{t} n_j + 2 \sum_{j=1}^{q} k_j + 4 \sum_{j=1}^{p} m_j = 2n$. The corresponding real Jordan matrix is

$$J = \bigoplus_{j=1}^{r} J_{\ell_j}(0) \oplus \bigoplus_{j=1}^{t} \left(J_{n_j}(\alpha_j) \oplus -J_{n_j}(\alpha_j)^T \right)$$
$$\oplus \bigoplus_{j=1}^{q} K_{2k_j}(i\beta_j, -i\beta_j) \oplus \bigoplus_{j=1}^{p} \left(K_{2m_j}(\mu_j, \bar{\mu}_j) \oplus -K_{2m_j}(\mu_j, \bar{\mu}_j)^T \right)$$

The matrix S_J given in Section 3.5.8 satisfies $S_J = S_J^{-1} = S_J^T$ and $S_J^2 = I$. The construction of \mathcal{S} -structured Jordan triples follows that of real symmetric quadratics in Section 4.3.1.

Note that for $S \in \{\text{real } T\text{-even}\}$ ($S \in \{\text{real } T\text{-odd}\}$), $S_J(JS_J)$ is a skew-symmetric matrix and thus $\operatorname{sig}(iS_J) = 0$ ($\operatorname{sig}(iJS_J) = 0$). Using the form of S_J for real T-even (real T-odd) in Section 3.5, $\operatorname{sig}(iS_J) = 0$ ($\operatorname{sig}(iJS_J) = 0$) does not impose any constraint on ε_j 's.

4.3.8 Complex *T*-odd quadratics

It follows from [66, Thm. 4.2] and Section 3.5.9 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r zero elementary divisors λ^{ℓ_j} with ℓ_j odd, j = 1: r, and
- t pairs of elementary divisors $(\lambda \lambda_j)^{m_j}$, $(\lambda + \lambda_j)^{m_j}$ with m_j even if $\lambda_j = 0$, j = 1: t,

with $\sum_{j=1}^{r} \ell_j + 2 \sum_{j=1}^{t} m_j = 2n$. The corresponding complex Jordan matrix is

$$J = \bigoplus_{j=1}^{r} J_{\ell_j}(0) \oplus \bigoplus_{j=1}^{t} \left(-J_{m_j}(\lambda_j) \oplus J_{m_j}(\lambda_j) \right).$$

The matrix S_J given in Section 3.5.9 satisfies $S_J = S_J^{-1} = -S_J^T$ and $JS_J = (JS_J)^T$.

Consequently, there exists an orthogonal matrix W such that $W^T S_J W = \sigma \begin{bmatrix} I_n & 0 \\ 0 & \Sigma^2 \end{bmatrix}$, where $\sigma = \pm 1$ and $\Sigma = \begin{bmatrix} I_{n-p} & 0 \\ 0 & iI_p \end{bmatrix}$ and $p \leq n$. The construction of X is then analogous to that for complex symmetric quadratics in Section 4.3.2.

4.3.9 *-(anti)-palindromic quadratics

For simplicity, we assume that $-1 \notin \sigma(Q)$. It follows from [68, Thm. 7.10] and Section 3.5.10 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- r elementary divisors $(\lambda \lambda_j)^{2\ell_j + 1}$ with $\lambda_j \in \mathbb{C}$ such that $|\lambda_j| = 1, j = 1; r$,
- t elementary divisors $(\lambda \lambda_j)^{2m_j}$ with $\lambda_j \in \mathbb{C}, |\lambda_j| = 1, j = 1: t$, and
- q pairs of elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda 1/\overline{\lambda}_j)^{k_j}$ with $\lambda_j \in \mathbb{C} \setminus \{0\}, |\lambda_j| \neq 1$, j = 1: q,

with $\sum_{j=1}^{r} (2\ell_j+1)+2\sum_{j=1}^{t} m_j+2\sum_{j=1}^{q} k_j = 2n$. The corresponding complex Jordanlike matrix is $J = -S_J S_J^{-*}$ with S_J as in Section 3.5.10. The sign characteristic associated with the eigenvalues of unit modulus of $Q(\lambda)$ must satisfy a constraint similar to (4.11). However, we will not go through the details here as we can use the correspondence via Cayley transformations between \star -even/odd quadratics and \star -(anti)-palindromic quadratics as described in [65, Chap. 6]. The effect of Cayley transformations on the sign characteristic is left for future investigation. Consider for example using the complex *-even case to handle the *-palindromic one. Assume that the list of elementary divisors of a complex *-palindromic quadratic matrix polynomial $Q(\lambda)$ is given as above. Since $-1 \notin \sigma(Q)$, we can apply the Cayley transformation

$$\lambda_j \mapsto \rho_j := \frac{\lambda_j - 1}{\lambda_j + 1} \tag{4.12}$$

on the given list \mathcal{L} to obtain $\widetilde{\mathcal{L}}$. The transformation (4.12) maps 1 to zero, other eigenvalues of unit modulus are mapped to the imaginary axis. The pairs $(\lambda_j, 1/\overline{\lambda}_j)$ for which $\lambda_j \neq 0$ and $|\lambda_j| \neq 1$ are mapped to $(\rho_j, -\overline{\rho}_j), \rho_j \neq 0$. Now, we construct Jand S_J according to Section 4.3.3 to build a complex *-even $\widetilde{Q}(\lambda)$ that corresponds to $\widetilde{\mathcal{L}}$. Finally, $Q(\lambda) = (\lambda + 1)^2 \widetilde{Q} \left(\frac{\lambda - 1}{\lambda + 1}\right)$ is a complex *-palindromic quadratic matrix polynomial with elementary divisor list \mathcal{L} .

Example 4.3.5 Consider the complex *-palindromic structure and

$$\mathcal{L} = \left\{ (\lambda - 1), \left(\lambda - \frac{4i - 3}{5} \right), (\lambda - i)^2, (\lambda - (2i - 1)), \left(\lambda - \frac{1}{2i - 1} \right) \right\}.$$

The transformation (4.12) maps \mathcal{L} to

$$\widetilde{\mathcal{L}} = \{\lambda, (\lambda - 2i), (\lambda - i)^2, (\lambda - (1 + i)), (\lambda - (-1 + i))\},\$$

which, with sign characteristic $\{-1, +1, -1\}$, leads to the quadratic $\widetilde{Q}(\lambda) = \lambda^2 A_2 + \lambda A_1 + A_0$ of Example 4.3.4. Now,

$$Q(\lambda) = \lambda^2 (A_2 + A_1 + A_0) + \lambda (-2A_2 + 2A_0) + (A_2 - A_1 + A_0)$$

is *-palindromic with elementary divisor list \mathcal{L} .

Note that concerning the case where $-1 \in \sigma(Q)$ but $1 \notin \sigma(Q)$, the admissible list of elementary divisors together with the corresponding matrices J and S_J can be derived similarly from [68] and [87, Section 2.2.2]. The construction of Q then, using a suitable Cayley transformation, is along the same lines as the case where $-1 \notin \sigma(Q)$.

4.3.10 Real *T*-(anti)-palindromic quadratics

We assume that $-1 \notin \sigma(Q)$. It follows from [68, Thm. 7.6] and Section 3.5.11 that the list of elementary divisors \mathcal{L} in (4.4) is made up of

- p pairs of real elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda 1/\lambda_j)^{k_j}$ with $\lambda_j \in \mathbb{R} \setminus \{0\}$, $|\lambda_j| \neq 1, j = 1; p$,
- q quadruples of nonreal elementary divisors $(\lambda \lambda_j)^{n_j}$, $(\lambda \overline{\lambda}_j)^{n_j}$, $(\lambda 1/\lambda_j)^{n_j}$, $(\lambda -$
- v elementary divisors $(\lambda 1)^{2m_j}$, j = 1: v,
- t pairs of elementary divisors $(\lambda 1)^{2\ell_j + 1}$, $(\lambda 1)^{2\ell_j + 1}$ j = 1: t,
- *u* pairs of elementary divisors $(\lambda \lambda_j)^{\ell'_j}$, $(\lambda \overline{\lambda}_j)^{\ell'_j}$ with $|\lambda_j| = 1$, $\lambda_j \neq 1$, ℓ'_j odd, j = 1: u,
- r pairs of elementary divisors $(\lambda \lambda_j)^{m'_j}$, $(\lambda \overline{\lambda}_j)^{m'_j}$ with $|\lambda_j| = 1$, $\lambda_j \neq 1$, m'_j even, j = 1: r,

with $2\sum_{j=1}^{p} k_j + 4\sum_{j=1}^{q} n_j + 2\sum_{j=1}^{v} m_j + 2\sum_{j=1}^{t} (2\ell_j + 1) + 2\sum_{j=1}^{u} \ell'_j + 2\sum_{j=1}^{r} m'_j = 2n.$ The corresponding real Jordan-like matrix is $J = -S_J S_J^{-T}$ with S_J as in Section 3.5.11.

The rest follows from the case of real T-even/real T-odd quadratics as explained in Section 4.3.9.

4.3.11 Complex *T*-(anti)-palindromic quadratics

We assume that $-1 \notin \sigma(Q)$. It follows from [68, Thm. 7.6] and Section 3.5.12 that the list of elementary divisors \mathcal{L} in (4.4) for a complex *T*-palindromic $Q(\lambda)$ is made up of

- t elementary divisors $(\lambda 1)^{m_j}$ with m_j even, j = 1: t,
- q pairs of elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda 1/\lambda_j)^{k_j}$ with k_j odd when $\lambda_j = 1$, j = 1; q,

with $\sum_{j=1}^{t} m_j + 2 \sum_{j=1}^{q} k_j = 2n$. The corresponding complex Jordan-like matrix is $J = -S_J S_J^{-T}$ with S_J as in Section 3.5.12.

Now for a complex *T*-antipalindromic $Q(\lambda)$ with $-1 \notin \sigma(Q)$, see [68, Thm. 7.6], \mathcal{L} in (4.4) must be made of

- t elementary divisors $(\lambda 1)^{\ell_j}$ with ℓ_j odd, j = 1: t,
- q pairs of elementary divisors $(\lambda \lambda_j)^{k_j}$, $(\lambda 1/\lambda_j)^{k_j}$ with k_j even if $\lambda_j = 1$, j = 1: q,

with $\sum_{j=1}^{t} \ell_j + 2 \sum_{j=1}^{q} k_j = 2n$. The corresponding complex Jordan-like matrix $J = -S_J S_J^{-T}$ with the corresponding S_J as in Section 3.5.12.

The rest follows from the case of complex T-even/complex T-odd quadratics as explained in Section 4.3.9.

4.4 Other structured inverse eigenvalue problems

Now, we consider the problems of building a cubic $P(\lambda)$ that belongs to some classes presented in Figure 2.1 and generating hyperbolic tridiagonal quadratics.

4.4.1 Generating quasidefinite cubics

By Theorem 3.3.5 and (3.13)–(3.14), an (m, n)-self-adjoint Jordan triple $(X, J, S_J X^*)$ generates a uniquely defined $P(\lambda) = \sum_{j=0}^{m} \lambda^j A_j \in \mathcal{P}_{\mathcal{S}}(\mathbb{F}^n)$ where A_j , given by (3.13)– (3.14) for j = 0: m with $Y = S_J X^*$, are Hermitian and [29, Thm. 1.3] shows that

$$\operatorname{sig}(S_J) = \begin{cases} 0 & \text{if } m \text{ is even,} \\ \operatorname{sig}(A_m) & \text{if } m \text{ is odd.} \end{cases}$$
(4.13)

The columns of X determine the eigenvectors of a quasihyperbolic $P(\lambda)$. In order to build quasihyperbolic cubics, we want to construct an $n \times 3n$ matrix X, a $3n \times 3n$ real Jordan matrix J and a $3n \times 3n$ matrix S_J such that

$$\det \begin{bmatrix} XJ^2\\ XJ\\ X \end{bmatrix} \neq 0, \quad \begin{cases} XS_JX^{\star} = 0\\ XJS_JX^{\star} = 0 \end{cases}, \quad \det(XJ^2S_JX^{\star}) \neq 0.$$
(4.14)

Theorem 4.4.1 (quasihyperbolic cubic) For a given $n \times n$ matrix $S_1 = \text{diag}(\pm 1)$, a given $n \times n$ real diagonal matrix J_1 and a given $n \times n$ matrix X_1 with columns normalized such that $I_n - X_1S_1X_1^* > 0$, let X_2 be the Hermitian positive definite square root of $I_n - X_1S_1X_1^*$ and let X_3, J_3 be such that $X_3J_3X_3^* = X_1J_1S_1X_1^* + X_2J_2X_2^*$, where J_2 is diagonal with real entries chosen such that $X_1J_1^2S_1X_1^* + X_2J_2^2X_2^* - X_3J_3^2X_3^*$ is nonsingular. Define

$$X = \begin{bmatrix} X_1 & X_2 & X_3 \end{bmatrix}, \quad J = \begin{bmatrix} J_1 & 0 & 0 \\ 0 & J_2 & 0 \\ 0 & 0 & J_3 \end{bmatrix}, \quad S_J = \begin{bmatrix} S_1 & 0 & 0 \\ 0 & I_n & 0 \\ 0 & 0 & -I_n \end{bmatrix}.$$
(4.15)

If J has pairwise distinct diagonal entries then $(X, J, S_J X^*)$ is a (3, n)-self-adjoint Jordan triple that generates a unique quasihyperbolic cubic matrix polynomial $P(\lambda) = \lambda^3 A_3 + \lambda^2 A_2 + \lambda A_1 + A_0$. Moreover if $S_1 = I_n$ and J_1 , J_2 are chosen so that

$$\lambda_{\max}(J_2) < \lambda_{\min}(J_3) \le \lambda_{\max}(J_3) < \lambda_{\min}(J_1), \tag{4.16}$$

then $P(\lambda)$ is hyperbolic.

Proof. Note that since $S_J = S_J^*$ and $JS_J = (JS_J)^*$, $(X, J, S_J X^*)$ is a (3, n)-selfadjoint Jordan triple if the constraints in (4.14) hold. Note that we just have to check the first two constraints since the last one holds by assumption. The matrix X is clearly of full rank because X_2 is of full rank and, because J has distinct entries, the first constraint in (4.14) is satisfied. For the second constraint we have that

$$XS_J X^{\star} = X_1 S_1 X_1^{\star} + X_2 X_2^{\star} - X_3 X_3^{\star}$$

= $X_1 S_1 X_1^{\star} + I_n - X_1 S_1 X_1^{\star} - I_n = 0,$
 $XJS_J X^{\star} = X_1 J_1 S_1 X_1^{\star} + X_2 J_2 X_2^{\star} - X_3 J_3 X_3^{\star} = 0.$

If $S_1 = I_n$ then by (4.13), $A_3 > 0$. If (4.16) holds then $J_2 < J_3 < J_1$ and hence all eigenvalues belong to three distinct intervals on \mathbb{R} , each containing *n* eigenvalues of one type and the eigenvalue type of each interval alternates in sign with the right most interval being of positive type. It follows from Theorem 2.3.4 that $P(\lambda)$ is hyperbolic. \blacksquare Note that one can use Weyl's theorem [42, p. 181] together with [42, Thm. 4.5.9] to chose X_1, J_1, J_2 so that (4.16) is satisfied.

Example 4.4.2 For n = 2 the matrices

$$J_1 = \begin{bmatrix} 3.5 & 0 \\ 0 & 2 \end{bmatrix}, \quad J_2 = \begin{bmatrix} -1.5 & 0 \\ 0 & -1 \end{bmatrix}, \quad S_1 = I_2,$$

and

$$X_1 = \begin{bmatrix} -0.8819i & 0.4714i \\ 0.4714i & 0.8819i \end{bmatrix}, \quad X_2 = \begin{bmatrix} 1.4142 & 0 \\ 0 & 1.4142 \end{bmatrix},$$

satisfy the assumptions of Theorem 4.4.1 where $\star = T$ and thus, with $X_3 J_3 X_3^T$ being the eigendecomposition of $X_1 J_1 X_1^T + X_2 J_2 X_2^T$, define a (3, 2)-self-adjoint Jordan triple that uniquely determines a hyperbolic cubic matrix polynomial whose coefficient matrices are given by (3.13)–(3.14) with m = 3 and $Y = S_J X^T$.

Once more, homogeneous rotations are the tool to generate quasidefinite cubics with a singular leading coefficient.

Note that, for any m, we may use Section 2.5 to generate (diagonal) hyperbolic matrix polynomials by choosing real eigenvalues that respect the characterization (P1) in Theorem 2.3.4. These can be used to generate infinitely many (full) hyperbolic matrix polynomials. The same applies for quasihyperbolics that are diagonalizable by structure preserving congruences.

4.4.2 Generating tridiagonal hyperbolic quadratics

Ram and Elhay in [81] solve an IPEP for symmetric tridiagonal quadratic matrix polynomials in the following settings: Given two sets of distinct complex numbers $\{\lambda_j\}_{j=1}^{2n}$ and $\{\mu_j\}_{j=1}^{2n-2}$, two tridiagonal $n \times n$ symmetric matrices A_1 and A_0 are determined such that

$$\sigma(Q) = \{\lambda_j\}_{j=1}^{2n}, \quad \sigma(\widehat{Q}) = \{\mu_j\}_{j=1}^{2n-2}, \tag{4.17}$$

where $Q(\lambda) = \lambda^2 I_n + \lambda A_1 + A_0$ and $\widehat{Q}(\lambda)$ is the matrix polynomial obtained by deleting the last row and column of $Q(\lambda)$. Using [81, Alg. 2.1], we can generate infinitely many quasihyperbolic tridiagonal symmetric polynomials starting with $\{\lambda_j\}_{j=1}^{2n} \subset \mathbb{R}$. Note that all λ_j 's are of definite type as they are simple eigenvalues. We recall the following algorithms.

Algorithm 4.4.1 [81, Alg. 3.1]

Input A set $\{(z_j, y_j)\}_{j=1}^m$, where $m \ge 2$ and z_j distinct.

Output At most 2(m-1) solution sets $\{r, \{d_j\}_{j=0}^m\}$. Each set of d_j defines a polynomial $y(z) = \sum_{j=0}^m d_j z^j$ for which r is double zero and $y(z_j) = y_j$, j = 1:m.

The algorithm

(a) Solve the linear systems of equations

$$Vg = [y_1 \ y_2 \ \cdots \ y_m]^T$$
, $Vh = [z_1^m \ z_2^m \ \cdots \ z_m^m]^T$,

for the vectors g and h where

$$V = \begin{bmatrix} 1 & z_1 & z_1^2 & \cdots & z_1^{m-1} \\ 1 & z_2 & z_2^2 & \cdots & z_2^{m-1} \\ \vdots & \vdots & \vdots & & \vdots \\ 1 & z_m & z_m^2 & \cdots & z_m^{m-1} \end{bmatrix}.$$

- (b) Form the polynomials $\pi_1(z) = \sum_{j=1}^m g_j z^{j-1}$, $\pi_2(z) = -z^m + \sum_{j=1}^m h_j z^{j-1}$ and $\theta(z) = \pi_1(z)\pi_2'(z) \pi_2(z)\pi_1'(z)$.
- (c) Solve $\theta(z) = 0$.
- (d) For each zero r of $\theta(z)$,
 - (i) Use $\pi_1(r) d_m \pi_2(r) = 0$ to find d_m .
 - (ii) Use $d = g d_m h$ to determine the vector $d = \begin{bmatrix} d_0 & d_1 & \cdots & d_{m-1} \end{bmatrix}^T$.

Algorithm 4.4.2 [81, Alg. 2.1]

Input Two sets of distinct eigenvalues $\{\lambda_k^{(n)}\}_{k=1}^{2n}$ and $\{\lambda_k^{(n-1)}\}_{k=1}^{2n-2}$.

Output Two symmetric tridiagonal matrices

$$A_{1} = \begin{bmatrix} \alpha_{1} & \beta_{1} & 0 & \cdots & 0 \\ \beta_{1} & \alpha_{2} & \beta_{2} & \cdots & 0 \\ 0 & \beta_{2} & \alpha_{3} & \ddots & 0 \\ \vdots & \vdots & \ddots & \ddots & \vdots \\ 0 & 0 & 0 & \cdots & \alpha_{n} \end{bmatrix}, \quad A_{0} = \begin{bmatrix} \gamma_{1} & \delta_{1} & 0 & \cdots & 0 \\ \delta_{1} & \gamma_{2} & \delta_{2} & \cdots & 0 \\ 0 & \delta_{2} & \gamma_{3} & \ddots & 0 \\ \vdots & \vdots & \ddots & \ddots & \vdots \\ 0 & 0 & 0 & \cdots & \gamma_{n} \end{bmatrix},$$

for which $\sigma(Q) := \sigma(\lambda^2 I_n + \lambda A_1 + A_0) = \{\lambda_k^{(n)}\}_{k=1}^n \text{ and } \sigma(\widehat{Q}) = \{\lambda_k^{(n-1)}\}_{k=1}^{2n-2}$. The algorithm

(a) For
$$k = n, n - 1, \dots, 2$$
,

(i) Compute
$$y_j = y(\lambda_j^{k-1}) = -\prod_{i=1}^{2k} (\lambda_j^{(k-1)} - \lambda_i^{(k)}), \quad j = 1: 2k - 2.$$

(ii) Choose one solution set $\{r, \{c_i\}_{i=0}^{2k-2}\}$ from the choices produced by applying Algorithm 4.4.1 to $\{(\lambda_j^{(k-1)}, y_j)\}_{j=1}^{2k-2}$ and define $p(\lambda) = \sum_{i=0}^{2k-2} c_i \lambda^i$.

$$\alpha_k = \sum_{i=1}^{2k-2} \lambda_i^{(k-1)} - \sum_{i=1}^{2k} \lambda_i^{(k)}, \quad \beta_{k-1} = \pm \sqrt{c_{2k-2}},$$
$$\gamma_k = \frac{\prod_{i=1}^{2k} (r - \lambda_i^{(k)})}{\prod_{i=1}^{2k-2} (r - \lambda_i^{(k-1)})} - r^2 - r\alpha_k, \quad \delta_{k-1} = -r\beta_{k-1}.$$

- (*iv*) If k > 2,
 - (1) Determine the polynomial $q_{2k-4}(\lambda)$ of degree 2k 4 by the synthetic division of $p(\lambda)$ with the quadratic factor $(\beta_{k-1}\lambda + \delta_{k-1})^2$.
 - (2) Find the zeros $\{\lambda_i^{(k-2)}\}_{i=1}^{2k-4}$ of q_{2k-4} .

(b) Assign

$$\alpha_1 = -(\lambda_1^{(1)} + \lambda_2^{(1)}), \quad \gamma_1 = \lambda_1^{(1)} \lambda_2^{(1)}.$$

Example 4.4.3 Given $\{\lambda_j\}_{j=1}^6 = \{-6.5, -5, -4, -2, -1.5, 3\}$, we may take $\{\mu_j\}_{j=1}^4 =$

 $\{-3, -1, 0, 2\}$ and construct the following quasihyperbolic tridiagonal polynomial using Algorithm 4.4.1 and Algorithm 4.4.2.

$$Q(\lambda) = \lambda^2 I_3 + \lambda \begin{bmatrix} 4.0400 & 0.4834 & 0 \\ 0.4834 & -2.0400 & 18.7285 \\ 0 & 18.7285 & 14 \end{bmatrix} + \begin{bmatrix} 3.1312 & 1.0388 & 0 \\ 1.0388 & 0.3449 & -19.3303 \\ 0 & -19.3303 & 406.5050 \end{bmatrix},$$

with $\sigma(Q) = \{\lambda_j\}_{j=1}^6$ and $\sigma(\widehat{Q}) = \{\mu_j\}_{j=1}^4$.

Now, if the interest is in generating hyperbolic quadratic tridiagonal symmetric matrix polynomials then extra conditions must be imposed. Theorem 2.3.5 is useful in solving the SIPEP of generating hyperbolic quadratic tridiagonal matrix polynomials given two sets of real numbers $\sigma(Q) = \{\lambda_j\}_{j=1}^{2n}$ and $\sigma(\widehat{Q}) = \{\mu_j\}_{j=1}^{2n-2}$ that satisfy 1,2 and 3 of Theorem 2.3.5. This problem is always solvable, [9, Thm. 5.1].

Example 4.4.4 Given

$$\{\lambda_j\}_{j=1}^6 = \{-7, -5, -4, -3, -2, -0.5\}, \quad \{\mu_j\}_{j=1}^4 = \{-6, -4.5, -2.5, -1.5\},$$

we construct the following hyperbolic tridiagonal quadratic matrix polynomial which is consistent with the given data using Algorithm 4.4.1 and Algorithm 4.4.2.

$$\lambda^2 I_3 + \lambda \begin{bmatrix} 7.0484 & 1.2162 & 0 \\ 1.2162 & 7.4516 & 2.4121 \\ 0 & 2.4121 & 7 \end{bmatrix} + \begin{bmatrix} 9.89 & 3.9526 & 0 \\ 3.9526 & 11.8173 & 8.4171 \\ 0 & 8.4171 & 11.0684 \end{bmatrix}$$

Note that, by Theorem 2.3.5, any matrix polynomial constructed from the data given in Example 4.4.3 cannot be hyperbolic. The tridiagonal matrices produced by Algorithm 4.4.2 are unique up to signs change along the subdiagonal, see (5) on [9, p. 35].
4.5 Concluding remarks

The hardest part in treating a SIPEP via building an associated \mathcal{S} -structured Jordan triple $(X, J, S_J v_{\mathcal{S}}(J) X^{\star})$ is to solve

$$XJ^{j}S_{J}v_{\mathcal{S}}(J)X^{\star} = 0, \quad j = 0: m - 2,$$
(4.18)

for X. Equations (4.18) get more difficult to solve as m increases. We have shown how to solve these equations when m = 2 for structures $S \in S$, where

 $S = \{\text{Hermitian, symmetric, *-even, *-odd, real T-even, real T-odd, complex T-even, complex T-odd, *-palindromic, *-antipalindromic, real T-palindromic, real T-antipalindromic, complex T-palindromic, complex T-antipalindromic}\}.$

We have discussed the related quasihyperbolic inverse eigenvalue problem for m = 2, 3. Building an *S*-structured Jordan triple associated with an unknown matrix polynomial of degree grater than two given some spectral information is a challenge. Another challenge is to directly handle an IPEP when infinity is a prescribed eigenvalue. Approaches to solve (structured or unstructured) inverse quadratic eigenvalue problems without constructing standard triples are possible [64], [71].

Chapter 5

Conclusions and Future Work

The thesis provides an extensive study of a group of structured matrix polynomials that are associated with applications in science and engineering, in particular, vibration analysis in structural mechanics. Hermitian, symmetric real or complex, *-alternating and *-palindromic matrix polynomials have been investigated from different aspects. Hermitian matrix polynomials with only definite type eigenvalues in $\mathbb{R} \cup \{\infty\}$ are related to systems of differential equations with bounded solutions under small perturbations. We call such polynomials quasidefinite. The well known definite pencils and overdamped quadratics are just special cases of quasidefinite matrix polynomials. Other subclasses of quasidefinite matrix polynomials are hyperbolic, definite, quasihyperbolic and gyroscopically stabilized matrix polynomials. We have collected the quasidefinite matrix polynomials appearing in the literature and presented them in a unified framework, leading to a clear classification of these polynomials and their many subclasses, as shown in Figure 2.1. The main tools we have used to produce this diagram are homogeneous rotations of matrix polynomials and a new characterization of hyperbolic matrix polynomials that depends on the distribution of their eigenvalue types along the real line. This characterization itself relies mainly on the fact that a matrix polynomial is hyperbolic if and only if it has a definite linearization in $\mathbb{H}(P)$. Moreover, while studying quasihyperbolic matrix polynomials, we were able to spot a new class of diagonalizable matrix polynomials. Methods to determine whether a given matrix polynomial is overdamped, definite or hyperbolic have been highlighted.

We have developed a general theory of structured standard triples for structured

matrix polynomials by introducing the notion of S-structured standard triples, which extends the known notion of self-adjoint standard triples linked to Hermitian matrix polynomials. With the exception of T-(anti)palindromic matrix polynomials of even degree with both -1 and 1 as eigenvalues, we have shown that $P(\lambda)$ has structure S if and only if $P(\lambda)$ admits an S-structured standard triple, and moreover that every standard triple of a matrix polynomial with structure S is S-structured. We have investigated the important special case of S-structured Jordan triples. Indeed, it turns out that structured standard triples involve a parameter matrix, usually denoted by S as it is connected with the sign characteristic attached to real, purely imaginary eigenvalues or eigenvalues of modulus 1, according to the structure of the polynomial. The explicit form of S has been a topic of research in some recent papers that handle real symmetric and T-even quadratic matrix polynomials [18], [45]. However, we have shown that the explicit form of S, for any matrix polynomial with structure $S \in S$ of arbitrary degree, can be obtained easily by employing the existing literature on canonical forms of structured pencils.

Structured standard triples can be used to solve an SIPEP in the setting that some eigeninformation about the polynomial is given beforehand. The difficulty with this approach of treating the inverse problem comes from the conditions that the eigenvectors/generalized eigenvectors have to satisfy in order to define a structured standard triple. For structures in S, we have dealt with a structured quadratic inverse eigenvalue problem and we have shown how to generate quasidefinite quadratic and cubic matrix polynomials by solving proper IPEPs. We are considering the possibility of having a MATLAB toolbox generating such structured quadratics and quasidefinite cubics. For more general structured matrix polynomials of higher degrees, the inverse problem is much more complicated. Another approach for handling structured quadratic inverse eigenvalue problems is the quadratic realizability problem (QRP) [64] which is to explicitly construct a specific $Q(\lambda)$ that realizes a given admissible list of elementary divisors. The solution of the QRP in [64] does not involve eigenvectors at all which allows more freedom and simplifies the inverse problem significantly. The quadratic matrix polynomials constructed in [64] are sparse and more specifically they consist of blocks that are lower anti-diagonal with very low (anti-)bandwidth. Structured version of the QRP, in particular Hermitian, is a topic of an ongoing research [71]. The structured version is more complicated due to the required symmetries plus the role that the concept of sign characteristic plays.

There is a Cayley correspondence between *-alternating and *-palindromic matrix polynomials [65, Section 6.3]. In [2], we are working on a similar correspondence between Hermitian and *-palindromic matrix polynomials which leads to a clear definition of the eigenvalue type for eigenvalues of modulus 1 associated with a *palindromic matrix polynomial of either even or odd degree. Consequently we can extend [55, Thm. 14], which provides stable boundedness conditions for the solutions of *-palindromic difference equations of even degree, to include the case of odd degree equations. The result we obtain is consistent with the one in [82] which handles the linear *-palindromic difference equations.

Bibliography

- Y. Agranovich, T. Azizov, A. Barsukov, and A. Dijksma. On the inverse spectral problem for a quadratic Jacobi matrix pencil. J. Math. Anal. Appl., 306:1–17, 2005.
- [2] M. Al-Ammari, D. S. Mackey, and F. Tisseur. On the association between Hermitian and *-palindromic matrix polynomials. Manuscript, 2011.
- [3] M. Al-Ammari and F. Tisseur. Hermitian matrix polynomials with real eigenvalues of definite type. Part I: Classification. *Linear Algebra Appl.*, doi:10.1016/j.laa.2010.08.035, 2010.
- [4] M. Al-Ammari and F. Tisseur. Standard triples for structured matrix polynomials. MIMS EPrint 2011.37, Manchester Institute for Mathematical Sciences, The University of Manchester, UK, May 2011. 20 pp.
- [5] A. Amiraslani, R. M. Corless, and P. Lancaster. Linearization of matrix polynomials expressed in polynomial bases. *IMA J. Numer. Anal.*, 29:141–157, 2009.
- [6] A. L. Andrew, K. E. Chu, and P. Lancaster. Derivatives of eigenvalues and eigenvectors of matrix functions. SIAM J. Matrix Anal. Appl., 14(4):903–926, 1993.
- [7] E. N. Antoniou and S. Vologiannidis. A new family of companion forms of polynomial matrices. *Electron. J. Linear Algebra.*, 11:78–87, 2004.
- [8] E. N. Antoniou and S. Vologiannidis. Linearizations of polynomial matrices with symmetries and their applications. *Electron. J. Linear Algebra.*, 15:107– 114, 2006.

- [9] T. Ya. Azizov, A. Dijksma, K.-H. Föster, and P. Jonas. Quadratic (weakly) hyperbolic matrix polynomials: Direct and inverse spectral problems. *Oper. Theory Adv. Appl.*, 198:11–40, 2009.
- [10] Z. Bai, J. Demmel, J. Dongarra, A. Ruhe, and H. Van der Vorst. Templates for the solution of Algebraic Eigenvalue Problems: A Practical Guide. Society for Industrial and Applied Mathematics, Philadelphia, PA, USA, 2000. ISBN 0-89871-471-0. xx+410 pp.
- [11] L. Barkwell and P. Lancaster. Overdamped and gyroscopic vibrating systems. Trans. AME: J. Appl. Mech., 59:176–181, 1992.
- [12] L. Barkwell, P. Lancaster, and A. Markus. Gyroscopically stabilized systems: A class of quadratic eigenvalue problems with real spectrum. *Canad. J. Math.*, 44(1):42–53, 1992.
- [13] T. Betcke, N. J. Higham, V. Mehrmann, C. Schröder, and F. Tisseur. NLEVP: A collection of nonlinear eigenvalue problems. MIMS EPrint 2010.98, Manchester Institute for Mathematical Sciences, The University of Manchester, UK, November. 2010. 25 pp.
- [14] R. Byers, V. Mehrmann, and H. Xu. Trimmed linearizations for structured matrix polynomials. *Linear Algebra Appl.*, 429:2373–2400, 2008.
- [15] Y.-F. Cai, Y.-C. Kuo, W.-W. Lin, and S.-F. Xu. Solutions to a quadratic inverse eigenvalue problem. *Linear Algebra Appl.*, 430:1590–1606, 2009.
- [16] D. Chu, M. Chu, and W.-W. Lin. Quadratic model updating with symmetry, positive definiteness and no spill-over. SIAM J. Matrix Anal. Appl., 31(2): 546–564, 2009.
- [17] M. T. Chu and G. Golub. Inverse Eigenvalue Problems: Theory, Algorithms, and Applications. Oxford University Press, 2005. ISBN 0-19-856664-6. 387 pp.
- [18] M. T. Chu and S.-F. Xu. Spectral decomposition of real symmetric quadratic λ -matrices and its applications. *Math. Comp.*, 78:293–313, 2009.
- [19] C. R. Crawford and Y. S. Moon. Finding a positive definite linear combination of two Hermitian matrices. *Linear Algebra Appl.*, 51:37–48, 1983.

- [20] H. Dai and P. Lancaster. Newton's mathod for a generalized inverse eigenvalue problem. Numer. Linear Algebra Appl., 4(1):1–21, 1997.
- [21] B. N. Datta. Numerical Linear Algebra and Applications. Brooks/Cole Publishing Company, Pacific Grove, CA, USA, 1995. ISBN 0-534-17466-3. xxii+680 pp.
- [22] P. I. Davies, N. J. Higham, and F. Tisseur. Analysis of the Cholesky method with iterative refinement for solving the symmetric definite generalized eigenproblem. SIAM J. Matrix Anal. Appl., 23(2):472–493, 2001.
- [23] F. De Terán and F. M. Dopico. Sharp lower bounds for the dimension of linearizations of matrix polynomials. *Electron. J. Linear Algebra.*, 17:518–531, 2008.
- [24] F. De Terán, F. M. Dopico, and D. S. Mackey. Structured linearizations for palindromic matrix polynomials of odd degree. MIMS EPrint 2010.33, Manchester Institute for Mathematical Sciences, The University of Manchester, UK, April. 2010. 24 pp.
- [25] R. J. Duffin. A minimax theory for overdamped networks. J. Rat. Mech. Anal., 4:221–233, 1955.
- [26] R. A. Frazer, W. J. Duncan, and A. R. Collar. Elementary Matrices and Some Applications to Dynamics and Differential Equations. Cambridge University Press, 1963. xii+416 pp.
- [27] S. Friedland, J. Nocedal, and M. L. Overton. The formulation and analysis of numerical methods for inverse eigenvalue problems. *SIAM J. Numer Anal.*, 24 (3):634–667, 1987.
- [28] S. D. Garvey, P. Lancaster, A. A. Popov, U. Prells, and I. Zaballa. Filters connecting isospectral quadratic systems. *Linear Algebra Appl.*, doi:10.1016/j.laa.2011.03.040, 2011.
- [29] I. Gohberg, P. Lancaster, and L. Rodman. Spectral analysis of selfadjoint matrix polynomials. Ann. Math., 112(1):33–71, 1980.

- [30] I. Gohberg, P. Lancaster, and L. Rodman. Indefinite Linear Algebra and Applications. Birkhäuser, Basel, Switzerland, 2005. ISBN 3-7643-7349-0. xii+357 pp.
- [31] I. Gohberg, P. Lancaster, and L. Rodman. *Matrix Polynomials*. Society for Industrial and Applied Mathematics, Philadelphia, PA, USA, 2009. Unabridged republication of book first published by Academic Press in 1982.
- [32] G. H. Golub and C. F. Van Loan. *Matrix Computations*. Johns Hopkins University Press, Baltimore, MD, USA, third edition, 1996. ISBN 0-8018-5413-X (hardback), 0-8018-5414-8 (paperback). xxvii+694 pp.
- [33] C.-H. Guo, N. J. Higham, and F. Tisseur. Detecting and solving hyperbolic quadratic eigenvalue problems. SIAM J. Matrix Anal. Appl., 30(4):1593–1613, 2009.
- [34] C.-H. Guo, N. J. Higham, and F. Tisseur. An improved arc algorithm for detecting definite Hermitian pairs. SIAM J. Matrix Anal. Appl., 31(3):1131– 1151, 2009.
- [35] C.-H. Guo and P. Lancaster. Algorithms for hyperbolic quadratic eigenvalue problems. *Math. Comp.*, 74(252):1777–1791, 2005.
- [36] N. J. Higham. Functions of Matrices: Theory and Computation. Society for Industrial and Applied Mathematics, Philadelphia, PA, USA, 2008. ISBN 978-0-898716-46-7. xx+425 pp.
- [37] N. J. Higham, D. S. Mackey, N. Mackey, and F. Tisseur. Symmetric linearizations for matrix polynomials. SIAM J. Matrix Anal. Appl., 29(1):143–159, 2006.
- [38] N. J. Higham, D. S. Mackey, and F. Tisseur. Definite matrix polynomials and their linearization by definite pencils. SIAM J. Matrix Anal. Appl., 31(2): 478–502, 2009.
- [39] N. J. Higham, D. S. Mackey, and F. Tisseur. The conditioning of linearizations of matrix polynomials. SIAM J. Matrix Anal. Appl., 28(4):1005–1028, 2006.

- [40] N. J. Higham, F. Tisseur, and P. Van Dooren. Detecting a definite Hermitian pair and a hyperbolic or elliptic quadratic eigenvalue problem, and associated nearness problems. *Linear Algebra Appl.*, 351-352:455–474, 2002.
- [41] A. Hilliges, C. Mehl, and V. Mehrmann. On the solution of palindromic eigenvalue problems. Proceedings of the 4th European Congress on Computational Methods in Applied Sciences and Engineering (ECCOMAS), Jyväskylä, Finland, 2004. CD-ROM.
- [42] R. A. Horn and C. R. Johnson. *Matrix Analysis*. Cambridge University Press, 1985. ISBN 0-521-30586-1. xiii+561 pp.
- [43] R. A. Horn and C. R. Johnson. *Topics in Matrix Analysis*. Cambridge University Press, 1991. ISBN 0-521-30587-X. viii +607 pp.
- [44] T.-M. Hwang, W.-W. Li, J.-L. Liu, and W. Wang. Jacobi Davidson methods for cubic eigenvalue problems. *Numer. Linear Algebra Appl.*, 12:605–624, 2005.
- [45] Z. Jia and M. Wei. A real-valued spectral decomposition of the undamped gyroscopic system with applications. SIAM J. Matrix Anal. Appl., 23(2):584– 604, 2011.
- [46] W. Kerner. Large-scale complex eigenvalue problems. J. Comput. Phys., 85 (1):1–85, 1989.
- [47] D. Kressner, C. Schröder, and D. S. Watkins. Implicit QR algorithms for palindromic and even eigenvalue problems. *Numer. Algorithms*, 51(2):209–238, 2009.
- [48] P. Lancaster. Inverse spectral problems for semisimple damped vibrating systems. SIAM J. Matrix Anal. Appl., 29(1):279–301, 2007.
- [49] P. Lancaster. Isospectral vibrating systems. Part 1. The spectral method. Linear Algebra Appl., 409:51–69, 2005.
- [50] P. Lancaster. Lambda-Matrices and Vibrating Systems. Pergamon Press, Oxford, 1966. xiii+196 pp.

- [51] P. Lancaster. Quadratic eigenvalue problems. *Linear Algebra Appl.*, 150:499– 506, 1991.
- [52] P. Lancaster. Strongly stable gyroscopic systems. *Electron. J. Linear Algebra.*, 5:53–66, 1999.
- [53] P. Lancaster, A. S. Markus, and Q. Ye. Low rank perturbations of strongly definitizable transformations and matrix polynomials. *Linear Algebra Appl.*, 197-198:3–29, 1994.
- [54] P. Lancaster and U. Prells. Isospectral families of high-order systems. ZAMM.
 Z. Angew. Math. Mech., 87(3):219–234, 2007.
- [55] P. Lancaster, U. Prells, and L. Rodman. Canonical structures for palindromic matrix polynomials. Oper. Matrices, 1(3):469–489, 2007.
- [56] P. Lancaster and L. Rodman. Canonical forms for Hermitian matrix pairs under strict equivalence and congruence. SIAM Rev., 47(3):407–443, 2005.
- [57] P. Lancaster and L. Rodman. Canonical forms for symmetric/skew-symmetric real matrix pairs under strict equivalence and congruence. *Linear Algebra Appl.*, 406:1–76, 2005.
- [58] P. Lancaster and M. Tismenetsky. The Theory of Matrices. Academic Press, London, second edition, 1985. ISBN 0-12-435560-9. xv+570 pp.
- [59] P. Lancaster and Q. Ye. Definitizable Hermitian matrix pencils. Aequationes Mathematicae, 46:44–55, 1993.
- [60] P. Lancaster and I. Zaballa. A review of canonical forms for selfadjoint matrix polynomials. Gohberg Memorial Volume. Springer, 2011.
- [61] P. Lancaster and I. Zaballa. Diagonalizable quadratic eigenvalue problems. Mech. Systems Signal Process., 23(4):1134–1144, 2009.
- [62] P. Lancaster and I. Zaballa. Parametrizing structure preserving transformations of matrix polynomials. Manuscript, 2010.
- [63] J. Liesen and B. N. Parlett. On nonsymmetric saddle point matrices that allow conjugate gradient iterations. *Numer. Math.*, 108(4):605–624, 2008.

- [64] D. S. Mackey. Quadratic realizability of matrix polynomials. Part I: The unstructured case. Manuscript, 2011.
- [65] D. S. Mackey. Structured Linearizations for Matrix Polynomials. PhD Thesis, MIMS Eprint 2006.68, Manchester Institute for Mathematical Sciences, The University of Manchester, UK, April 2006.
- [66] D. S. Mackey, N. Mackey, C. Mehl, and V. Mehrmann. Jordan structures of alternating matrix polynomials. *Linear Algebra Appl.*, 432(4):867–891, 2010.
- [67] D. S. Mackey, N. Mackey, C. Mehl, and V. Mehrmann. Numerical methods for palindromic eigenvalue problems: Computing the anti-triangular Schur form. *Numer. Linear Algebra Appl.*, 16:63–86, 2009.
- [68] D. S. Mackey, N. Mackey, C. Mehl, and V. Mehrmann. Smith forms of palindromic matrix polynomials. *Electron. J. Linear Algebra.*, 22:53–91, 2011.
- [69] D. S. Mackey, N. Mackey, C. Mehl and V. Mehrmann. Structured polynomial eigenvalue problems: Good vibrations from good linearizations. SIAM J. Matrix Anal. Appl., 28(4):1029–1051, 2006.
- [70] D. S. Mackey, N. Mackey, C. Mehl, and V. Mehrmann. Vector spaces of linearizations for matrix polynomials. SIAM J. Matrix Anal. Appl., 28(4):971– 1004, 2006.
- [71] D. S. Mackey and F. Tisseur. Quadratic realizability of matrix polynomials. Part II: The Hermitian case. Manuscript, 2011.
- [72] A. S. Markus. Introduction to the Spectral Theory of Polynomial Operator Pencils. American Mathematical Society, Providence, RI, USA, 1988. ISBN 0-8218-4523-3. iv+250 pp.
- [73] J. G. McWhirter, P. D. Baxter, T. Cooper, S. Redif, and J. Foster. An EVD algorithm for para-Hermitian polynomial matrices. *IEEE Trans SP.*, 55(5): 2158–2169, 2007.
- [74] C. Mehl. Jacobi-like algorithms for the indefinite generalized Hermitian eigenvalue problem. SIAM J. Matrix Anal. Appl., 25:964–985, 2004.

- [75] V. Mehrmann, C. Schröder, and V. Simoncini. An implicitly-restarted Krylov method for real symmetric/skew-symmetric eigenproblems. *Linear Algebra Appl.*, doi:10.1016/j.laa.2009.11.009, 2009.
- [76] V. Mehrmann and H. Voss. Nonlinear eigenvalue problems: A challenge for modern eigenvalue methods. GAMM-Mitteilungen (GAMM-Reports)., 27:121– 152, 2004.
- [77] V. Mehrmann and H. Xu. Perturbation of purely imaginary eigenvalues of Hamiltonain matrices under structured perturbations. *Electron. J. Linear Al*gebra., 17:234–257, 2008.
- [78] V. Niendorf and H. Voss. Detecting hyperbolic and definite matrix polynomials, *Linear Algebra Appl.*, 432:1017–1035, 2010.
- [79] B. Plestenjak. Numerical methods for the tridiagonal hyperbolic quadratic eigenvalue problem. SIAM J. Matrix Anal. Appl., 28(4):1157–1172, 2006.
- [80] P. J. Psarrakos. Definite triples of Hermitian matrices and matrix polynomials. J. Comput. Appl. Math., 151:39–58, 2003.
- [81] Y. M. Ram and S. Elhay. An inverse eigenvalue problem for the symmetric tridiagonal quadratic pencil with application to damped oscillatory systems. *SIAM J. Appl. Math.*, 56(1):232–244, 1996.
- [82] L. Rodman. Bounded and stably bounded palindromic difference equations of first order. *Electron. J. Linear Algebra.*, 15:22–49, 2006.
- [83] L. Rodman. Comparison of congruences and strict equivalences for real, complex, and quaternionic matrix pencils with symmetries. *Electron. J. Linear Algebra.*, 16:248–283, 2007.
- [84] K. Rothe. Least squares element method for boundary eigenvalue problems. Internat. J. Numer. Methods Eng., 33: 2129–2143, 1992.
- [85] K. Schreiber. Nonlinear Eigenvalue Problems: Newton-Type Methods and Nonlinear Rayleigh Functionals. PhD thesis, Technische Universität Berlin, Germany, May 2008.

- [86] C. Schröder. A canonical form for palindromic pencils and palindromic factorizations. Technical Report No. 316, DFG Research Center, Matheon, Technische Universität Berlin, Germany, 2006.
- [87] C. Schröder. Palindromic and Even Eigenvalue Problems Analysis and Numerical Methods. PhD thesis, Technische Universität Berlin, Germany, April 2008.
- [88] G. W. Stewart and J.-G. Sun. Matrix Perturbation Theory. Academic Press, London, 1990. ISBN 0-12-670230-6. xv+365 pp.
- [89] F. Tisseur, S. D. Garvey, and C. Munro. Deflating quadratic matrix polynomials with structure preserving transformations. *Linear Algebra Appl.*, 435:464–479, 2011.
- [90] F. Tisseur and K. Meerbergen. The quadratic eigenvalue problem. SIAM Rev., 43(2):235–286, 2001.
- [91] K. Veselić. Note on interlacing for hyperbolic quadratic pencils. Oper. Theory Adv. Appl., 198:305–307, 2009.